



# STUDY PROTOCOL

# AN INTERNATIONAL MULTICENTRE, OPEN-LABEL FIRST IN HUMAN PHASE I/II STUDY TO EVALUATE THE SAFETY, TOLERABILITY, **BIODISTRIBUTION AND ANTITUMOUR ACTIVITY OF <sup>177</sup>LU-3BP-227 FOR THE** TREATMENT OF SUBJECTS WITH SOLID TUMOURS EXPRESSING **NEUROTENSIN RECEPTOR 1**

# Study Number D-FR-01087-001 Protocol Version 8.0, dated 12 June 2020

EUDRACT Number 2017-001263-20 **IND # Not Applicable** 

# **Sponsor**

Ipsen Pharma SAS 65, quai Georges Gorse 92100 Boulogne-Billancourt, France

### **Serious Adverse Event Reporting**

Fax Number: PPD USA and Latin America Fax Number: PPD Email: PPD

### **Pharmacovigilance/Emergency Contact**

PPD **Ipsen** Innovation ZI de Courtaboeuf – 5 Avenue du Canada

91940 Les Ulis, France

Phone: PPD (Ipsen Biopharm Ltd, 102 Park Drive, Milton Park, Abingdon, OX14 4RY, United Kingdom)

Mobile: PPD (for emergencies)

The person listed above is medically designated by the sponsor as the first point of contact for emergency situations.

Persons supplied with this information must understand that it is strictly confidential. Information contained herein cannot be disclosed, submitted for publication or used for any purpose other than that contemplated herein without the sponsor's prior written authorisation

### **Sponsor Authorised Protocol Approver**

| Name, Title: | PPD                   |                  |      |     |      |        |        |       |           |      |      |
|--------------|-----------------------|------------------|------|-----|------|--------|--------|-------|-----------|------|------|
| Address:     | Ipsen Bio<br>United K | opharm<br>ingdom | Ltd, | 102 | Park | Drive, | Milton | Park, | Abingdon, | OX14 | 4RY, |

# **Sponsor Medical Monitor**

Qualified physician designated by the sponsor who is responsible for all study-site related medical decisions that have to be taken.

| Name, Title: | PPD                                 |              |              |                    |           |
|--------------|-------------------------------------|--------------|--------------|--------------------|-----------|
| Address:     | Ipsen Biopharm Lt<br>United Kingdom | td, 102 Park | Drive, Milto | on Park, Abingdon, | OX14 4RY, |
| Mobile:      | PPD                                 |              |              |                    |           |
| Email:       | PPD                                 |              |              |                    |           |

# **Sponsor Clinical Pharmacologist**

| Name, Title: | PPD   |        |
|--------------|---|--------|
| Address:     | Ipsen Innovation, 5 avenue du Canada, 91940 Les Ulis Cedex, | France |
| Phone:       | PPD   |        |
| Fax:         | PPD   |        |
| Email:       | PPD   |        |

### **Coordinating Investigator**

| Name, Title: | PPD  |
|--------------|--|
|              |  |
| Address:     | CHU-Hôtel Dieu, 1 place Alexis-Ricordeau, 44093 Nantes Cedex 1, France |
| Phone:       | PPD  |
| Mobile:      | PPD  |
| Fax:         | PPD  |
| Email:       | PPD  |

# **Project Management and Monitoring Office**

| Address: | Aepodia France S | SARL, 3 avenue du Canada, | 91940 Les Ulis, France |
|----------|------------------|---------------------------|------------------------|
| Phone:   | PPD              |                           |                        |
| Fax:     | PPD              | -                         |                        |

### AGREEMENT – SIGNATURE PAGE

### **Protocol Title:**

# AN INTERNATIONAL MULTICENTRE, OPEN-LABEL FIRST IN HUMAN PHASE I/II STUDY TO EVALUATE THE SAFETY, TOLERABILITY, BIODISTRIBUTION AND ANTITUMOUR ACTIVITY OF <sup>177</sup>LU-3BP-227 FOR THE TREATMENT OF SUBJECTS WITH SOLID TUMOURS EXPRESSING NEUROTENSIN RECEPTOR 1

### Protocol Version 8.0, dated 12 June 2020

By signing below, I hereby confirm that I have read, discussed and understood the above mentioned version of the protocol and the background information concerning the study drug. I attest that I will carry out the study according to this protocol.

I also agree that the work will be performed according to Good Clinical Practice (GCP) guidelines, the ethical principles, as referenced in Section 13, and all currently applicable laws and regulations of the country(ies) where the study will be conducted.

### **Coordinating Investigator**

| Name       | PPD |
|------------|-----|
| Title:     | PPD |
| Date:      |     |
| Signature: |     |
|            |     |

### **Sponsor**

| Name       | PPD |  |
|------------|-----|--|
| Title:     | PPD |  |
| Date:      |     |  |
| Signature: |     |  |

### **SYNOPSIS**

| a                         |  |
|---------------------------|--|
| Sponsor name              | Ipsen Pharma SAS   |
| Name of finished product  | 177Lu-3BP-227  |
| Name of active ingredient | 1. Lu-3BP-22/  |
| Title of the study        | AN INTERNATIONAL MULTICENTRE, OPEN-LABEL FIRST IN HUMAN  |
|                           | PHASE I/II STUDY TO EVALUATE THE SAFETY, TOLERABILITY,   |
|                           | BIODISTRIBUTION AND ANTITUMOUR ACTIVITY OF "LU-3BP-22/   |
|                           | FOR THE TREATMENT OF SUBJECTS WITH SULID TUMOURS   |
| Spansor study number      | D ED 01097 001   |
| EUDPACT number            | D-FR-01087-001<br>2017 001262 20   |
| IND Pafaranca             | Not applicable   |
| Type of study/            | Phase I / phase II   |
| Phase of development      |  |
| Number of planned centres | Approximately eight centres in Europe and three centres in the United States of  |
| Number of plained centres | America for the phase I (dose escalation) and additional centres in Europe and   |
|                           | the United States of America for the potential phase I dose expansion and  |
|                           | phase II.  |
| Study hypothesis and      | STUDY HYPOTHESIS   |
| objectives                | The present study consists of two parts that aim to test the following hypotheses:   |
| 5                         | Phase I  |
|                           | <sup>177</sup> Lu-3BP-227 is sufficiently well-tolerated to permit clinical investigation in                                     |
|                           | phase II.  |
|                           | Phase II   |
|                           | <sup>177</sup> Lu-3BP-227 yields higher objective response rates in subjects who have  |
|                           | Neurotensin Receptor 1 (NTSR1) expressing cancers that are unresectable,   |
|                           | locally advanced or metastatic, based on Response Evaluation Criteria in Solid   |
|                           | Tumours (RECIST) version 1.1 by central review and as compared with the  |
|                           | historical objective response rate (ORR) obtained with current standard-of-care  |
|                           | treatment for each tumour type.  |
|                           | STUDY OD IECTIVES  |
|                           | Drimany objectives   |
|                           | Phase I  |
|                           | To establish the safety and tolerability of fractionated intravenous $(i,y_i)$   |
|                           | administrations of $^{177}$ Lu-3BP-227 in subjects with unresectable. locally  |
|                           | advanced or metastatic cancers expressing NTSR1.   |
|                           | Phase II   |
|                           | To estimate ORR of fractionated i.v. administrations of <sup>177</sup> Lu-3BP-227 in   |
|                           | subjects with unresectable, locally advanced or metastatic cancers expressing  |
|                           | NTSR1.   |
|                           |  |
|                           | Secondary objectives   |
|                           | Phase I  |
|                           | a) To determine the whole-body distribution of $^{1/2}$ Lu-3BP-227 and   |
|                           | pharmacokinetics (PK) of both <sup>1//</sup> Lu-3BP-227 and 3BP-227.   |
|                           | b) To determine the radiation dosimetry of $17$ Lu-3BP-227 (organ exposure   |
|                           | to radiation). The describes the number of the second state $C^{177}$ to 2DD 2027  |
|                           | c) To describe the preliminary antitumour activity of "Lu-3BP-22/.   |
|                           | $\begin{bmatrix} 1 \text{ Hasy II} \\ 2 \end{bmatrix}$ To further evaluate the safety profile of $\frac{177}{10}$ 2PD 227 at the |
|                           | radioactivity recommended by the phase I results   |
|                           | b) To further assess the response to treatment with <sup>177</sup> I u_3RP_227 using   |
|                           | RECIST version 1.1 and/or positron emission tomography (PFT)   |
|                           | Response Criteria in Solid Tumours (PERCIST) version 1.0 criteria  |
|                           | c) To further characterise the whole-body distribution and dosimetry of  |
|                           | <sup>177</sup> Lu-3BP-227 and PK of both <sup>177</sup> Lu-3BP-227 and 3BP-227.  |
|                           | d) To describe the influence of $^{177}$ Lu-3BP-227 on the health-related quality  |
|                           | of life of treated subjects.   |

|              | Exploratory objectives   |
|--------------|--|
|              | Phase I/II   |
|              | a) To explore the correlation between the tumour uptake of $^{177}$ Lu-3BP-227   |
|              | and the NTSR1 expression on tumours.   |
|              | b) To explore renal safety by measuring urinary specific biomarkers.   |
|              | c) To evaluate the tumour microenvironment, transcriptomics, and other<br>markers of interest for the disease through assessment of tumour             |
|              | hionsies   |
|              | d) To explore genomic alterations in circulating cell-free DNA (cfDNA)   |
|              | and in germline DNA.   |
|              | e) To collect gene mutation status for correlation with clinical outcome.  |
|              | f) To collect biobank samples for future analysis of circulating markers (optional, additional informed consent required).                             |
|              | g) To generate a model integrating PK, dosimetry, antitumour activity and safety data.   |
| Study design | This is a multicentre, open-label phase I/II study of <sup>177</sup> Lu-3BP-227 in subjects  |
|              | with unresectable, locally advanced or metastatic solid tumours expressing NTSR1 who have exhausted their available standard-of-care treatment options |
|              | and/or are deemed suitable for treatment with <sup>177</sup> Lu-3BP-227 as per the   |
|              | investigator's clinical assessment and/or their individual disease state. The study  |
|              | consists of a phase I with a dose escalation part (and potential expansion cohorts)  |
|              | and a phase II either in selected or over multiple indications in a basket approach.   |
|              | Phase I  |
|              | During phase I, it is planned to enrol subjects with unresectable, locally   |
|              | advanced or metastatic tumours expressing NTSR1 originating from either the:   |
|              | • Pancreas (pancreatic ductal adenocarcinoma, PDAC)  |
|              | • Colon and rectum (colorectal cancer, CRC)  |
|              | • Stomach (gastric cancer, GC)   |
|              | Gastrointestinal stromal tumours (GIST)  |
|              | • Head and neck region (squamous-cell carcinoma of head and neck, SCCHN)   |
|              | • Bone (Ewing Sarcoma, ES)   |
|              | For the dose escalation part, CCI  |
|              | Following eligibility confirmation it is   |
|              | Following engloting confirmation, it is  |
|              | therapeutic dose in up to six cohorts with four escalation steps. Three to five  |
|              | subjects will be treated per cohort in order to yield a minimum of three evaluable   |
|              | subjects per radioactivity level, treated at the full planned radioactivity amount   |
|              | fractionated into two administrations. Once five subjects are enrolled in a cohort,  |
|              | the enrolment will be stopped in that cohort. Once the dose escalation part has  |
|              | been completed, the maximum tolerated cumulative activity (MTCA) level may   |
|              | be repeated in an additional cohort.   |
|              | The sumulative starting activity will be 5 GBa fractionated into two   |
|              | administrations $(2 \times 2.5 \text{ GBa})$ The cumulative maximum activity will be 15 GBa  |
|              | activity ( $2 \times 7.5$ GBq). However, if the MTCA is not reached and if limiting organ  |
|              | dose levels are not exceeded, an additional cohort with three administrations at   |
|              | 7.5 GBq may be added, leading to a cumulative activity of 22.5 GBq.  |
|              | Of note, for each cohort in the dose escalation part if a subject has clinical   |
|              | benefit and an acceptable tolerability profile, and if the organ dose limits are not   |
|              | exceeded, up to four additional cycles of <sup>177</sup> Lu-3BP-227 can be administered  |
|              | every 4 weeks after the end of the core trial (EOCT). The safety data evaluation   |
|              | will be conducted by the sponsor. The decision to administer additional cycles   |
|              | is based on the investigator's judgement and subject's discretion and must be  |
|              | discussed with and agreed upon by the sponsor.   |

| The    | MTCA is defined as the maximum tolerated cumulative activity that may   |
|--------|---|
| be a   | dministered following fractionated i.v. administrations of at least 4 weeks   |
| apar   | t, so that:   |
| •      | No more than 33% of the subjects experience a dose limiting toxicity  |
|        | (DLT) during Cycles 1 and/or 2 and  |
| •      | The cumulative radiation in each target organ does not exceed the   |
|        | acceptability limits.   |
| The    | DLTs are defined for any of the following IMP-related AEs according to  |
| Nati   | onal Cancer Institute - Common Terminology Criteria for Adverse Events  |
| (NC    | I-CTCAE) scale version 5.0, that occur during the defined DLT assessment  |
| perio  | od (from the first administration of <sup>177</sup> Lu-3BP-227 to EOCT/ED):   |
| •      | Grade 4 neutropenia for seven or more consecutive days;   |
| •      | Febrile neutropenia or neutropenic infection (defined as a documented   |
|        | infection with neutrophil count decreased Grade 3 or 4);  |
| •      | Grade 3 or 4 thrombocytopenia (platelet count decreased) with clinically  |
|        | meaningful bleeding (i.e. requiring urgent hospitalisation or transfusion   |
|        | to manage the bleeding);  |
| •      | Grade 4 thrombocytopenia for seven or more consecutive days;  |
| •      | Any Grade 3 anaemia (Hb<8.0 g/dL; transfusion indicated) or Grade 4   |
|        | anaemia (life-threatening consequences; urgent intervention indicated):   |
| •      | Any Grade 3 or higher laboratory abnormalities in aspartate   |
|        | aminotransferase/alanine aminotransferase (AST/ALT) with  |
|        | accompanying Grade 2 or higher bilirubin (Hy's law);  |
| •      | Any Grade 3 or higher renal injury/toxicity (estimated glomerular   |
|        | filtration rate (eGFR) <30 mL/min/1.73 m <sup>2</sup> );  |
| •      | Any Grade 3 or higher GI AE, not resolved to Grade ≤2 within 48 hours   |
|        | despite optimal adequate medical management, with the following   |
|        | specifications:   |
|        | - Grade 3 nausea, vomiting (inadequate oral caloric or fluid intake;  |
|        | tube feeding, total parenteral nutrition or hospitalisation   |
|        | indicated)  |
|        | - Grade 3 diarrhoea (increase of >7 stools per day over baseline:   |
|        | hospitalisation indicated: severe increase in ostomy output   |
|        | compared to baseline; limiting self-care activities of daily living   |
|        | (ADL)) or Grade 4 diarrhoea (life-threatening consequences;   |
|        | urgent intervention indicated)  |
|        | - Grade 3 constipation (obstipation with manual evacuation  |
|        | indicated; limiting self-care ADL) or Grade 4 constipation  |
|        | (life-threatening consequences; urgent intervention indicated);   |
| •      | Any toxicity related to <sup>177</sup> Lu-3BP-227 resulting in a treatment delay of   |
|        | more than four weeks due to either delayed recovery to baseline or  |
|        | resolution of any AE to Grade $\leq 2$ (with the exception of alopecia and  |
|        | lymphopenia).   |
| •      | Grade 5 toxicity (death)  |
| Stuc   | y design following phase I dose escalation results  |
| Upo    | a completion of the phase I dose escalation or upon reaching the MTCA and   |
| conf   | irmed jointly by the safety review committee (SRC) and the sponsor, and in  |
| cons   | ideration of the accumulated subject data, cohorts of subjects will be studied  |
|        | runer characterise the safety and efficacy of "Lu-5BP-227.  |
|        | e case of acceptable toterability and evident antitumour activity across all<br>led subjects in phase L a phase II basket trial design will be utilised to study. |
| the    | antitumour activity of <sup>177</sup> Lu-3BP-227 in subjects with NTSR1 expression  |
| time   | surs Sample size estimations for this design will be provided as part of a  |
| brot   | col amendment However if the antitumour activity is driven by a type of   |
|        | our, tumour-specific phase II cohort(s) will be initiated utilising an Optimal  |
| Sime   | m's Two Stage design (see Phase II).  |
| I If s | afety evaluation and dose schedules of <sup>177</sup> Lu-3BP-227 cannot be fully  |
| expl   | ored during the phase I dose escalation part, the expansion part will serve to  |
| acco   | mplish this objective including, but not limited to, schedules of high loading  |

|                           | doses followed by fractionated lower doses or evaluation of <sup>177</sup> Lu-3BP-227 in  |
|---------------------------|---|
|                           | combination with other antitumoral treatments (to be defined). The expansion  |
|                           | part will also serve to clarify any uncertainties of antitumour responses.  |
|                           | from the dose escalation part and the modelling and simulation approach   |
|                           | Phase II  |
|                           | Phase II study will be conducted either with a basket design trial or   |
|                           | indication-specific cohorts with an Optimal Simon's Two Stage design,   |
|                           | according to the scenarios described above.   |
|                           | CCI   |
|                           |   |
|                           |   |
|                           | One or two further  |
|                           | cohorts may be opened (subject to results emerging from ongoing preclinical   |
|                           | studies and antitumour efficacy seen during dose escalation and amending the  |
|                           | current protocol) likely to enrol subjects with GC and/or SCCHN.  |
|                           | • The PDAC cohort will enrol approximately 55 subjects and will   |
|                           | investigate whether "Lu-SBP-227 attains an OKK superior to a clinically accepted historical threshold of current standard of care     |
|                           | treatment for subjects with unresectable locally advanced or metastatic   |
|                           | disease   |
|                           | • The CRC cohort will enrol approximately 70 subjects and will  |
|                           | investigate whether <sup>177</sup> Lu-3BP-227 attains an ORR superior to a  |
|                           | clinically accepted historical threshold of current standard-of-care for  |
|                           | subjects with unresectable, locally advanced or metastatic disease.   |
|                           | The current protocol will be amended at the end of the phase I to document the  |
|                           | rationale of the phase II design. In any case, the cumulative activity administered   |
| Number of subjects        | During the phase Lescalation part   |
| Number of subjects        | up to 30 subjects will receive the <sup>177</sup> Lu-3BP-227  |
|                           | therapeutic dose. For the phase II, approximately 125 subjects (55 PDAC and 70  |
|                           | CRC subjects) are planned to be enrolled for Optimal Simon's Two Stage design.  |
|                           | In case of the implementation of phase I expansion cohorts, up to 45 additional   |
|                           | subjects will be enrolled. Similarly, if additional cohorts of subjects with GC,  |
|                           | ES or SCCHN in the phase II are to be studied, approximately 120 additional   |
|                           | subjects will be enrolled. In total, the phase I/II study would therefore enrol up  |
|                           | For a single-arm basket trial approach sample size justification will be provided   |
|                           | as part of a protocol amendment.  |
| Main eligibility criteria | Inclusion criteria:   |
|                           | Eligible subjects must fulfil all the following inclusion criteria:   |
|                           | Phase I   |
|                           | (1) Signed informed consent form prior to all study procedures.   |
|                           | <ul> <li>(2) Aged 18 years or older.</li> <li>(3) Histologically or cytologically confirmed unresectable locally advanced.</li> </ul> |
|                           | or metastatic disease and has received prior lines of standard-of-care  |
|                           | chemotherapy/treatment and has no further suitable treatment options  |
|                           | and a documented decision by a multidisciplinary oncology board   |
|                           | including a specialist of the concerned pathology.  |
|                           | (4) Subjects have:  |
|                           | (a) PDAC, or<br>(b) CPC (colorectal adargagering ma) ar   |
|                           | (b) CKC (colorectal adenocarcinoma), or<br>(c) GC (gastric adenocarcinoma), or  |
|                           | (d) Gastrointestinal Stromal Tumours (GIST), or   |
|                           | (e) SCCHN, or   |
|                           | (f) ES.   |
|                           | (5) Tumours showing:  |
|                           | (a) uptake of <sup>177</sup> Lu-3BP-227 (screening formulation) in known  |
|                           | primary or metastatic sites as judged by the investigator to be   |
|                           | greater than background; or   |

|          | (b) uptake of <sup>111</sup> In-3BP-227 in known primary or metastatic sites<br>(for subjects who participated in Study D-FR-01087-002) as<br>judged by the investigator to be greater than background. |
|----------|---|
| (6       | 6) Measurable disease (based on RECIST version 1.1).  |
| (        | 7) Criterion 7 is removed by protocol amendment.  |
| ()       | B) Documentation of progressive disease in the 6 months prior to study start (treatment).   |
| (9       | D) Eastern Cooperative Oncology Group (ECOG) performance status of 0<br>or 1 (unless if disability is related to surgery in ES and agreed with the  |
| (        | sponsor).<br>10) Adequate organ function as evidenced by:   |
|          | (a) Leukocytes $\geq 3000/\mu L$  |
|          | (b) Absolute neutrophil count $\geq 1500/\mu L$   |
|          | (c) Platelets $\geq$ 75,000/µL  |
|          | (d) Hb >9g/dL or >10 g/dL (if history of cardiac disease)   |
|          | (e) I otal serum bilirubin $\leq 2 \times$ upper normal institutional limits (ULN)  |
|          | (f) Aspartate aminotransferase/alanine aminotransferase $\leq 2.5 \times ULN$<br>( $\leq 5 \times ULN$ if subject has liver metastases)   |
|          | (g) Estimated glomerular filtration rate $\geq$ 55mL/min.   |
|          | Estimated life expectancy of $>3$ months.<br>Estimated subjects must not be pregnant or lectoting at study entry and  |
|          | during the course of the study and must not become pregnant for at least  |
|          | 6 months following the last study treatment. Women of childbearing  |
|          | potential must agree to use a highly effective method of contraception  |
|          | (see note below).   |
|          | 13) Male subjects must not father children during the study and for at least 6  |
|          | months after the last study treatment and in addition must agree to use a   |
|          | IMP. For males with partners who are of child bearing potential effective   |
|          | contraception is a combination of male condom with either cap.  |
|          | diaphragm or sponge with spermicide (double barrier methods), but these   |
|          | are not considered to be highly effective. A man is considered to be  |
|          | infertile if he has had bilateral orchidectomy or successful vasectomy.   |
|          | Effective contraception includes a female partner of childbearing   |
|          | potential if she is using highly efficacious contraception (see note below),<br>but the male subject must agree to use a condom to protect his partner as   |
|          | described above   |
|          | 14) Must be willing and able to comply with study restrictions and to remain  |
|          | at the clinic for the required time during the study period and willing to  |
|          | return to the clinic for the follow-up evaluation, as specified in the  |
|          | protocol.   |
|          | ote: Highly effective methods of contraception that result in a low failure rate $10^{-10}$ par year) when used consistently and connectly include combined   |
|          | estrogen and progestogen containing) hormonal contracention associated  |
| N N      | with inhibition of ovulation (oral, intravaginal, or transdermal).  |
| p        | rogestogen-only hormonal contraception associated with inhibition of  |
| 0        | vulation (oral, injectable, or implantable), intrauterine device, intrauterine  |
| h        | ormone-releasing system, bilateral tubal occlusion, vasectomised partner (the   |
| v.       | asectomised partner has received medical assessment of the surgical success   |
|          | i leasi o monins prior to the first study treatment and provided that partner is<br>the sole sexual partner of the female subject of childbearing potential trial                                       |
|          | articipant), or sexual abstinence:  |
|          | rue abstinence, when in line with the preferred and usual lifestyle of the subject,   |
| is       | considered a highly effective method only if defined as refraining from   |
| h        | eterosexual intercourse during the entire period of study treatment and for 6   |
| m        | nonths after the last dose of 11/Lu-3BP-227. The reliability of sexual abstinence   |
| n n      | eeas to be evaluated in relation to the duration of the clinical study and the  |
| <i>p</i> | rejerreu ana usuai iijestyle of the subject. Perioaic abstinence (e.g. calendar,  |

| ovula<br>acce<br>Fem<br>mena<br>Pern<br>and<br>12 m<br>horm<br>post<br>repla<br>sing<br>For<br>abst<br>(i.e.<br>subja<br>The<br>and i<br>vill   | auton, symptothermal, and post-ovulation method) and withdrawal are not<br>ptable methods of contraception;<br>ale subject is considered of childbearing potential i.e. fertile, following<br>arche and until becoming postmenopausal unless permanently sterile.<br>nanent sterilisation methods include hysterectomy, bilateral salpingectomy<br>bilateral oophorectomy. A postmenopausal state is defined as no menses for<br>nonths without an alternative medical cause. A high follicle stimulating<br>none (FSH) level in the postmenopausal range may be used to confirm a<br>menopausal state in women not using hormonal contraception or hormonal<br>accement therapy. However, in the absence of 12 months of amenorrhea, a<br>le FSH measurement is insufficient.<br>male subjects, acceptable methods of effective contraception include sexual<br>mence, successful vasectomy, bilateral orchidectomy and barrier methods<br>condom). In addition, male subjects should not donate sperm and female<br>exts should not donate eggs for at least 2 years after the last study treatment.<br><b>Se II</b><br>inclusion criteria for phase II will be revised based on the scenario adopted<br>ndication(s) selected for investigation based on the results from phase I and<br>be documented as part of a protocol amendment. |
|---|--|
| WIII  | e accontented as part of a protocol amenament.   |
|   | · · · ·  |
| Excl  |  |
| Eligi   | ble subjects must not have any of the following:   |
| Phase | se 1/11  |
| (1)   | Prior treatment received   |
|   | (a) Any antitumour treatment since last documented disease   |
|   | progression  |
|   | (b) Any chemotherapy within 3 weeks or nitrosourea within 6 weeks<br>prior to first treatment investigational medicinal product (IMP)<br>administration  |
|   | (c) Any curative radiotherapy within 4 weeks or palliative<br>radiotherapy within 7 days prior to first treatment IMP<br>administration  |
|   | (d) Any, monoclonal antibodies within 4 weeks or tyrosine kinases<br>inhibitors within 2 weeks prior to the first treatment IMP<br>administration  |
|   | (e) Any other IMP within 2 weeks prior to first treatment IMP administration, if the previous compound is a mechanism-based molecularly targeted agent whose half-life $(t_{1/2})$ is not well-characterised.  |
| (2)   | Brain metastases.  |
|   | Nephrectomy, renal transplant or concomitant nephrotoxic therapy putting the subject at high risk of renal toxicity during the study.  |
| (4)   | Only nonmeasurable metastatic bone lesions.  |
| (5)   | Existing or planned colostomy during study participation.  |
| (6)   | Any history of inflammatory bowel disease.   |
| (7)   | Any uncontrolled significant medical, psychiatric or surgical condition<br>or laboratory finding, that would pose a risk to subject safety or interfere<br>with study participation or interpretation of individual subject results  |
| (8)   | Clinically significant abnormalities on electrocardiogram (ECG) at screening including corrected QT interval (Fridericia's formula)  |
| (9)   | >450 msec for males or 470 msec for females at screening.<br>Previously received external beam irradiation to a field that includes  |
|   | more than $30\%$ of the bone marrow or kidneys.  |
|   | Criterion 10 is removed by protocol amendment.   |
| (11)  | Any unresolved NCI-CICAE Grade 2 or higher (except alopecia) from  |
|   | previous antitumour treatment and/or medical/surgical  |
|   | procedures/interventions.  |
| (12)  | Known allergy to IMP or its excipients administered in this study,   |
|   | including imaging contrast media.  |
| (13)  | Positive pregnancy test (female subjects).   |

| judgment of the investigator.(15)Unable to understand the nature, scope and possible consequences of the<br>study, in the judgment of the investigator.(16)Sponsor employees or investigator site personnel directly affiliated with<br>this study, and their immediate families. Immediate family is defined as<br>a spouse, parent, child or sibling, whether biological or legally adopted.Treatment: route, strength,<br>regimenFor both screening and treatment formulations, the specific activity of the IMP<br>is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.<br>The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.<br>The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in<br>a total volume of 20 mL.<br>The total radioactivity of the treatment IMP formulation will be fractionated and<br>administered in two i.v. infusions separated by at least 4 weeks (28 days). A<br>100 mL saline solution will be administered intravenously over a period of<br>30 minutes concomitantly with every IMP administration. The subject will be<br>instructed to drink water (at least 1.5 L/24 hours) on the days following each<br>IMP administration.Reference treatment: route,<br>strength, regimenSTUDY ENDPOINTS<br>Primary endpointsPhase 1<br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>UT un3RP-227 to FOCTED  |
|---|
| (15)       Unable to understand the nature, scope and possible consequences of the study, in the judgment of the investigator.         (16)       Sponsor employees or investigator site personnel directly affiliated with this study, and their immediate families. Immediate family is defined as a spouse, parent, child or sibling, whether biological or legally adopted.         Treatment: route, strength, regimen       For both screening and treatment formulations, the specific activity of the IMP is 25 μg 3BP-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL. The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       Phase 1         For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>17</sup> Lu-3BP-227 to FOCT/ED   |
| study, in the judgment of the investigator.(16)Sponsor employees or investigator site personnel directly affiliated with<br>this study, and their immediate families. Immediate family is defined as<br>a spouse, parent, child or sibling, whether biological or legally adopted.Treatment: route, strength,<br>regimenFor both screening and treatment formulations, the specific activity of the IMP<br>is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.<br>The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.<br>The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in<br>a total volume of 20 mL.<br>The total radioactivity of the treatment IMP formulation will be fractionated and<br>administered in two i.v. infusions separated by at least 4 weeks (28 days). A<br>100 mL saline solution will be administered intravenously over a period of<br>30 minutes concomitantly with every IMP administration. The subject will be<br>instructed to drink water (at least 1.5 L/24 hours) on the days following each<br>IMP administration.Reference treatment: route,<br>strength, regimenSTUDY ENDPOINTS<br>Primary endpointsCriteria for evaluation<br>(endpoints)For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>$1^{77}$ Lu.3BP-227 to FOCT/ED   |
| (16)       Sponsor employees or investigator site personnel directly affiliated with this study, and their immediate families. Immediate family is defined as a spouse, parent, child or sibling, whether biological or legally adopted.         Treatment: route, strength, regimen       For both screening and treatment formulations, the specific activity of the IMP is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.       The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Phase I         For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> In aBP-227 to EOCTED  |
| Treatment: route, strength, regimen       For both screening and treatment formulations, the specific activity of the IMP is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 10 <sup>71</sup> U. 3BP-227 to EOCT.ED  |
| Treatment: route, strength,       For both screening and treatment formulations, the specific activity of the IMP is 25 μg 3BP-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Primary endpoint second during the dose escalation part. The primary variables used for the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 170 LTCA is not identified during the dose escalation of the primary endpoint starts at the first administration of 170 LTCA is approximation of 170 LTCA is approximated for the dose escalation of the primary endpoint starts at the first administration of 170 LTCA is approximated for the dose escalation of the primary endpoint starts at the first administration of 170 LTCA is approximated for the dose escalation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of |
| Treatment: route, strength,       For both screening and treatment formulations, the specific activity of the IMP         regimen       For both screening and treatment formulations, the specific activity of the IMP         is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.       The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.       The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route,       STUDY ENDPOINTS         primary endpoints       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of $1^{17}$ Lu.  |
| Treatment: route, streining and treatment formulations, the specific activity of the IMP is 25 µg 3BP-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> Lu-3BP-277 to FOCT/ED   |
| regimen       is 25 µg 3Br-227 per 1 GBq of <sup>177</sup> Lu.         The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>17</sup> /Lu-3BP-227 to EOCT/ED  |
| The screening IMP formulation consists of 1 GBq in a total volume of 10 mL.         The treatment IMP formulation consists of 2.5 to 7.5 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> Lu-3BP-227 to EOCT/ED   |
| The treatment IMP formulation consists of 2.5 to 7.5 GBq of 17/Lu-3BP-227 in a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       STUDY ENDPOINTS         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Phase I         For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 177Lu-3BP-227 to EOCT/ED   |
| a total volume of 20 mL.         The total radioactivity of the treatment IMP formulation will be fractionated and administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 1771 u-38P-227 to EOCT/ED  |
| The total radioactivity of the treatment IMP formulation will be fractionated and<br>administered in two i.v. infusions separated by at least 4 weeks (28 days). A<br>100 mL saline solution will be administered intravenously over a period of<br>30 minutes concomitantly with every IMP administration. The subject will be<br>instructed to drink water (at least 1.5 L/24 hours) on the days following each<br>IMP administration.Reference treatment: route,<br>strength, regimenSTUDY ENDPOINTS<br>Primary endpointsPhase I<br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>$1^{17}$ Tu s3Bp-277 to EOCT/ED   |
| administered in two i.v. infusions separated by at least 4 weeks (28 days). A 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of $1^{171}$ Lu-3BP-227 to EOCT/ED   |
| 100 mL saline solution will be administered intravenously over a period of 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 1771 us3BP-227 to EOCT/ED   |
| 30 minutes concomitantly with every IMP administration. The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 177L ur3BP-227 to EOCT/ED  |
| instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 177L u-3BP-227 to EOCT/ED  |
| IMP administration.         Reference treatment: route, strength, regimen       Not applicable         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Phase I         For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 177L u-3BP-227 to EOCT/ED   |
| Reference treatment: route,<br>strength, regimen       Not applicable         Criteria for evaluation<br>(endpoints)       STUDY ENDPOINTS<br>Primary endpoints         Phase I<br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED   |
| strength, regimen       TT         Criteria for evaluation (endpoints)       STUDY ENDPOINTS         Primary endpoints       Primary endpoints         Phase I       For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of 177L u-3BP-227 to EOCT/ED   |
| Criteria for evaluation<br>(endpoints)       STUDY ENDPOINTS<br>Primary endpoints         Phase I<br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED   |
| (endpoints)<br><i>Primary endpoints</i><br><i>Phase I</i><br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED   |
| Phase I<br>For the dose escalation, the primary endpoint is MTCA or the maximum<br>administered cumulative activity (MACA), if the MTCA is not identified during<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED   |
| For the dose escalation, the primary endpoint is MTCA or the maximum administered cumulative activity (MACA), if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> Lu-3BP-227 to EOCT/ED   |
| administered cumulative activity (MACA), if the MTCA of the maximum<br>the dose escalation part. The primary variables used for the MTCA<br>determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED  |
| the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined above) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> Lu-3BP-227 to EOCT/ED  |
| determination will be the incidence of DLTs (as defined above) and the organ<br>exposure to radiation during two cycles of treatment. The DLT period for the<br>determination of the primary endpoint starts at the first administration of<br>177L u-3BP-227 to EOCT/ED  |
| exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint starts at the first administration of <sup>177</sup> Lu-3BP-227 to EOCT/ED   |
| determination of the primary endpoint starts at the first administration of   |
| determination of the primary endpoint starts at the first administration of $177L_{\rm H}$ 3BP-227 to EOCT/ED   |
| $1^{1/2}$ III-3 RP-227 to EOC T/ED  |
|   |
| Safety evaluation will encompass DLTs, frequency and nature of adverse events   |
| (AEs), abnormal findings from physical examination, vital signs, 12-lead ECG  |
| and 24-hour 3-lead ECG Holter, ECOG performance status treatment related  |
| deterioration and clinical laboratory tests abnormalities (including haematology,   |
| blood biochemistry, hormone analysis, urinalysis and pregnancy test).   |
| In case the phase I dose expansion cohorts are implemented, the primary   |
| endpoint will be safety and tolerability measured by the type, severity.  |
| expectedness and frequency of AEs.  |
|   |
| Phase II  |
| The primary endpoint is ORR measured by CT or MRI using RECIST  |
|   |
| version 1.1 Turnous represente and participation of the   |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the  |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by   |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.   |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.   |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.<br>Secondary endpoints  |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.<br>Secondary endpoints Phase I  |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the<br>time of occurrence of first clinical signs of disease progression as determined by<br>the investigator.<br>Secondary endpoints<br>Phase I<br>Pharmacokinetics, biodistribution and dosimetry  |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the<br>time of occurrence of first clinical signs of disease progression as determined by<br>the investigator.<br>Secondary endpoints<br><u>Phase I</u><br><u>Pharmacokinetics, biodistribution and dosimetry</u><br>For biodistribution and dosimetry of <sup>177</sup> Lu-3BP-227, the secondary endpoints   |
| version 1.1. Tumour response assessments are performed every 8 weeks or at the<br>time of occurrence of first clinical signs of disease progression as determined by<br>the investigator.<br>Secondary endpoints<br><u>Phase I</u><br><u>Pharmacokinetics, biodistribution and dosimetry</u><br>For biodistribution and dosimetry of <sup>177</sup> Lu-3BP-227, the secondary endpoints<br>are:   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post</li> </ul> </li> </ul>  |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve</li> </ul> </li> </ul>   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood: terminal t<sub>1/2</sub> of         </li> </ul></li></ul>   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood.     </li> </ul></li></ul>   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood.         </li> </ul> </li></ul>  |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry      </li> <li>For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood.</li> <li>b) Highest absorbed dose, specific absorbed dose to the target lesions (Gv/GBq) and cumulative</li> </ul> </li> </ul>   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood.         b) Highest absorbed dose, specific absorbed dose to the target lesions (Gy/GBq), specific absorbed dose per organ (Gy/GBq) and cumulative absorbed organ doses (Gy)         </li> </ul></li></ul>   |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints         Phase I         Pharmacokinetics, biodistribution and dosimetry         For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:         <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood.         b) Highest absorbed dose, specific absorbed dose to the target lesions (Gy/GBq), specific absorbed dose per organ (Gy/GBq) and cumulative absorbed organ doses (Gy).         For PK of 3BP 227 the secondary endpoints are:         <ul> <li>a) For PK of 3BP 227 the secondary endpoints are;</li> <li>b) Highest absorbed doses (Gy).</li> </ul> </li> </ul></li></ul>  |
| <ul> <li>version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first clinical signs of disease progression as determined by the investigator.</li> <li>Secondary endpoints Phase I Pharmacokinetics, biodistribution and dosimetry For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are: <ul> <li>a) Maximal uptake (%); maximal concentration achieved (C<sub>max</sub>); time post injection to achieve maximal concentration (T<sub>max</sub>); area under the curve (AUC) at the target lesions, discernible organs and blood; terminal t<sub>1/2</sub> of activity concentrations in blood. </li> <li>b) Highest absorbed dose, specific absorbed dose to the target lesions (Gy/GBq), specific absorbed dose per organ (Gy/GBq) and cumulative absorbed organ doses (Gy). For PK of 3BP-227, the secondary endpoints are: <ul> <li>c) Pharmacokinatic parameters including but not limited to C = AUC</li> </ul></li></ul></li></ul>   |

|                         | unchanged drug excreted into the urine (Ae), renal clearance of the drug from plasma ( $CL_R$ ), as measured in plasma and urine at defined timepoints. |
|-------------------------|---|
|                         | Di  |
|                         | Pharmacodynamic/efficacy  |
|                         | a) Objective response rate and disease control rate (DCR), as determined  |
|                         | by RECISI version 1.1 in subjects who received IVIP.  |
|                         | b) Progression-life survival (PFS) and overall survival (OS) rates as   |
|                         | and of absorption pariod  |
|                         | end of observation period.  |
|                         | c) Evaluation of inclabolic fulliour response using (Version 1.0)   |
|                         | d) Changes in serum tumour markers relevant and specific to the underlying  |
|                         | tumour disease from Day of the first treatment administration to EOCT.  |
|                         | which is planned 6 weeks after the second <sup>177</sup> Lu-3BP-227 dose  |
|                         | administration.   |
|                         |   |
|                         | Phase II  |
|                         | Efficacy  |
|                         | a) Disease control rate, time to progression, time to response, duration of   |
|                         | response as per RECIST version 1.1.   |
|                         | b) Qualitative and quantitative changes in tumour-to-background uptake using PERCIST version 1.0.   |
|                         | c) Progression-free survival (PFS) and OS as determined from start of study   |
|                         | treatment until occurrence of event and/or 6 and 12 months after start of   |
|                         | study treatment.  |
|                         | d) Changes in serum tumour markers relevant and specific to the underlying  |
|                         | tumour disease from baseline to EOCT.   |
|                         | Subject Reported Outcomes   |
|                         | a) Changes in health-related quality of life scores from baseline to EOCT   |
|                         | measured by validated questionnaires.   |
|                         | Safety  |
|                         | a) Safety and tolerability measured by the type, severity, expectedness and   |
|                         | frequency of AEs.   |
|                         | Dhammagalination highistribution and desimates  |
|                         | Pharmacokinetics, biodistribution and dosimetry   |
|                         | a) For the observation and dosined y, the endpoints will be similar as  |
|                         | for phase i.  |
|                         | Exploratory endpoints   |
|                         | Phase I/II  |
|                         | a) Tumour uptake of $^{177}$ Lu-3BP-227 and the correlation with NTSR1  |
|                         | expression on tumour biopsies.  |
|                         | b) If applicable, tumour microenvironment and other markers of interest   |
|                         | (such as NTSR1 expression, Ki67, gene expression and DNA damage)  |
|                         | in tumour biopsies taken at baseline, at EOCT visit or at disease   |
|                         | progression, whichever occurs earlier.  |
|                         | c) Genomic profiling in circulating cfDNA and in germline DNA.  |
|                         | d) Gene mutation status in correlation with clinical outcome.   |
|                         | e) Specific renal safety biomarkers specific for proximal tubulus toxicity.   |
|                         | Biobanking (optional):  |
|                         | Serum and whole blood ribonucleic acid samples will be stored for further   |
|                         | biomarker analysis after the end of the study. Analysis of additional biomarkers  |
|                         | trom the biobank samples will be performed outside the scope of the main study  |
| Statistical math - 1-1  | and reported separately.  |
| Statistical methodology | Statistical methodology in phase I  |
|                         | Continued monitoring of DITs and toxisities will be newformed during does   |
|                         | Continued monitoring of DL1s and toxicities will be performed during dose   |
|                         | escalation. At the time of the SKC, an interim analysis of the DL1s, toxicities   |

| and selected organ absorbed doses (kidney, liver and bone marrow) will be performed to guide the dose selection during the dose escalation part.  |
|---|
| <u>Biodistribution, radiation dosimetry, PK of <sup>177</sup>Lu-3BP-227</u><br>Descriptive summaries of PK and biodistribution parameters will be presented<br>for each cohort over the treatment period.   |
| <u>Pharmacokinetics of 3BP-227</u><br>If 3BP-227 levels are measurable in plasma and urine, PK parameters of 3BP-227 will be derived using a noncompartmental approach. An attempt to build an integrated model taking into account PK, dosimetry, antitumour and safety data will be made.   |
| <u>Pharmacodynamics/Efficacy Analysis</u><br>In order to perform preliminary antitumour activity, ORR and DCR will be<br>tabulated. Tumour response will be evaluated by the site investigator.<br>Independent review of tumour assessment may be requested at the discretion of<br>the sponsor. In both cases, response and progression will be evaluated using the<br>revised RECIST guideline (version 1.1). Only subjects with measurable disease<br>at baseline, who have received at least two administrations of <sup>177</sup> Lu-3BP-227 and<br>reached the end of Cycle 2 or EOCT visit will be considered evaluable for<br>response. |
| Statistical methodology in phase II   |
| Efficacy Analysis<br>For phase II cohorts, tumour response will be assessed in imaging modalities of<br>CT or MRI scans after Cycle 2 and subsequently every 8 weeks for the first<br>6 months and every 12 weeks thereafter. Images will be reviewed by an<br>independent central review core laboratory. ORR, best overall response (BOR)<br>and other efficacy parameters will be tabulated. Objective response rate will be<br>calculated combining the number of subjects with a BOR of confirmed complete<br>response (CR) or partial response (PR) per RECIST version 1.1.   |
| For the Simon Two-Stage design, the hypotheses that will be tested for each cohort are: $H_0: ORR \le ORR_0$ versus the alternative H1: $ORR > ORR_0$ where ORR is the true objective response rate following <sup>177</sup> Lu-3BP-227 treatment that warrants further clinical development, and $ORR_0$ is the minimum objective response rate to be excluded from further clinical development. The thresholds for ORR and $ORR_0$ may be updated based on results from phase I and the  |
| evolving scientific knowledge.<br>Objective response rate will be analysed at the end of Stage 1 (and no later than<br>after the 16-week visit of the last evaluable subject of the Stage 1 cohort for each<br>PDAC and CRC cohort). If the observed number of responders is below a<br>predefined threshold, the respective study cohort will be stopped for futility.<br>Otherwise, additional subjects will be treated to complete the planned enrolment.<br>At the end of Stage 2, the null hypothesis will be rejected depending on the total<br>observed number of responders based on a predefined threshold.                            |
| At the end of phase II, descriptive summaries will be provided for all primary<br>and secondary efficacy endpoints. For the primary endpoint, final analysis will<br>take into account the sequential sampling procedure of the design and the<br>underlying binomial distribution assumed by the Simon Two-Stage design.   |
| <u>Safety</u><br>Descriptive statistics will be calculated on the safety parameters. No formal statistical analyses of safety data are planned.   |
| Biodistribution, radiation dosimetry, PK of <sup>177</sup> Lu-3BP-227 and 3BP-227<br>Analysis will be performed as for phase I.   |

# **TABLE OF CONTENTS**

| RAT  | IONALE F  | OR PROTOCOL AMENDMENT #7                               | 19        |  |
|------|---|--|-----------|--|
| LIST | IST OF DEFINITIONS AND ABBREVIATIONS                          |  |           |  |
| 1    | BACKGR  | OUND INFORMATION                                       | 27        |  |
| 1.1  | Introduction27  |  |           |  |
| 1.2  | Name and Description of the Investigational Medicinal Product |  | 27        |  |
| 1.3  | Nonclinica  | l Data   | 27        |  |
|      | 1.3.1   | In Vitro Studies                                       | 27        |  |
|      | 1.3.2   | Toxicology   | 27        |  |
|      | 1.3.3   | Pharmacokinetic Properties of Unlabelled 3BP-227       | 27        |  |
|      | 1.3.4   | Biodistribution and Dosimetry of Radiolabelled 3BP-227 | <i>28</i> |  |
|      | 1.3.5   | Pharmacodynamic Effect In Vivo Study                   | 29        |  |
| 1.4  | <b>Clinical Da</b>  | ata  | 29        |  |
| 1.5  | Rationale   | for the Study  | 30        |  |
| 1.6  | Selection o   | f Dosage   | 32        |  |
| 1.7  | Population  | to be Studied  | 34        |  |
| 1.8  | Known and   | d Potential Risks to Human Subjects                    | 35        |  |
| 2    | STUDY HYPOTHESIS  |  | 37        |  |
| 2.1  | Phase I   |  | 37        |  |
| 2.2  | Phase II  |  | 37        |  |
| 3    | STUDY O   | BJECTIVES  | 38        |  |
| 3.1  | <b>Primary O</b>  | bjectives  | 38        |  |
|      | 3.1.1   | Phase I  | <u>38</u> |  |
|      | 3.1.2   | Phase II   | <u>38</u> |  |
| 3.2  | Secondary   | Objectives   | 38        |  |
|      | 3.2.1   | Phase I  | <u>38</u> |  |
|      | 3.2.2   | Phase II   | <u>38</u> |  |
| 3.3  | Explorator  | ry Objectives  | 38        |  |
|      | 3.3.1   | Phase I/II   | <u>38</u> |  |
| 4    | INVESTIC  | GATIONAL PLAN  | 38        |  |
| 4.1  | <b>Overall St</b>   | udy Design and Plan                                    | 38        |  |
| 4.2  | Phase I   |  | 39        |  |
|      | 4.2.1   | Dose Escalation  | <u>39</u> |  |
|      | 4.2.1.1   | Definition of the MTCA                                 | 40        |  |
|      | 4.2.1.2   | Cohorts Description                                    | <i>40</i> |  |
|      | 4.2.1.3   | Dose Escalation Mode                                   | 41        |  |
|      | 4.2.1.4   | Safety Review Committee                                | 41        |  |
|      | 4.2.2   | Dose Expansion   | <i>43</i> |  |
| 4.3  | Phase II  |  | 43        |  |
| 4.4  | <b>Study End</b>  | points   | 43        |  |
|      | 4.4.1   | Primary Endpoint                                       | <i>43</i> |  |

|     | 4.4.1.1                        | Phase I  | 43 |
|-----|--------------------------------|--|----|
|     | 4.4.1.2                        | Phase II   | 44 |
|     | 4.4.2                          | Secondary Endpoints  | 44 |
|     | 4.4.2.1                        | Phase I  | 44 |
|     | 4.4.2.2                        | Phase II   | 44 |
|     | 4.4.3                          | Exploratory Endpoints  | 45 |
| 4.5 | Study Du                       | ration   | 45 |
| 4.6 | Randomi                        | sation and Blinding  | 45 |
|     | 4.6.1                          | Method of Randomisation  | 45 |
|     | 4.6.1.1                        | Subject Unique Identifier  | 45 |
|     | 4.6.1.2                        | Randomisation Number   | 45 |
|     | 4.6.2                          | Blinding   | 45 |
|     | 4.6.3                          | Code-break   | 45 |
| 4.7 | Discontin                      | uation Rules   | 46 |
|     | 4.7.1                          | Individual Discontinuation Rules                                   | 46 |
|     | 4.7.1.1                        | Definition of Dose Limiting Toxicity (Dose Escalation)             | 46 |
|     | 4.7.1.2                        | Procedures for Subject Discontinuation                             | 47 |
|     | 4.7.1.3                        | Replacement Rules  | 47 |
|     | 4.7.2                          | Discontinuation of a Cohort or a Site or Study Termination         | 47 |
| 5   | SELECT                         | ION OF SUBJECTS  | 49 |
| 5.1 | Inclusion                      | Criteria   | 49 |
|     | 5.1.1                          | Phase I  | 49 |
|     | 5.1.2                          | Phase II   | 50 |
| 5.2 | Exclusion                      | ı Criteria (Phase I/II)  | 50 |
| 5.3 | Rationale                      | e for Specific Inclusion/Exclusion Criteria                        | 51 |
| 6   | STUDY CONDUCT                  |  | 52 |
| 6.1 | Study Schedule of Assessments5 |  | 52 |
| 6.2 | Study Vis                      | sits and Procedures  | 52 |
|     | 6.2.1                          | Screening  | 52 |
|     | 6.2.1.1                        | Informed Consent   | 52 |
|     | 6.2.1.2                        | Screening Procedures   | 52 |
|     | 6.2.2                          | Treatment Period and Additional Cycles for Phase I Dose Escalation | 53 |
|     | 6.2.2.1                        | Early Discontinuation or End of Core Trial Visit                   | 53 |
|     | 6.2.2.2                        | End of Additional Cycles   | 53 |
|     | 6.2.2.3                        | End of Study   | 54 |
|     | 6.2.3                          | Long-term Follow-up  | 54 |
|     | 6.2.4                          | Unscheduled Visits   | 54 |
| 6.3 | Study Re                       | strictions   | 54 |
|     | 6.3.1                          | Concomitant Medications  | 54 |
| 7   | STUDY 7                        | <b>FREATMENT</b>   | 57 |
| 7.1 | Investiga                      | tional Medicinal Product   | 57 |

|     | 7.1.1                                | Dosage Form and Strength  | 57    |
|-----|--------------------------------------|---|-------|
|     | 7.1.1.1                              | Decentralized Supply Dosage Form and Strength                       | 57    |
|     | 7.1.1.2                              | Centralized Supply Dosage Form and Strength                         | 57    |
|     | 7.1.2                                | Supplies, Packaging and Labelling                                   | 57    |
|     | 7.1.3                                | Storage   | 57    |
|     | 7.1.4                                | Preparation, Dispensing and Administration Procedures               | 58    |
|     | 7.1.4.1                              | Route of Administration   | 58    |
|     | 7.1.4.2                              | Dosing Regimen in Phase I   | 58    |
|     | 7.1.4.3                              | Dosing Regimen in Phase II  | 58    |
|     | 7.1.5                                | Accountability  | 58    |
|     | <i>7.1.6</i>                         | Spillage  | 58    |
| 7.2 | Product                              | Complaints  | 59    |
| 8   | SAFETY                               | ASSESSMENTS   | 60    |
| 8.1 | Adverse                              | Events  | 60    |
|     | <i>8.1.1</i>                         | Definition of an Adverse Event                                      | 60    |
|     | 8.1.2                                | Adverse Events Categorisation, Recording, and Follow-up             | 61    |
|     | 8.1.2.1                              | Assessment of Expectedness  | 61    |
|     | 8.1.2.2                              | Follow-up of Adverse Events   | 61    |
|     | 8.1.2.3                              | Reporting of Adverse Events   | 62    |
|     | <i>8.1.3</i>                         | Serious Adverse Event Assessment and Reporting to Sponsor           | 62    |
|     | 8.1.4                                | Suspected Unexpected Serious Adverse Reactions                      | 63    |
|     | 8.1.5                                | Pregnancy   | 63    |
|     | <i>8.1.6</i>                         | Adverse Events of Special Interest                                  | 64    |
|     | <b>8.1.</b> 7                        | Deaths  | 64    |
|     | <i>8.1.8</i>                         | Reporting to Competent Authorities, IECs/IRBs and other Investigato | rs 64 |
| 8.2 | Specific                             | Safety Assessments  | 64    |
|     | <i>8.2.1</i>                         | Physical Examinations   | 64    |
|     | 8.2.2                                | Vital Signs   | 65    |
|     | 8.2.3                                | Height and Weight   | 65    |
|     | 8.2.4                                | Electrocardiograms  | 65    |
|     | 8.2.5                                | Performance Status  | 66    |
|     | 8.2.6                                | Clinical Laboratory Tests   | 66    |
|     | 8.2.6.1                              | Blood Analyses  | 67    |
|     | 8.2.6.2                              | Urinalysis  | 67    |
|     | 8.2.6.3                              | Pregnancy Test  | 68    |
|     | <b>8.</b> <i>2</i> <b>.</b> <i>7</i> | Specific Renal Safety Biomarkers                                    | 68    |
| 9   | PHARM                                | ACOKINETIC ASSESSMENTS  | 69    |
| 9.1 | Pharmac                              | cokinetics of <sup>177</sup> Lu-3BP-227                             | 69    |
|     | <i>9.1.1</i>                         | Blood Sample Collection   | 69    |
|     | 9.1.2                                | Urine Sample Collection   | 69    |
| 9.2 | Nuclear                              | Medicine Imaging for Dosimetry                                      | 70    |

| 9.3  | Pharmaco    | kinetics of 3BP-227   | . 70      |
|------|-------------|---|-----------|
|      | 9.3.1       | Blood Sample Collection   | 70        |
|      | 9.3.2       | Urine Sample Collection   | 71        |
| 10   | PHARMA      | CODYNAMIC/EFFICACY ASSESSMENTS  | . 72      |
| 10.1 | Imaging A   | ssessments  | . 72      |
|      | 10.1.1      | Screening Imaging Assessments   | 72        |
|      | 10.1.2      | Phase I Imaging Assessments   | 72        |
|      | 10.1.3      | Phase II Imaging Assessments  | 73        |
| 10.2 | Quality of  | Life Questionnaire  | . 73      |
| 10.3 | Tumour N    | Iarkers   | .74       |
| 10.4 | Genomic I   | Profiling in Circulating Cell-free DNA and Germline DNA                   | .74       |
| 10.5 | Gene Muta   | ation Status  | . 74      |
| 10.6 | Tumour B    | iopsy   | . 74      |
| 11   | BIOBANK     | XING  | . 76      |
| 12   | STATIST     | ICAL ANALYSES   | . 77      |
| 12.1 | Analysis S  | ets   | . 77      |
| 12.2 | Statistical | Methodology for Phase I   | . 77      |
|      | 12.2.1      | Sample Size Calculation   | . 77      |
|      | 12.2.1.1    | Dose Escalation   | 77        |
|      | 12.2.1.2    | Dose Expansion  | 77        |
|      | 12.2.2      | Statistical Methods   | . 77      |
|      | 12.2.3      | Demographic and Other Baseline Characteristics                            | 78        |
|      | 12.2.4      | Maximum Tolerated Single and Cumulative Activity                          | 78        |
|      | 12.2.5      | Safety Evaluation   | 78        |
|      | 12.2.6      | Pharmacokinetics, Biodistribution and Dosimetry                           | <b>79</b> |
|      | 12.2.6.1    | Biodistribution and Pharmacokinetic Analysis of <sup>177</sup> Lu-3BP-227 | 79        |
|      | 12.2.6.2    | Radiation Dosimetry Analysis of <sup>177</sup> Lu-3BP-227                 | 79        |
|      | 12.2.6.3    | Pharmacokinetic Analysis of 3BP-227                                       | 80        |
|      | 12.2.7      | Pharmacodynamic/Efficacy Evaluation                                       | 80        |
|      | 12.2.8      | Interim Analysis/Safety Review Committee                                  | 81        |
|      | 12.2.9      | Clinical Study Report   | 82        |
| 12.3 | Statistical | Methodology for Phase II  | . 82      |
|      | 12.3.1      | Sample Size Calculation   | 82        |
|      | 12.3.2      | Analysis Sets   | 83        |
|      | 12.3.3      | Demographic and Other Baseline Characteristics                            | 83        |
|      | 12.3.4      | Efficacy Evaluation   | 83        |
|      | 12.3.5      | Safety Evaluation   | 84        |
| 13   | DATA HA     | NDLING AND RECORD KEEPING   | 85        |
| 13.1 | Data Colle  | ection  | 85        |
| 13.2 | Data Repo   | orting  | . 85      |
| 13.3 | Data Man    | agement   | . 85      |

| 13.4 | Record Ke  | eeping   |
|------|--|--|
| 14   | <b>REGULATORY AND ETHICAL CONSIDERATIONS</b>                             |  |
| 14.1 | Regulatory Considerations  |  |
| 14.2 | <b>Ethical Re</b>  | view Considerations  |
| 14.3 | Subject In   | formation Sheet and Consent  |
| 14.4 | Final Repo   | ort Signature  |
| 15   | INSURAN  | CE AND FINANCE   |
| 15.1 | Insurance  |  |
| 15.2 | <b>Financial</b>   | Agreement  |
| 16   | QUALITY  | CONTROL AND QUALITY ASSURANCE  |
| 16.1 | Protocol A   | mendments and Protocol Deviations and Exceptions   |
|      | <i>16.1.1</i>  | Protocol Amendments  |
|      | 16.1.2   | Protocol Deviations and Exceptions   |
|      | 16.1.3   | Information to Study Personnel   |
| 16.2 | Monitorin  | g  |
| 16.3 | Investigate  | or's Regulatory Obligations  |
|      | 16.3.1   | Audit and Inspection   |
|      | 16.3.2   | Data Quality Assurance   |
| 17   | PUBLICATION POLICY   |  |
| 18   | REFERENCES   |  |
| 19   | ATTACHMENTS  |  |
| 19.1 | Attachment 1 – Study Schedule of Assessments for Phase I Dose Escalation |  |
| 19.2 | 2 Attachment 2 – Blood Sampling Summary 104                              |  |
| 19.3 | Attachmen  | nt 3 – Clinical Laboratory Tests 105   |
| 19.4 | Attachmen  | nt 4 – RECIST Version 1.1 Guidelines107  |
| 19.5 | Attachmer<br>Adaptive l  | nt 5 – Statistical Details and Simulation Results for Implementation of<br>Bayesian Dose Escalation Design |
|      | 1951   | Statistical Model  |
|      | 19.5.1   | Trial Simulation Details   |
|      | 19.5.2   | Simulation Scenarios   |
|      | 19.5.5<br>19.5 <i>A</i>  | Operational Characteristics 120  |
|      | 19.5.4   | Simulation Results   |
| 19.6 | Attachmo   | $\frac{131}{136}$  |
| 10.7 | Attachmo   | 110 = RECIST  to TERCIST   |
| 17.1 | Attachinel   | t = 1 rotocor Amenument #7 Summary   |

# LIST OF TABLES

| Table 1  | Radiation Dose Estimates for Adult Humans based on Dosimetry Studies in      |   |
|----------|--|---|
|          | Rat and Dogs   | ) |
| Table 2  | Estimated Safety Margin in Study/Species based on Human Equivalent           |   |
|          | Dose Calculated Based on Body Surface Area                                   | 3 |
| Table 3  | Radiochemical Physical Characteristics of <sup>177</sup> Lutetium            | 1 |
| Table 4  | Preliminary NTSR1 Expression Profile in Targeted Tumours                     | 5 |
| Table 5  | Cohorts and Radioactivity Escalation Plan 4                                  | l |
| Table 6  | Timepoints and Replicates of 12-lead ECG Recordings                          | 6 |
| Table 7  | Schedule of Assessments for Phase I – For Subjects who are Screen Failures   |   |
|          | after IMP Screening Administration   | 3 |
| Table 8  | Schedule of Assessments for Phase I – Dose Escalation                        | ) |
| Table 9  | Schedule of Assessments for Additional Cycles in Case of Clinical Benefit    |   |
|          | and Good Tolerability  | 3 |
| Table 10 | Prior Probability Distribution of MTSA12'                                    | 7 |
| Table 11 | Prior Probability Distribution of MTCA 12'                                   | 7 |
| Table 12 | Simulated Dose-DLT Probabilities   | ) |
| Table 13 | Main Operational Characteristics for the Five Trial Simulation Scenarios 132 | 2 |

# LIST OF FIGURES

| Figure 1 | Phase I Study Design Scheme  |
|----------|--|
| Figure 2 | Flow of Dose Escalation Cohorts and SRC Meetings 42  |
| Figure 3 | 40 Random Observations (Curves) from the Prior Distribution 127                                |
| Figure 4 | Dose versus DLT Rates After Cycle 1 and Cycle 2 Simulation Scenarios 130                       |
| Figure 5 | Mean (SD) Sample Size per Single or Cumulative Dose Level for the Five<br>Simulation Scenarios |
| Figure 6 | Boxplot of Total Number of Subjects During Cycle 1 for the Five Simulation<br>Scenarios        |
| Figure 7 | Boxplot of Total Number of Subjects During Cycle 2 for the Five Simulation<br>Scenarios        |
| Figure 8 | Boxplot of Total Number of Cohorts for the Five Simulation Scenarios 135                       |

## **RATIONALE FOR PROTOCOL AMENDMENT #7**

The protocol was amended to update the following:

- Clarification of the inclusion criteria for subject selection as follows:
  - to clearly state nonresectable locally advanced disease
  - to clearly state that no further suitable treatment options are available for subjects eligible for the study
- Allow subjects screened and found positive for NTSR1 in the <sup>111</sup>In-IPN01087 phase I diagnostic study to take part in this study without having the diagnostic dose of <sup>177</sup>Lu-IPN01087 during the screening phase
- Extend the long-term follow-up period from 2 years to a maximum of 5 years or until lost to follow-up, withdrawal of consent or death, whichever occurs first.
- Revision of the DLT criteria (Section 4.7.1.1) to adequately describe the grading as stated in the CTCAE 5.0 dictionary
- Revision of the subject discontinuation rules so if there are life threatening toxicities outside of the DLT period, treatment is discontinued
- Optimise the dosimetry evaluation through adaptation of the imaging schedule
- Clarification of biopsy collection
- Clarification about COVID-19 added following the recent pandemic
- . CCI
- Make various clarifications and minor corrections for consistency

All modifications (except minor changes) are presented in Attachment 19.7.

| Protocol version                   | Rationale for amendment  |
|------------------------------------|--|
| V1.0, 02AUG2017                    | NA – initial version   |
| V2.0, 16OCT2017,<br>Amendment #1   | To update tumour biopsy inclusion criteria as per Ethical Committee review.  |
| V3.0, 09NOV17,<br>Amendment #2     | To incorporate changes to inclusion criteria, dose escalation part,<br>physical examination and ECG assessments as per review by the<br>Health Authorities in France.  |
| V4.0, 6DEC2017<br>Amendment #3     | To include Ewing Sarcoma as an additional indication in the phase<br>I/II study and to provide updated information regarding the IMP.<br>In addition, some inconsistencies in the protocol have been corrected.  |
| V5.0, 02MAR2018<br>Amendment #4    | To give precision on the calculation of the TGR, information about<br>drug-drug interactions, clarification of discontinuation process,<br>information about infusion rate in response to adverse events and<br>increase time for use of contraception for females in the inclusion<br>criteria from 30 days to 6 months, as well as information about<br>spillages.<br>In addition, some administration changes and minor inconsistencies<br>in the protocol have been corrected. |
| V6.0, 17 July 2018<br>Amendment #5 | The protocol has been amended to improve the determination of the biokinetics of <sup>177</sup> Lu-3BP-227 and perform an absolute quantification of radioactivity in target organs. Whole body scans (planar  |

# PROTOCOL HISTORY

| Protocol version                   | Rationale for amendment  |  |  |  |
|------------------------------------|--|--|--|--|
|                                    | scintigraphy) have been added to single photon emission computed tomography (SPECT /CT) during treatment period. Whole body scans will allow the calculation of whole body time-integrated activity coefficient ("residence time") that is needed for dosimetry analysis as it accounts for nonspecific activity in the body. Inclusion criteria n°3 has been updated to enable the recruitment of patients who do not have a compelling standard-of-care option.  |  |  |  |
|                                    | In addition, some administration changes and minor corrections have been added.  |  |  |  |
| V7.0, 20 June 2019<br>Amendment #6 | <ul> <li>The protocol was amended to update personnel (the sponsor authorised protocol approver and sponsor medical monitor), to update the background information, especially new nonclinical toxicology data, to update the number of subjects receiving screening and therapeutic dose, <sup>OCI</sup> and and gene mutation status as exploratory objectives and endpoints, to change pharmacokinetic timepoints to improve the clinical feasibility, to specify the biopsy conditions and put them as optional assessments, to remove tumour markers assessments for gastric cancer (CA72-4) and squamous-cell carcinoma of head and neck (TPA), to refine the exclusion criteria regarding body weight, to clarify discontinuation rules, to clarify the duration of the safety follow-up period after the IMP screening dose administration and the reporting of AE collection after the last IMP administration, to specify that death due to disease progression will be reported as an SAE, to specify details on the preparation of the CSR, to add schedule of assessments for screen failure subjects and to make various clarifications and corrections for consistency.</li> <li>Following local competent authority feedback, the sponsor was requested to provide an updated study protocol including only highly effective methods of contraception for female subjects and female partner of male participants. Additionally, the sponsor was requested to:</li> <li>Clarify the duration of use of effective contraception for male subjects</li> <li>Amend the study protocol so that only subjects who fulfil all of the inclusion criteria are included in the clinical study by clarify the anging protocol deviation definition</li> <li>Specify in detail under which conditions re-screening of subjects will be possible</li> <li>Clarify that all AEs during the core trial will be followed up until resolution or stabilisation independent of the level of severity or causality for treated subjects</li> </ul> |  |  |  |

| Protocol version | Rationale for amendment   |  |  |
|------------------|---|--|--|
|                  | The protocol was amended to update the following as requested by the FDA:   |  |  |
|                  | • Clarification of the inclusion criteria for subject selection as follows:   |  |  |
| V71(USA)         | - to clearly state nonresectable locally advanced disease   |  |  |
| 20 August 2019,  | - to clearly state that no further suitable treatment options are available for subjects eligible for the study                                       |  |  |
|                  | • Revision of the DLT criteria to adequately describe the grading as stated in the CTCAE v5.0 dictionary  |  |  |
|                  | • Revision of the subject discontinuation rules so if there are life-threatening toxicities outside of the DLT period, treatment will be discontinued |  |  |

# LIST OF DEFINITIONS AND ABBREVIATIONS

| TERM             | Wording Definition   |
|------------------|--|
| Audit            | A systematic and independent examination of the study-related<br>activities and documents to determine whether the evaluated<br>study-related activities were conducted, and the data were recorded,<br>analysed, and accurately reported according to the protocol,<br>sponsor's standard operating procedures, good clinical practices, and<br>the applicable regulatory requirement(s).   |
| Complaint        | A complaint is any written, electronic, or oral communication that<br>alleges deficiencies related to the identity, quality, purity, durability,<br>reliability, safety or effectiveness, or performance of a drug or drug<br>delivery system.   |
| Compliance       | Adherence to all the study-related requirements, good clinical practices requirements and the applicable regulatory requirements.  |
| End of study     | End of study (EOS) is the date of the last visit or last scheduled procedure shown in the Study Schedule for the last active subject in the study.   |
| Enrol/Randomise  | The act of assigning a subject to a treatment. Subjects who are<br>enrolled in the study are those who have been assigned to a<br>treatment.   |
| Enter/Consent    | The act of obtaining informed consent for participation in a clinical<br>study from subjects deemed or potentially eligible to participate in<br>the clinical study. Subjects entered into a study are those who sign<br>the informed consent document directly or through their legally<br>acceptable representatives.  |
| Ethics Committee | A board or committee (institutional, regional, or national) composed<br>of medical professionals and non-medical members whose<br>responsibility is to verify that the safety, welfare and human rights of<br>the subjects participating in a clinical study are protected.  |
| Investigator     | A physician responsible for the conduct of a clinical study at a study<br>site. If a study is conducted by a team of individuals at a study site,<br>the investigator is the responsible leader of the team and may be<br>called the principal investigator.   |
| Screen           | The act of determining if an individual meets minimum requirements<br>to become part of a pool of potential candidates for participation in a<br>clinical study. In this study, screening involves invasive or<br>diagnostic procedures and/or tests (for example, diagnostic<br>psychological tests, x-rays, blood draws). Informed consent for these<br>screening procedures and/or tests shall be obtained; this consent may<br>be separate from obtaining consent for the study. |
| Subject          | An individual who is or becomes a participant in clinical research,<br>either as a recipient of the test article or as a control. A subject may<br>be either a healthy human or a patient.   |

| ABBREVIATION    | Wording Definition   |  |
|-----------------|--|--|
| %               | Percent  |  |
| βHCG            | Beta human chorionic gonadotrophin                               |  |
| μL              | Microlitre   |  |
| ACE             | Angiotensin-converting enzyme                                    |  |
| ADL             | Activity of daily living   |  |
| Ae              | Cumulative amount of unchanged drug excreted into the urine      |  |
| AE              | Adverse event  |  |
| AKI             | Acute kidney injury  |  |
| ALT             | Alanine aminotransferase   |  |
| AST             | Aspartate aminotransferase                                       |  |
| AUC             | Area under the (plasma concentration versus time) curve          |  |
| AUCinf          | Area under the curve from time zero to infinity                  |  |
| BOR             | Best overall response  |  |
| ceCT/MRI        | Contrast enhanced computed tomography/magnetic resonance imaging |  |
| cfDNA           | Cell-free deoxyribonucleic acid                                  |  |
| CL              | Clearance  |  |
| CL <sub>R</sub> | Renal clearance of the drug from plasma                          |  |
| Cmax            | Observed maximal (peak) concentration                            |  |
| CR              | Complete responder   |  |
| CRC             | Colorectal cancer  |  |
| CRO             | Contract research organisation                                   |  |
| CSR             | Clinical study report  |  |
| СТ              | Computed tomography  |  |
| СҮР             | Cytochrome P   |  |
| DCR             | Disease control rate   |  |
| DLT             | Dose limiting toxicity   |  |
| DNA             | Deoxyribonucleic acid  |  |
| DOM             | Dosimetry operational manual                                     |  |
| DOR             | Duration of response   |  |
| EC              | Ethics committee   |  |
| ECG             | Electrocardiogram  |  |
| ECOG            | Eastern Cooperative Oncology Group                               |  |
| eCRF            | Electronic case report form                                      |  |

| ABBREVIATION | Wording Definition                           |  |
|--------------|--|--|
| ED           | Early discontinuation                        |  |
| EDC          | Electronic data capture                      |  |
| eGFR         | Estimated glomerular filtration rate         |  |
| EGFR         | Epidermal growth factor receptor             |  |
| EOAC         | End of additional cycles                     |  |
| EOCT         | End of core trial                            |  |
| EOS          | End of study                                 |  |
| ES           | Ewing Sarcoma                                |  |
| FDA          | Food and Drug Administration                 |  |
| FIH          | First in human                               |  |
| FSH          | Follicle stimulating hormone                 |  |
| GBq          | Gigabecquerel                                |  |
| GC           | Gastric cancer                               |  |
| GCP          | Good clinical practice                       |  |
| GI           | Gastrointestinal                             |  |
| GIST         | Gastrointestinal Stromal Tumours             |  |
| GPS          | Global Patient Safety                        |  |
| Gy           | Gray   |  |
| hERG         | Human ether- à-go-go-related gene            |  |
| HPLC         | High performance liquid chromatography       |  |
| IB           | Investigator's brochure                      |  |
| ICH          | International Council for Harmonisation      |  |
| IEC          | Independent Ethics Committee                 |  |
| IMP          | Investigational medicinal product            |  |
| IRB          | Institutional review board                   |  |
| IU/L         | International unit per litre                 |  |
| i.v.         | Intravenous                                  |  |
| kg           | Kilogram                                     |  |
| MACA         | Maximum administered cumulative activity     |  |
| MAD          | Maximum absorbed dose                        |  |
| mCRC         | Metastatic colorectal cancer                 |  |
| MedDRA       | Medical Dictionary for Regulatory Activities |  |
| mg           | Milligram                                    |  |
| mL           | Millilitre                                   |  |

| Millilitre per minute   |  |
|---|--|
| Millilitre per minute   |  |
| Mass spectrometer   |  |
| Maximum tolerated cumulative activity   |  |
| Maximum tolerated dose  |  |
| Maximum tolerated single activity   |  |
| National Cancer Institute - Common Terminology Criteria for<br>Adverse Events |  |
| No-observed-adverse-effect level  |  |
| Nonsteroidal anti-inflammatory drugs  |  |
| Neurotensin receptor 1  |  |
| Objective response rate   |  |
| Overall survival  |  |
| Progressive disease/progression of disease                                    |  |
| Pancreatic ductal adenocarcinoma  |  |
| PET Response Criteria in Solid Tumours  |  |
| Positron emission tomography  |  |
| Progression-free survival   |  |
| Pharmacokinetic   |  |
| Partial responder   |  |
| QT interval Fridericia's correction   |  |
| Response Evaluation Criteria in Solid Tumours                                 |  |
| Radioactivity level   |  |
| Ribonucleic acid  |  |
| Serious adverse event   |  |
| Statistical analysis plan   |  |
| Squamous-cell carcinoma of head and neck                                      |  |
| Standard operating procedure  |  |
| Single photon emission computed tomography                                    |  |
| Safety review committee   |  |
| Suspected unexpected serious adverse reaction                                 |  |
| Half-life   |  |
| Treatment emergent adverse event  |  |
| Time to maximum observed plasma concentration                                 |  |
| Time to progression   |  |
|   |  |

| ABBREVIATION | Wording Definition        |
|--------------|---------------------------|
| TTR          | Time to response          |
| ULN          | Upper limit of normal     |
| Vd           | Volume of distribution    |
| WHO          | World Health Organization |

# **1 BACKGROUND INFORMATION**

# 1.1 Introduction

This phase I/II study will be the first administration of <sup>177</sup>Lu-3BP-227 in humans under controlled study conditions. The study will generate safety and antitumour activity data and is expected to provide a better understanding of the mechanism of action of <sup>177</sup>Lu-3BP-227.

The results observed in terms of antitumour activity during dose escalation, will determine whether phase I expansion cohorts will be conducted to further investigate the safety of other activity levels and/or other administration schedule (e. g. hyperfractionation) or the investigation of efficacy of <sup>177</sup>Lu-3BP-227 in the context of indication-specific cohorts or over multiple indications.

# **1.2** Name and Description of the Investigational Medicinal Product

<sup>177</sup>Lu-3BP-227 is a drug substance consisting of a peptidomimetic neurotensin receptor 1 (NTSR1) targeting moiety, linked to a chelating DOTA moiety (1,4,7,10-tetraazacyclododecane-1,4,7,10-tetraacetic acid (DOTA), which chelates the radioactive Lutetium ( $^{177}$ Lu); it is applied in a theranostic approach.

# **1.3** Nonclinical Data

# 1.3.1 In Vitro Studies

<sup>nat</sup>Lu-3BP-227 has been extensively tested in vitro. <sup>nat</sup>Lu-3BP-227 showed specific and high binding affinity (IC<sub>50</sub> of 2.05 nM) to NTSR1, <sup>nat</sup>Lu-3BP-227 has no agonistic activity in a Ca<sup>2+</sup> mobilisation assay and prevents the Ca<sup>2+</sup> efflux induced by an NTSR1 agonist, therefore, characterising the molecule as an NTSR1 antagonist. In the same calcium mobilisation assay, 3BP-227 displayed weak agonistic activity (EC<sub>50</sub> of 95 nM) on the neurotensin receptor 2. 3BP-227 at 10  $\mu$ M and <sup>nat</sup>Lu-3BP-227 at 1  $\mu$ M did not show any other potential binding in a screening panel (Cerep) of G protein-coupled receptor.

# 1.3.2 Toxicology



# 1.3.3 Pharmacokinetic Properties of Unlabelled 3BP-227

In pharmacokinetics (PK) and metabolism studies, 3BP-227 appeared to be highly bound to human plasma proteins (86%), as well as in mouse plasma proteins (95%), with lesser binding to rat and dog plasma proteins (71% and 69%, respectively) (Study N° IPS000354). 3BP-227 was very stable in human and dog plasma in vitro (>24 hours). Metabolic stability assays in human and dog liver microsomes showed a long half-life ( $t_{1/2}$ >60 minutes) and consequently a low intrinsic clearance at a concentration of 0.1 µM of 3BP-227.

3BP-227 had a poor inhibitory potential towards cytochrome P (CYP) 1A2, 2B6, 2C8, 2C9, 2C19, 2D6 and 3A4 with a maximum inhibition of 33% towards CYP3A4 (midazolam

substrate) at 10  $\mu$ M. Thus, it is unlikely that these CYPs will be inhibited by 3BP-227 in clinical settings.

The PK of unlabelled 3BP-227 administered as a single i.v. dose has been evaluated in mice. Rat and dog toxicokinetics were performed as part of extended single i.v. dose toxicity studies in rats and dogs and of a 13-week repeated dose study in rats with the unlabelled 3BP-227. After a single i.v. dose in mice, 3BP-227 showed a low plasma clearance (CL of 64.5 mL/hour/kg) and a moderate terminal elimination  $t_{1/2}$  (1.63 hours).

After single i.v. administration in rats, the exposure (area under the curve from time zero to infinity (AUC<sub>inf</sub>)) was comparable for males and females. The terminal  $t_{1/2}$  was short and approximately two times longer in males than females (0.45 hour and 0.26 hour, respectively). The apparent volume of distribution (V<sub>d</sub>) was small and approximately two times higher in males (402 mL/kg) than in females (221 mL/kg). The systemic CL was low and similar for both genders, representing approximately 15% of the hepatic blood flow of in rats. During the 13-week repeated-dose TK study in rats, a rapid elimination phase was observed. The exposure was dose-proportional and no or weak accumulation occurred between Day 1 and Day 85 (accumulation ratios ranged from 0.9 to 1.7).

In dogs, the single i.v. bolus PK data indicate that the plasma concentrations of 3BP-227 decreased in a biphasic manner, with a short elimination  $t_{1/2}$  of  $2.15\pm0.257$  hours for the second elimination phase and no gender-dependent differences. The apparent Vd was small ( $107\pm9.8$  mL/kg) and the CL very low, representing 2.2% of the dog hepatic blood flow.

# 1.3.4 Biodistribution and Dosimetry of Radiolabelled 3BP-227

The specificity of 3BP-227 accumulation in NTSR1-expressing tumours in vivo was demonstrated in biodistribution studies in mouse xenograft models with both <sup>111</sup>In-3BP-227 and <sup>177</sup>Lu-3BP-227. Besides NTSR1-expressing tumours, the uptake of radioactivity in the organs was very low, including the kidneys and the liver. It was shown that 3BP-227 did not penetrate the blood brain barrier. <sup>177</sup>Lu-3BP-227 showed a favourable biodistribution profile exhibiting a good tumour uptake associated with a low uptake in normal tissue and a persistent tumour uptake leading to favourable tumour-to-kidney activity ratio.

In rats, the biodistribution profiles and the effective radioactive doses of <sup>111</sup>In-3BP-227 and <sup>177</sup>Lu-3BP-227 were comparable. The radioactivity was rapidly cleared from the bloodstream. Tissue concentrations were low. Elimination of radioactivity was primarily through the urine and to a lesser extent via the faeces.

In a biodistribution and dosimetry study of <sup>177</sup>Lu-3BP-227 in Beagle dogs, the radioactivity was rapidly cleared from the blood. <sup>177</sup>Lu-3BP-227 was predominantly and rapidly excreted through the urinary tract with a secondary elimination route via the liver into the gastrointestinal (GI) tract. The main uptake of radioactivity in organs was observed at 2 and 6 hours after injection in muscles, intestines, bones and bone marrow, liver, lungs and kidneys. At 72 hours post injection, all tissues exhibited less than 1% of the injected dose. The highest absorbed dose coefficients were observed in the urinary bladder wall, the large intestinal wall and the kidneys.

Table 1 below summarises the extrapolation of the organ exposure results from these two studies to humans. The extrapolation to humans was performed using a mass-based extrapolation in which the concentration in the animal organs was converted to a concentration in human organs by multiplying the animal concentration by a ratio of the total body weight of the animals and humans [1]. The dosimetry studies in rats and dogs identified as target organs of special interest: urinary bladder, large intestinal wall, kidneys and osteogenic cells.

From the extrapolation of animal data to human, the expected radiation exposure in human study is acceptable.

| Compound             | <sup>177</sup> Lu-3BP-227 | <sup>111</sup> In-3BP-227 | <sup>177</sup> Lu-3BP-227 |
|----------------------|---------------------------|---------------------------|---------------------------|
| Species              | Rat                       | Rat                       | Dog                       |
| Specific activity    | 1.2 to 2.4 mCi/nmol       | 1.2 to 2.4 mCi/nmol       | 9.9 MBq/µg                |
| Radioactive dose     | 50 or 100 µCi/rat         | 50 or 100 µCi/rat         | 94.6±5.6 MBq              |
| API dose             |                           |                           | 2.2 µg/kg                 |
| Main organs          | M - F<br>(mSv/MBq)        | M – F<br>(mSv/MBq)        | M - F<br>(mSv/MBq)        |
| Kidney               | 0.170 - 0.280             | 0.160 - 0.250             | 0.16 - 0.11               |
| Liver                | 0.094 - 0.140             | 0.130 - 0.180             | 0.03 - 0.02               |
| Brain                | 0.007 - 0.006             | 0.033 - 0.040             | 0.00 - 0.00               |
| Lungs                | 0.048 - 0.045             | 0.093 - 0.120             | 0.02 - 0.01               |
| Osteogenic cells     | 0.650 - 0.540             | 0.260 - 0.330             | 0.07 - 0.05               |
| LLI wall             | 0.450 - 0.530             | 0.190 - 0.230             | 0.30 - 0.16               |
| ULI wall             | 0.290 - 0.290             | 0.150 - 0.190             | 0.27 - 0.15               |
| Red marrow           | 0.160 - 0.110             | 0.110 - 0.120             | 0.02 - 0.01               |
| Spleen               | 0.046 - 0.065             | 0.110 - 0.150             | 0.02 - 0.01               |
| Ovaries              | 0.150                     | 0.180                     | 0.04                      |
| Testes               | 0.200                     | 0.120                     | 0.02                      |
| Uterus               | 0.150                     | 0.190                     | 0.19                      |
| Urinary bladder wall | 0.430 - 0.400             | 0.240 - 0.320             | 1.19 - 1.27               |
| Total body           | 0.200 - 0.150             | 0.110 - 0.140             | 0.02 - 0.02               |
| Effective dose       | 0.200 - 0.190             | 0.130 - 0.160             | 0.03 - 0.10               |

| Table 1 | Radiation Dose Estimates for | Adult Humans based on | Dosimetry Studies in Rat a | nd Dogs |
|---------|------------------------------|-----------------------|----------------------------|---------|
|---------|------------------------------|-----------------------|----------------------------|---------|

API=active pharmaceutical ingredient; LLI=lower large intestine; F=female; M=male; mSv=millisievert; ULI=upper large intestine.

In *bold italic*, organs with the highest absorbed dose coefficients

Currently, no data from human use are available under controlled study conditions. Under full medical responsibility of Professor Baum (see Section 1.4).<sup>CCI</sup> subjects with either pancreatic adenocarcinoma (n=6), <sup>CCI</sup> have been administered a low dose of <sup>177</sup>Lu-3BP-227 (approximately 1 gigabecquerel (GBq)). Among them, five pancreatic ductal adenocarcinoma (PDAC) subjects <sup>CCI</sup> have shown a good tumour uptake after imaging without safety concern. <sup>CCI</sup> subjects had further administration of 5.1 to 7.5 GBq with therapeutic intention (see Section 1.4).

# 1.3.5 Pharmacodynamic Effect In Vivo Study

The potential efficacy of <sup>177</sup>Lu-3BP-227 has been investigated in an NTSR1-expressing tumour model in mice (HT29 xenograft model in nu/nu mice). <sup>177</sup>Lu-3BP-227 was administrated at two different doses (100 and 160 MBq) 7 days after tumour cells inoculation in mice (HT29 human colon carcinoma xenograft implanted in nude mice) and one control mice group received a vehicle. The results showed a dose-dependent tumour growth control for 3 or 5 weeks, without any overt signs of toxic effects [2].

Please see investigator brochure (IB) for more detailed information.

### 1.4 Clinical Data

In a preliminary salvage therapeutic administration (compassionate use with non-registered medicinal product), under full medical responsibility of Professor Baum at the Zentralklinik Bad Berka in Germany, hereinafter referred to as "individual clinical treatment use" <sup>CCI</sup> subjects received <sup>177</sup>Lu-3BP-227: Six subjects with PDAC, <sup>CCI</sup>

### [3].

Out of the six subjects with PDAC who were imaged with <sup>177</sup>Lu-3BP-227, five subjects had a good uptake in primary tumours and/or metastases following administration of 1.2 or 1.5 GBq

<sup>177</sup>Lu-3BP-227 and one subject imaged with 1.5 GBq <sup>177</sup>Lu-3BP-227 had no uptake. These results are in line with the 75% of the PDAC expressing NTSR1 as described in the literature [4, 5]. Due to the advanced disease stage, most of the subjects did not have the opportunity to receive further administration in a range of therapeutic activity; three out of the five subjects with NTSR1-expressing tumours had post-imaging administrations.

Dosimetry calculations are available for three subjects. Subjects received diagnostic doses of 1.2 to 1.5 GBq. Only Subject 3 received therapeutic activities (of 6.4 GBq, 7.5 GBq and 5.5 GBq given intra-peritoneally). Kidneys were identified as dose-limiting organs, with absorbed doses ranging from 0.48 Gray (Gy)/GBq to 1.42 Gy/GBq. However, none of the reported subjects received a dose to the kidney that exceeded 23 Gy (highest absorbed renal dose of 22 Gy for a cumulative activity of 20.9 GBq). The specific absorbed doses to the bone marrow was low (0.09 to 0.10 Gy/GBq), as well as the absorbed dose in liver, GI tract, thyroid and urinary bladder, ranging from 0.06 Gy/GBq to 0.09 Gy/GBq. Moreover, the cumulative absorbed doses in these organs were below the organ radiation dose limits of 2 Gy, 35 Gy, 45 Gy, 45 Gy and 60 Gy respectively (ICRP publication 41).

Of interest, following initial administration of 1.5 GBq <sup>177</sup>Lu-3BP-227 to assess the tumour uptake and calculate the organ dosimetry, one of the subjects, Subject 3 who received three therapeutic administrations of 6.4, 7.5 and 5.5 GBq of <sup>177</sup>Lu-3BP-227 was considered a partial responder (per Response Evaluation Criteria in Solid Tumours (RECIST) and positron emission tomography (PET) Response Criteria in Solid Tumours (PERCIST) criteria). The administered cumulative activity of 20.9 GBq in this subject led to a total organ dose of 22.0 Gy to the kidneys. The haemoglobin of this subject remained low during and after treatment, with a short and reversible episode of Grade 2 anaemia before the last administration.



Overall, no important safety concern was raised during administration during the individual clinical treatment use.

# **1.5** Rationale for the Study

The rationale for this phase I/II study is to use the NTSR1 overexpression as target to deliver screening and therapeutic radioactivity to tumour cells. Therefore, tumours with specific overexpression of NTSR1 will constitute the target tumour types for <sup>177</sup>Lu-3BP-227.

Previous radiopharmaceutical compounds (e.g. <sup>177</sup>Lu-DOTATATE, <sup>177</sup>Lu-DOTATOC) have been demonstrated as safe and effective in neuroendocrine tumour subjects [3, 7, 8]. This treatment principle can be applied to <sup>177</sup>Lu-3BP-227 for the treatment of cancers with increased expression of NTSR1. <sup>177</sup>Lu-3BP-227 has been shown to be effective in a NTSR1-expressing xenograft model inhibiting tumour growth. Dosimetry data showed a low accumulation of radioactivity in the organs and an acceptable tumour to tissue radioactivity uptake. Considering the favourable toxicity profile in vitro and in vivo through targeting the NTS/NTSR1 complex and the first clinical evidence from an individual clinical treatment use under full medical responsibility of Professor Baum (see Section 1.4), radiotherapy with <sup>177</sup>Lu-3BP-227 may be particularly attractive as it may offer a treatment option, beyond the standard-of-care, for cancers such as PDAC, colorectal cancer (CRC), ES, gastric cancer (GC) and cancers of the head and neck, where there are still high unmet clinical and therapeutic needs.

Even though promising safety data were collected from the individual clinical treatment use (compassionate use with non-registered medicinal product), the safety, tolerability and efficacy of <sup>177</sup>Lu-3BP-227 treatment for cancers expressing NTSR1, needs to be assessed in a well-designed prospective clinical study.

The dose escalation part of the study has been designed to primarily investigate the safety, tolerability, dosimetry and preliminary antitumour activity of <sup>177</sup>Lu-3BP-227 following fractionated i.v. administrations in subjects with unresectable, locally advanced or metastatic cancers expressing NTSR1. In radiopharmaceutical studies, this safety assessment also includes dosimetry studies to evaluate the radioactive exposure of organs. To optimise the benefit-risk ratio, it is essential to identify the proper target population for therapy. In this study, the target population will be identified by assessing the tumour uptake following the screening administration of <sup>177</sup>Lu-3BP-227. Samples will be collected for biomarker exploratory analyses to investigate associations between biological and clinical parameters (e.g. better characterisation of potential responders, response to treatment, tolerability and safety).

For phase I (dose escalation), a statistical Bayesian modelling approach [9] may be implemented to produce a more precise dose-toxicity response curve and predict subsequent activity levels. The objective of the model is to predict activity ranges to be administered that, in combination with the dose escalation rules, remain below the maximum tolerated cumulative activity (MTCA) level in subjects.

Upon termination of dose escalation and/or determination of MTCA by the safety review committee (SRC), and based on the assessment of the preliminary antitumour activity, cohorts of subjects will be studied to further characterise safety and efficacy of <sup>177</sup>Lu-3BP-227. Antitumour activity of the <sup>177</sup>Lu-3BP-227 is expected to be observed from cohort 2 of the dose escalation part. In case of a weak antitumour activity across all indications and dose cohorts, it is anticipated that further clinical investigations (expansion part) will be needed with optimal radioactivity and small molecule mass determination as well as administration schedule determination in selected indications. The activity levels will be decided based on the safety and preliminary antitumour effect observed in the dose escalation part. The selection of the small-molecule/radionuclide ratio will be driven by data from nonclinical studies that will be conducted in parallel with the dose escalation part. Finally, the targeted populations might be reviewed in light of the tumour uptake, safety, tolerability and preliminary antitumour effects observed during dose escalation.

If a strong antitumour activity signal is observed during phase I, cohorts will be enrolled in the phase II part of the protocol immediately after the end of the dose escalation or dose expansion parts to assess the efficacy of <sup>177</sup>Lu-3BP-227.

In case of medium objective response rate (ORR), a substantial clinical benefit would be observed in selected indications. Based on the actual knowledge built from preclinical studies, an individual clinical treatment use (see Section 1.4) and tissue/cancer expression of NTSR1, it is planned to rapidly enrol cohorts in PDAC and CRC in a Simon Two-Stage design. Pancreatic ductal adenocarcinoma is the most common form of pancreatic cancer and constitutes 90% of all cases. It is a highly aggressive tumour with an overall 5-year survival rate of less than 5%, [10] and between 1% to 3% after the second stage [11]. Pancreatic ductal adenocarcinoma, the worst of all malignancies, is predicted to be the second leading cause of cancer-associated mortality within the next 5 to 10 years [12, 13]. This poor overall survival (OS) for subjects with pancreatic cancer is mainly due to a lack of biomarkers to enable early diagnosis and a lack of prognostic markers that can inform decision making and facilitate personalised treatment and an optimal clinical outcome [14]. The overexpression of NTSR1 by PDAC tumoural tissue offer a new promise to demonstrate the safety and efficacy of a NTSR1 targeted radiopharmaceutical such as <sup>177</sup>Lu-3BP-227. This hypothesis is well supported by the results of

preclinical data (see Section 1.7) as well as by clinical data from an individual clinical treatment use (see Section 1.4).

Metastatic CRC (mCRC) constitutes another pressing unmet need, when current available treatment options are exhausted. Colorectal cancer is currently the second most commonly diagnosed cancer in Europe and the second leading cause of cancer-related death in the United States [15, 16]. Despite significant progress in the treatment of mCRC during the past two decades, the prognosis of subjects with mCRC remains poor, with a 5-year survival rate of approximately 13% [17, 18]. Although a subset of subjects with liver and/or lung isolated disease is potentially curable with surgery, for other subjects with mCRC, treatment is palliative and generally consists of systemic chemotherapy [19, 20]. <sup>177</sup>Lu-3BP-227 could be investigated in those subjects who have received prior lines of standard-of-care chemotherapy/treatment and have no further suitable treatment options and their tumour tissue overexpresses NTSR1, knowing that the percentage of CRC tumours expressing NTSR1 is very high, with high degree of expression (see Section 1.7). Moreover, the nonclinical data generated from an NTSR1 expressing tumour model in mice (HT29 xenograft model in nu/nu mice) are supportive of further investigating <sup>177</sup>Lu-3BP-227 in mCRC (see Section 1.3.5).

In case of high ORR, it would suggest that the NTSR1 target is the main driver for antitumour activity. In this particular situation, a basket single-arm trial will be conducted to further characterise the efficacy of <sup>177</sup>Lu-3BP-227 in the context of the multiple indications. This innovative approach would provide rapid access of the investigational medicinal product (IMP) to subjects with great medical need.

### **1.6** Selection of Dosage

The radioactivity range (5 to 15 GBq) applied in this study is derived mainly from previous experiences with <sup>177</sup>Lu-DOTATATE [21], although some caution should be applied as it refers to other radiolabelled compounds in other indications. In addition, data from an individual clinical treatment use (see Section 1.4) of <sup>177</sup>Lu-3BP-227 support the dose range selection and is endorsed by the dosimetry studies performed with the IMP in rats and dogs.

In the phase I dose escalation part of this first in human (FIH) study, 5 GBq fractionated into two administrations ( $2\times2.5$  GBq) will be used as starting radioactivity. In radiotherapy, fractionation of the cumulative activity is done for safety and tolerability reasons. Based on the biodistribution data obtained in nontumour bearing rats and dogs and consequent allometric calculation for human exposure and taking the most conservative approach, administration of 5 GBq leads to an organ dose of 0.8 Gy for the bone marrow and 1.4 Gy for the kidneys. These organ doses are more than 50% below the organs dose limits of 2 Gy and 23 Gy, respectively. In an individual clinical treatment use (see Section 1.4), four subjects received single administration of 5 GBq without reporting of adverse events (AEs) >Grade 2.

During dose escalation, it is planned to not exceed the cumulative activity of 15 GBq activity  $(2 \times 7.5 \text{ GBq})$ . However, if the MTCA is not reached and if limiting organ dose levels are not exceeded, an additional cohort with three administrations at 7.5 GBq may be added, leading to a cumulative activity of 22.5 GBq. The dosage used in the dose expansion part will be based on the results obtained from the dose escalation part and from the modelling and simulation approach. Similarly, results generated during the dose expansion part will drive the dosage that will be used in phase II. In any case, subjects will be administered with cumulative activities that may be either the MTCA determined in the escalation part or a lower dose with near optimal activity.

In this study,  $25 \ \mu g$  (22.1 nmol) of 3BP-227 per 1 GBq <sup>177</sup>Lu will be used. This radioactivity-small-molecule ratio was used in an individual treatment use (see Section 1.4) and yields a specific activity of approximately 45 MBq/nmol, which is routinely achieved with -DOTA-conjugated compounds and is considered clinically acceptable [21, 22].

In Table 2, the safety margins for each species are presented based on the human equivalent dose calculated from body surface area as per Food and Drug Administration (FDA) Guidance for Industry "Estimating the Maximum Safe Starting Dose in Initial Clinical Trials for Therapeutics in Adult Healthy Volunteers" [23]. The small-molecule 3BP-227 has been administered within a nonclinical safety program carried out in rats and dogs following a single administration. The NOAELs determined in the rats and dogs were 2 mg/kg and 1 mg/kg, respectively. Compared to the animal studies, a wide safety margin is observed (based on the most sensitive species, a safety factor margin of more than 300 for the starting dose activity and more than 100 for the targeted maximum radioactivity). At the dose range of 3BP-227 used in this study, no pharmacological effect is expected.

| Study             | Dose/activity                        | HED of 3BP-227 | Estimated safety margin |
|-------------------|--------------------------------------|----------------|-------------------------|
|                   |                                      |                | for 3BP-227 [a]         |
| Study phase I/IIa | Starting dose:                       |                |                         |
|                   | 2.5 GBq of <sup>177</sup> Lu-3BP-227 |                |                         |
|                   | Corresponding to:                    |                |                         |
|                   | • A total of 62.5 µg of              |                |                         |
|                   | 3BP-227 (2.5×25µg) per               |                |                         |
|                   | administration                       |                |                         |
|                   | • 1 μg/kg of 3BP-227 per             |                |                         |
|                   | administration                       |                |                         |
|                   | Maximal dose:                        |                |                         |
|                   | 7.5 GBq of <sup>177</sup> Lu-3BP-227 |                |                         |
|                   | Corresponding to:                    |                |                         |
|                   | • A total of 187.5 μg of             |                |                         |
|                   | 3BP-227 (7.5×25 μg) per              |                |                         |
|                   | administration                       |                |                         |
|                   | • <b>3.1 μg/kg</b> of 3BP-227 per    |                |                         |
|                   | administration                       |                |                         |
| Single dose rat   | NOAEL (3BP-227)=2 mg/kg              | 0.322 mg/kg    | Starting dose:          |
| Study             |                                      |                | 322 times below NOAEL   |
| 763.321.4274      |                                      |                | Maximal dose:           |
|                   |                                      |                | 104 times below NOAEL   |
| Single dose dog   | NOAEL (3BP-227)=1 mg/kg [b]          | 0.556 mg/kg    | Starting dose:          |
| Study             |                                      |                | 556 times below NOAEL   |
| 763.323.5173      |                                      |                | Maximal dose:           |
|                   |                                      |                | 179 times below NOAEL   |

| Table 2 | Estimated Safety Margin in Study/Species based on Human Equivalent Dose Calculated Based |
|---------|--|
|         | on Body Surface Area   |

HED=human equivalent dose; NOAEL=no observed adverse events level.

a based on HED and on body surface area (divided by 1.8 for dogs and 6.2 for rats)

The calculations are based on the following basis: 1 GBq/25 µg 3BP-227 and human body weight of 60 kg

b 2 mg/kg 3BP-227 was tested in study 763.323.5173 and was considered within the Maximum Tolerated Dose due to slight transient increase in QT interval. A safety pharmacology study tested 1 mg/kg 3BP-227 did not show any adverse cardiovascular effect. The NOAEL of 1 mg/kg 3BP-227 is derived from the combination of these two studies.

# Rationale for use of <sup>177</sup>Lu labelling for Screening Purposes and as a Threshold for Radiodiagnosis

<sup>177</sup>Lu is a beta (β)- and gamma (γ)-emitting radionuclide. It is a medium-energy β emitter with a maximum energy of 0.498 MeV and maximum and mean soft-tissue penetration depths of 1.7 mm and 0.23 mm, respectively. The  $t_{1/2}$  is 6.7 days (159.5 hours). <sup>177</sup>Lu also emits low-energy γ-rays at 208 and 113 KeV with 11% and 6% relative abundance, respectively, which allows scintigraphy and subsequent dosimetry with the same product. The radiochemical physical characteristics of <sup>177</sup>Lu are described in Table 3 [21].

<sup>177</sup>Lu presents the advantage of emitting just enough  $\gamma$  radiation for quantitative imaging necessary for dosimetry calculation while the  $\beta$  radiation induces the therapeutic effect.

For imaging purposes, the  $\gamma$  emission of <sup>177</sup>Lu is accompanied by the additional radiation burden of  $\beta$  emission of <sup>177</sup>Lu. Therefore, when radiotherapy is not required, the administration of <sup>177</sup>Lu may deliver an undesirable radioactivity to the subject. However, the theranostic potential of <sup>177</sup>Lu outweighs the above disadvantage particularly in advanced malignant tumours and avoids the uncertainties that may occur from the use of different isotopes with different kinetics that could result in less accurate dosimetry planning.

| Physical half-life t <sup>1</sup> / <sub>2</sub> | 6.65 days         |
|--|-------------------|
| Decay product                                    | <sup>177</sup> Hf |
| Maximum β-particle energy                        | 0.498 MeV         |
| Mean β-particle energy                           | 0.133 MeV         |
| Median tissue penetration                        | 0.23 mm           |
| Max. tissue penetration                          | 1.7 mm            |
| Main gamma emission lines                        | 113 keV (6%)      |
|  | 208 keV (11%)     |

 Table 3
 Radiochemical Physical Characteristics of <sup>177</sup>Lutetium

The administration of 1.00±0.1 GBq <sup>177</sup>Lu-3BP-227 is considered adequate to ensure a good uptake at tumour level for dosimetry based on a similar application scheme used by Wild et al. [24] for the evaluation of <sup>177</sup>Lu-DOTA-JR11 and will thus be used in this study for screening purpose, after all other required screening assessments have been completed.

A lower radioactivity administration would lead to longer acquisition time and a poor diagnostic signal (with an increase in signal noise to background ratio) associated with the risk of missing subjects who would be eligible for participation in the study and could benefit from the treatment.

### **1.7 Population to be Studied**

In this FIH study, the subject population enrolled to receive the IMP will be restricted to PDAC, CRC, GC, GIST, squamous-cell carcinoma of head and neck (SCCHN) and ES, which are unresectable, locally advanced or metastatic and expressing NTSR1. Due to the late metastatic or advanced stage of the disease and progression or failure of response to standard-of-care treatments, this subject population is expected to have a reduced life expectancy.

Tumours/metastases expressing NTSR1 will be identified and documented through the lesion uptake of <sup>177</sup>Lu-3BP-227 (at low dose) during the screening period. Subjects who participated in the imaging study, Study D-FR-01087-002, with <sup>111</sup>In-3BP-227 (also called <sup>111</sup>In-IPN01087) and who have an uptake of <sup>111</sup>In-3BP-227 in tumour lesions that is more avid than in the nontumoural surrounding tissue based on whole body imaging (planar scintigraphy), as judged by the investigator, can also be considered for enrolment in this study, provided that they fulfil all other inclusion criteria and do not meet any of the exclusion criteria and were imaged with <sup>111</sup>In-3BP-227 before the first treatment administration (therapeutic dose) of <sup>177</sup>Lu-3BP-227.

CCI



In addition, published data also support these NTSR1 expression profiles in the targeted indications [4, 5, 25, 26, 27]. In the PDAC population, the NTSR1 was found expressed in 73% (8/11) of primary tumours and 63% (14/23) of the metastases. Interestingly, the expression of NTSR1 in the membrane compartment was only observed in tumour tissue (85%; 34/40) and never in normal tissue. In addition, a strong expression of NTSR1 messenger ribonucleic acid (mRNA) was found in 76% (19/25) of the cases of colorectal adenocarcinoma (resection material). Protein overexpression has been documented by immunochemistry in 71% (149/210) of GC samples. For ES, expression of NTSR1 in the membrane compartment has been found in 65% (11/17) of the resection tissue material [6].

The five study populations will be enrolled in the phase I dose escalation part. Based on tumour uptake, safety and preliminary antitumour effect (ORR), the number of study populations might be revised for the potential phase I expansion cohorts and phase II through a protocol amendment.

# 1.8 Known and Potential Risks to Human Subjects

This is the FIH study conducted with <sup>177</sup>Lu-3BP-227 under controlled conditions. The study will primarily evaluate the safety and tolerability of <sup>177</sup>Lu-3BP-227 but also its potential antitumour effect in subjects with advanced cancers expressing NTSR1. With the exception of an individual clinical treatment use (see Section 1.4), <sup>177</sup>Lu-3BP-227 has never been tested in humans, but has shown no relevant toxicological effects in the animal studies conducted with high safety margins. However, it cannot be ruled out that <sup>177</sup>Lu-3BP-227 could have adverse effects in the human population, which have not been detected in animal studies.

As this is the FIH study, the function of organs identified at risk will be checked as part the of exclusion criteria. Based on the current level of knowledge, those organs are bone marrow, kidney, large intestine, liver and urinary bladder. In conventional fractionated external radiotherapy, the experience is that an absorbed dose of 23 Gy to the kidney gives a 5% expected risk of nephrotoxicity within 5 years [28, 29, 30]. For the bone marrow, a maximum absorbed dose (MAD) of 2 Gy gives approximately a 2% risk of developing leukaemia as a long-term toxicity. In this study, the well-established 2 Gy for the bone marrow MAD and the more conservative approach with 23 Gy as MAD for the kidney will be applied.

Safety measures have been taken into consideration to minimise the risk. Each subject recruited across the sites may be hospitalised for 24 hours following administration for observation at the discretion of the investigator. The level of radioactivity will be monitored until it has fallen to safe levels for discharge, to protect medical personnel and relatives. Depending on the local

regulation, the subject will be hospitalised either in the nuclear medicine or regular oncology ward.

The participating subjects will be closely monitored during the study and during the long-term follow-up period until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first. Subjects may be withdrawn from the study at any time, if judged to be in their best interest by the investigator or upon the subject's wish. The study will be carried out in specialised clinical centres with nuclear medicine and medical oncology specialists.

An SRC will review the safety and dosimetry data on a continuous basis (see Section 4.2.1.4).

Subject-specific dosimetry will be performed on a regular basis for up to 96 hours after each administration to describe the uptake by the tumour and organs identified at risk over the entire course of treatment. The cumulative organ doses of kidney, bone marrow and liver will be monitored on an ongoing basis as for the other organs identified at risk. If a previous cumulative radioactive dose indicates that the organ limit will be exceeded with the next cycle, the activity of the next cycle will be reduced.

Detailed criteria, based on the percentage of subjects experiencing dose limiting toxicities (DLTs), have been carefully defined to provide guidance on the next radioactive dose selection.

The risks associated with this study are considered adequately elucidated and controlled well by planned cautionary measures in the study design and the target population as well as with the potential benefit of the treatment.
### **2 STUDY HYPOTHESIS**

The present study consists of two parts that aim to test the following hypotheses:

### 2.1 Phase I

<sup>177</sup>Lu-3BP-227 is sufficiently well-tolerated to permit clinical investigation in phase II.

# 2.2 Phase II

<sup>177</sup>Lu-3BP-227 yields higher objective response rates in subjects who have NTSR1 expressing cancers that are unresectable, locally advanced or metastatic based on RECIST version 1.1 by central review and as compared with the historical ORR obtained by current standard-of-care treatment for each tumour type.

### **3 STUDY OBJECTIVES**

### **3.1 Primary Objectives**

### 3.1.1 Phase I

To establish the safety and tolerability of fractionated i.v. administrations of <sup>177</sup>Lu-3BP-227 in subjects with unresectable, locally advanced or metastatic cancers expressing NTSR1.

### 3.1.2 Phase II

To estimate ORR of fractionated i.v. administrations of <sup>177</sup>Lu-3BP-227 in subjects with unresectable, locally advanced or metastatic cancers expressing NTSR1.

### **3.2** Secondary Objectives

#### 3.2.1 Phase I

- (a) To determine the whole-body distribution of <sup>177</sup>Lu-3BP-227 and PK of both <sup>177</sup>Lu-3BP-227 and 3BP-227.
- (b) To determine the radiation dosimetry of <sup>177</sup>Lu-3BP-227 (organ exposure to radiation).
- (c) To describe the preliminary antitumour activity of  $^{177}$ Lu-3BP-227.

### 3.2.2 Phase II

- (a) To further evaluate the safety profile of <sup>177</sup>Lu-3BP-227 at the radioactivity recommended by the phase I results.
- (b) To further assess the response to treatment with <sup>177</sup>Lu-3BP-227 using RECIST version 1.1 and/or PERCIST version 1.0 criteria.
- (c) To further characterise the whole-body distribution and dosimetry of <sup>177</sup>Lu-3BP-227 and PK of both <sup>177</sup>Lu-3BP-227 and 3BP-227.
- (d) To describe the influence of <sup>177</sup>Lu-3BP-227 on the health-related quality of life of treated subjects.

#### **3.3 Exploratory Objectives**

#### 3.3.1 Phase I/II

- (a) To explore the correlation between the tumour uptake of <sup>177</sup>Lu-3BP-227 and the NTSR1 expression on tumours.
- (b) To explore renal safety by measuring urinary specific biomarkers.
- (c) To evaluate the tumour microenvironment, transcriptomics and other markers of interest for the disease through assessment of tumour biopsies.
- (d) To explore genomic alterations in circulating cell-free DNA (cfDNA) and in germline DNA.
- (e) To collect gene mutation status for correlation with clinical outcome.
- (f) To collect biobank samples for future analysis of circulating markers (optional; additional informed consent required).
- (g) To generate a model integrating PK, dosimetry, antitumour activity and safety data.

### 4 INVESTIGATIONAL PLAN

#### 4.1 Overall Study Design and Plan

This is a multicentre, open-label phase I/II study of <sup>177</sup>Lu-3BP-227 in subjects with unresectable, locally advanced or metastatic solid tumours expressing NTSR1 who have progressed after their available standard-of-care treatment options and/or are deemed suitable for the treatment with <sup>177</sup>Lu-3BP-227 as per the investigator's clinical assessment and/or their

individual disease state. The study consists of a phase I dose-escalation (and potential expansion cohorts) and a phase II assessing the efficacy of <sup>177</sup>Lu-3BP-227 in subjects with unresectable, locally advanced or metastatic solid tumours expressing NTSR1.

### 4.2 Phase I

### 4.2.1 Dose Escalation

Phase I will consist of a radioactivity escalation part with the objective of determining the MTCA. For phase I, subjects enrolled will have unresectable, locally advanced or metastatic tumours expressing NTSR1 originating from either the pancreas (PDAC), colon and rectum (CRC), stomach (GC), gastrointestinal stromal tumours (GIST), head and neck region (SCCHN) or bone (ES).

The study screening period includes a screening administration of <sup>177</sup>Lu-3BP-227 to assess the tumour uptake of each subject by whole body scan (planar scintigraphy, 1 or 2 timepoint(s) at the investigator's discretion) and optional single photon emission computed tomography (SPECT)/CT scans (up to 2 at the investigator's discretion). Only subjects with a tumour uptake higher than nontumoural surrounding tissue (based on investigator's decision) will be deemed eligible for treatment. The screening period will last 3 weeks but can be extended by up to 2 weeks if this is required for logistical reasons. Subjects who participated in the imaging study, Study D-FR-01087-002, with <sup>111</sup>In-3BP-227 (also called <sup>111</sup>In-IPN01087) and who have an uptake of <sup>111</sup>In-3BP-227 in tumour lesions that is more avid than in the nontumoural surrounding tissue based on whole body imaging (planar scintigraphy), as judged by the investigator, can also be considered for enrolment in this study, provided that they fulfil all other inclusion criteria and do not meet any of the exclusion criteria and were imaged with <sup>111</sup>In-3BP-227 before the first treatment administration (therapeutic dose) of <sup>177</sup>Lu-3BP-227.

During the treatment phase, it is planned that each subject will receive <sup>177</sup>Lu-3BP-227 fractionated into two i.v. administrations separated by 4 to 5 weeks (+4 weeks in case of reversible AEs) (Figure 1). The safety evaluation will be done over 4 weeks between Cycle 1 and Cycle 2 and over 6 weeks between Cycle 2 and end of core trial (EOCT). During this period, subjects will have repeated imaging for the calculation of individual dosimetry data and will be monitored for signs of toxicity.

Tumour response will be assessed on CT or MRI, as well as with <sup>18</sup>F-FDG-PET scans at the EOCT visit, (corresponding to 6 weeks after the second administration).

Subjects may receive up to four additional administrations of <sup>177</sup>Lu-3BP-227 after the EOCT, if they have clinical benefit and an acceptable tolerability profile and if the organ dose limits are not exceeded. The safety data evaluation will be conducted by the sponsor. The decision to administer additional cycles is based on the investigator's judgement and subject's discretion and must be discussed with and agreed upon by the sponsor. The clinical benefit will be defined as complete responder (CR), partial responder (PR) or stable disease or based on their biochemical response.

A long-term follow-up period will start after the EOCT, early discontinuation (ED) or end of additional cycles (EOAC) visit and subjects will be followed up every 3 months ( $\pm 2$  weeks) until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first.



Figure 1 Phase I Study Design Scheme

Note: Population screened includes subjects with unresectable disease.

<sup>1</sup>Additional cycles are optional depending on the clinical benefit observed in the subjects during the core trial, as well as on the toxicities and organ absorbed doses.

#### 4.2.1.1 Definition of the MTCA

Several radioactivity amounts of <sup>177</sup>Lu-3BP-227 are planned to be tested according to an adaptive radioactivity escalation plan. Each subject will only participate in one part of the protocol. The purpose of the radioactivity escalation part is to determine the MTCA. If not feasible, the maximum tolerated single activity (MTSA) or the maximum administered cumulative activity (MACA) will be determined according to the situation below:

- The MTCA is defined as the maximum tolerated cumulative activity that may be administered following fractionated i.v. administrations of at least 4 weeks apart, so that:
  - No more than 33% of the subjects experience a DLT during Cycles 1 and/or 2 (see DLT definition in Section 4.7.1.1), and
  - The cumulative radiation in each target organ does not exceed the acceptability limits (see Section 4.7.1).
- The MTSA is defined as the highest single radioactivity that can be given so that no more than 33% of the subjects experience a DLT during Cycle 1. The MTSA will be determined in case of unacceptable toxicity seen after Cycle 1.
- The MACA will be determined if the MTCA is not reached during the dose escalation part.

#### 4.2.1.2 Cohorts Description

The dose escalation part will be conducted over five cohorts from a starting cumulative activity of 5 GBq up to a maximum cumulative activity of 15 GBq, fractionated into two administrations of 2.5 GBq and 7.5 GBq respectively.

The size of the cohorts will be three to five treated subjects, to secure a minimum of three evaluable subjects per cohort who completed 2 cycles of treatment. A cohort will be considered as completed once three subjects of the cohort complete Cycle 2 or early discontinue during Cycle 2 (except for cohort 1, see *Proceeding to the Next Cohort* described in Section 4.2.1.4). If a subject replacement is needed to complete the cohort enrolment, see Section 4.7.1.3 for replacement rules.

For safety reasons, the first three subjects of a cohort will be administered sequentially, one week apart to allow a sufficient observation period for any adverse reaction.

#### 4.2.1.3 Dose Escalation Mode

Dose escalation is planned to proceed in two steps, with two different increments for the radioactivity escalation. In the first step, in order to minimise the number of subjects receiving subtherapeutic activity, the inter-cohort radioactivity escalation will be performed by increment of 3 GBq (1.5 GBq per administration), up to the first study drug related treatment-emergent AE (TEAE) (Grade 2 or higher, except hair loss) or until a DLT occurs. Afterwards, the maximum increment will be reduced to 2 GBq (1 GBq per cycle) for all subsequent cohorts if dose escalation is allowed. However, even if no Grade 2 or higher AE are observed up to a cumulative activity of 11 GBq, radioactivity will be increased by 2 GBq (1 GBq per cycle).

Table 5 illustrates the escalation plan based on the escalation scheme defined above. However, if DLTs are reported, a statistical Bayesian modelling approach will be implemented to produce a more precise dose-DLT curve to guide the dose selection (see details in Attachment 5, Section 19.5) and predict an activity not exceeding the MTCA/MTSA that could be tested in the next cohort.

| Planned<br>radioactivity<br>escalation | Cohort 1  | Cohort 2 | Cohort 3  | Cohort 4  | Cohort 5  |
|--|-----------|----------|-----------|-----------|-----------|
| Cumulative                             | 5 GBq     | 8 GBq    | 11 GBq    | 13 GBq    | 15 GBq    |
| Per cycle                              | 2.5 GBq/C | 4 GBq/C  | 5.5 GBq/C | 6.5 GBq/C | 7.5 GBq/C |

 Table 5
 Cohorts and Radioactivity Escalation Plan

C=cycle; GBq=gigabecquerel.

Once the dose escalation has been completed, the MTCA level may be repeated in a last cohort (see Section 4.2.1.4 for stopping rules and Section 4.2.1.1 for MTCA definition). If the MTCA is not reached and if limiting organ dose levels are not exceeded at the highest planned activity ( $2 \times 7.5$  GBq) and if no individual withdrawal criteria are met, an additional cohort (Cohort 6) with three administrations at the highest single administration level may be added ( $3 \times 7.5$  GBq).

#### 4.2.1.4 Safety Review Committee

The SRC will review the safety and radiation exposure data and jointly decide with the sponsor whether to proceed with the enrolment of the next cohort dose level and the second dosing cycle within a cohort. For this purpose, the SRC will be composed of all phase I investigators, who have treated at least one subject in the study cohort with study medication, one independent expert, a dosimetry expert evaluating the dosimetry data of the study (from a centralised dosimetry centre) and at least one sponsor representative. Regular meetings will be held. A specific charter will be developed to define roles and responsibilities, as well as the dataset to be reviewed by the SRC and the review timepoints.





A=administration of study compound; EOCT=end of core trial; SRC=Safety Review Committee. SRC1=Decision on start of next cohort SRC2=Confirmation or adjustment of A2 in next cohort

#### Proceeding to the Next Cohort

At the time of radioactivity escalation, the available clinical safety and radiation exposure information will be reviewed and discussed by the SRC to jointly decide with the sponsor if the escalation can proceed as planned.

For each cohort, an SRC meeting will be held after the end of the third subject's Cycle 1. If during that observation period (after the first administration) no DLT occurs and all three subjects can receive the full radioactivity level (RL) as planned for the respective cohort (i.e. all three subjects can receive the second administration), the next cohort can be started with the first administration of the first subject at the earliest 3 weeks after the third subject of the previous cohort has received the second administration. No more than two dose-escalation cohorts will be conducted in parallel (Figure 2). If a DLT occurs after the first administration) of the cohort n.

Due to the possible subtherapeutic activity of the lowest dose, the escalation decision to the planned dose of 4 GBq/cycle in Cohort 2 will be taken if only one subject from Cohort 1 has received the second administration without any safety problems at the end of the second cycle (i.e. at Cycle 2 Day 29, see Figure 2).

#### Proceeding to the Second Administration in a Cohort

For each cohort, a SRC meeting will be held after the end of the third subject's Cycle 2 of the preceding cohort and before the second administration of the first subject in a cohort, (Figure 2) to review all available safety data and radiation exposure from previous cohort(s). The second radioactivity administration will be confirmed based on the rules described in Attachment 5, Section 19.5.

#### Stopping Rules for Dose Escalation

The MTCA is defined in Section 4.2.1.1.

The radioactivity escalation will be stopped as soon as:

- The MTCA and/or MTSA have been defined with good precision; or
- The maximum planned radioactivity of 15 GBq, fractionated into two administrations separated by 4 weeks, is administered without safety concerns.

Transition from Dose Escalation to Other Study Parts

Upon completion of the phase I dose escalation or upon reaching the MTCA and confirmed jointly by the SRC and the sponsor, and in consideration of the accumulated subject data, cohorts of subjects will be studied to further characterise the safety and efficacy of <sup>177</sup>Lu-3BP-227. In the case of acceptable tolerability and evident antitumour activity across all enrolled subjects in phase I, a phase II basket trial design will be utilised to study the antitumour activity of <sup>177</sup>Lu-3BP-227 in subjects with NTSR1 expressing tumours. However, if the antitumour activity is driven by a certain type of tumour, tumour-specific phase II cohort(s) will be initiated utilising an Optimal Simon's Two Stage design (see Section 4.3).

### 4.2.2 Dose Expansion

If safety evaluation and dose schedules of <sup>177</sup>Lu-3BP-227 cannot be fully explored during the phase I dose escalation part, a phase I expansion part will serve to accomplish this objective including, but not limited to, schedules of high loading doses followed by fractionated lower doses or evaluation of <sup>177</sup>Lu-3BP-227 in combination with other antitumoral treatments (to be defined). The expansion part will also serve to clarify any uncertainties of antitumour responses. The number of cohorts and subjects will be determined based on emerging data from the dose escalation part and the modelling and simulation approach, and will be documented as part of a protocol amendment.

### 4.3 Phase II

Phase II study will be conducted either with a basket design trial or indication-specific cohorts with an Optimal Simon's Two Stage design, according to the scenarios described above.

One, two or three further cohorts may be initiated (subject to results emerging from ongoing preclinical studies and antitumour efficacy seen during dose escalation and amending the current protocol) likely to enrol subjects with GC, ES and/or SCCHN.

- The PDAC cohort will enrol approximately 55 subjects and will investigate whether <sup>177</sup>Lu-3BP-227 attains an ORR superior to a clinically accepted historical threshold of current standard-of-care treatment for subjects with unresectable, locally advanced or metastatic disease.
- The CRC cohort will enrol approximately 70 subjects and will investigate whether <sup>177</sup>Lu-3BP-227 attains an ORR superior to a clinically accepted historical threshold of current standard-of-care for subjects with unresectable, locally advanced or metastatic disease.

The current protocol will be amended at the end of the phase I to document the rationale of the phase II design. In any case, the cumulative activity administered during phase II will not exceed the MTCA determined during phase I.

Sample size calculation for cohorts other than PDAC and CRC will be documented as part of a protocol amendment.

### 4.4 Study Endpoints

### 4.4.1 Primary Endpoint

#### 4.4.1.1 Phase I

For the dose escalation, the primary endpoint is MTCA or MACA, if the MTCA is not identified during the dose escalation part. The primary variables used for the MTCA determination will be the incidence of DLTs (as defined in Section 4.7.1.1) and the organ exposure to radiation during two cycles of treatment. The DLT period for the determination of the primary endpoint

starts at the first administration of <sup>177</sup>Lu-3BP-227 to EOCT/ED. Safety evaluation will encompass DLTs, frequency and nature of AEs, abnormal findings from physical examination, vital signs, 12-lead electrocardiogram (ECG), 3-lead ECG Holter, Eastern Cooperative Oncology Group (ECOG) performance status deterioration, clinical laboratory tests (including haematology, blood biochemistry, hormone analysis, urinalysis and pregnancy test) and changes in specific renal safety biomarkers.

In case the phase I dose expansion cohorts are implemented, the primary endpoint will be safety and tolerability measured by the type, severity, expectedness and frequency of AEs.

## 4.4.1.2 Phase II

The primary endpoint is ORR measured by CT or MRI using RECIST version 1.1. Tumour response assessments are performed every 8 weeks or at the time of occurrence of first signs of clinical progression as determined by the investigator.

### 4.4.2 Secondary Endpoints

## 4.4.2.1 Phase I

For biodistribution and dosimetry of <sup>177</sup>Lu-3BP-227, the secondary endpoints are:

- Maximal uptake (%), maximal concentration achieved ( $C_{max}$ ); time post injection to achieve maximal concentration ( $T_{max}$ ); AUC at the target lesions, discernible organs and blood;  $t_{1/2}$  of activity concentrations in blood;
- Highest absorbed dose, specific absorbed dose to the target lesions (Gy/GBq), specific absorbed dose per organ (Gy/GBq) and cumulative absorbed organ doses (Gy).

For PK of 3BP-227, the secondary endpoints are:

• PK parameters including, but not limited to, C<sub>max</sub>, AUC, t<sub>1/2</sub>, CL, V<sub>d</sub>, cumulative amount of unchanged drug excreted into the urine (Ae), renal clearance of the drug from plasma (CL<sub>R</sub>), as measured in plasma and urine at defined timepoints.

For pharmacodynamics, the secondary endpoints are:

- Objective response rate and DCR, as determined by RECIST version 1.1 in subjects who received IMP;
- Progression-free survival (PFS) and OS rates as determined from start of study treatment until occurrence of event and/or end of observation period;
- Evaluation of metabolic tumour response using PERCIST (version 1.0) or practical PERCIST;
- Changes in serum tumour markers relevant and specific to the underlying tumour disease from baseline to EOCT, which is planned 6 weeks after the second <sup>177</sup>Lu-3BP-227 dose administration.

### 4.4.2.2 Phase II

The secondary endpoints are:

- Disease control rate, time to progression (TTP), time to response (TTR), duration of response (DOR) as per RECIST version 1.1;
- Qualitative and quantitative changes in tumour-to-background uptake using PERCIST version 1.0;
- Progression-free survival (PFS) and OS as determined from start of study treatment until occurrence of event and/or 6 and 12 months after start of study treatment;
- Changes in serum tumour markers relevant and specific to the underlying tumour disease from baseline to EOCT;

- Changes in health-related quality of life scores from baseline to EOCT measured by validated and disease-specific questionnaires;
- Safety and tolerability measured by the type, severity, expectedness and frequency of AEs;
- For PK, biodistribution and dosimetry, the endpoints will be similar as for phase I.

## 4.4.3 Exploratory Endpoints

Exploratory endpoints are the same for phase I and II:

- Tumour uptake of <sup>177</sup>Lu-3BP-227 and the correlation with NTSR1 expression on tumour biopsies;
- Tumour microenvironment and other markers of interest (such as NTSR1 expression, Ki67, gene expression and DNA damage) in tumour biopsies taken at baseline, at EOCT visit or at disease progression, whichever occurs earlier;
- Genomic profiling in circulating cfDNA and in germline DNA;
- Gene mutation status in correlation with clinical outcome;
- Specific renal safety biomarkers specific for proximal tubulus toxicity.

### 4.5 Study Duration

For phase I, the maximum duration of subject participation in the core trial is 21 weeks. However, if, according to the investigator, a subject has clinical benefit, the subject may receive up to four additional cycles after the EOCT, provided they have an acceptable tolerability profile and the organ dose limits are not exceeded. This is at the subject's discretion and must be discussed and agreed upon with the sponsor. The clinical benefit will be defined as CR, PR or stable disease (SD) as per RECIST criteria version 1.1 or based on biochemical responses.

For phase I dose expansion and phase II, the study duration will be refined in light of phase I results and recommended treatment regimen (doses and intervals).

In all cases, a long-term follow-up period will start after the EOCT, ED or EOAC visit and subjects will be followed up every three months ( $\pm 2$  weeks) until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first.

### 4.6 Randomisation and Blinding

### 4.6.1 *Method of Randomisation*

### 4.6.1.1 Subject Unique Identifier

At screening, potential subjects will be allocated an 11-digit subject number consisting of:

- The country code (three digits);
- The centre number (three digits);
- The sequential order of entry of the subject at the centre (five digits).

### 4.6.1.2 Randomisation Number

Not applicable for phase I, as this is not a randomised study.

For the phase II, the protocol could be amended to include randomisation.

### 4.6.2 Blinding

Not applicable, this is an open-label study.

### 4.6.3 Code-break

Not applicable, this is not a blinded study.

### 4.7 Discontinuation Rules

### 4.7.1 Individual Discontinuation Rules

In accordance with the Declaration of Helsinki and the applicable country's regulations, each subject is free to withdraw from the study at any time and for any reason without prejudice to their future medical care by the investigator at the institution.

The investigator will be responsible for monitoring subject compliance. A subject can be withdrawn from the study at any time if the investigator or the sponsor determines that the subject is not in compliance with the study protocol.

The investigator has the right to withdraw a subject from the study in the event of concurrent illness, AEs, or other reasons concerning the health or well-being of the subject, or in the case of lack of cooperation.

If any of the following occur, no further treatment will be administered:

- life-threatening toxicities outside of the DLT reporting period
- subject withdraws their consent to further treatment;
- cumulative kidney dose exceeds 23 Gy;
- cumulative bone marrow dose exceeds 2 Gy, as determined by dosimetry of peripheral blood samples and imaging;
- cumulative liver dose exceeds 30 Gy;
- occurrence of a DLT for phase I dose escalation only (see Section 4.7.1.1).

All cases of discontinuation will be discussed between the investigator and the sponsor.

### 4.7.1.1 Definition of Dose Limiting Toxicity (Dose Escalation)

The DLTs are defined for any of the following IMP-related AEs according to National Cancer Institute - Common Terminology Criteria for Adverse Events (NCI-CTCAE) scale version 5.0 [31], that occur during the defined DLT assessment period (from the first administration of <sup>177</sup>Lu-3BP-227 to EOCT/ED):

- Grade 4 neutropenia for seven or more consecutive days;
- Febrile neutropenia or neutropenic infection (defined as a documented infection with neutrophil count decreased Grade 3 or 4);
- Grade 3 or 4 thrombocytopenia (platelet count decreased) with clinically meaningful bleeding (i.e. requiring urgent hospitalisation or transfusion to manage the bleeding);
- Grade 4 thrombocytopenia for seven or more consecutive days;
- Any Grade 3 anaemia (Hb<8.0 g/dL; transfusion indicated) or Grade 4 anaemia (life-threatening consequences; urgent intervention indicated);
- Any Grade 3 or higher laboratory abnormalities on aspartate aminotransferase/alanine aminotransferase (AST/ALT) with accompanying Grade 2 or higher bilirubin (Hy's law);
- any Grade 3 or higher renal injury/toxicity (estimated glomerular filtration rate (eGFR) <30 mL/min/1.73 m<sup>2</sup>);
- Any Grade 3 or higher GI AE, not resolved to Grade ≤2 within 48 hours despite optimal adequate medical management, with the following specifications:
  - Grade 3 nausea, vomiting (inadequate oral caloric or fluid intake; tube feeding, total parenteral nutrition or hospitalisation indicated)
  - Grade 3 diarrhoea (increase of ≥7 stools per day over baseline; hospitalisation indicated; severe increase in ostomy output compared to baseline; limiting self-care activities of daily living (ADL)) or Grade 4 diarrhoea (life-threatening consequences; urgent intervention indicated)

- Grade 3 constipation (obstipation with manual evacuation indicated; limiting self-care ADL) or Grade 4 constipation (life-threatening consequences; urgent intervention indicated);
- Any toxicity related to <sup>177</sup>Lu-3BP-227 resulting in a treatment delay of more than four weeks due to either delayed recovery to baseline or resolution of any AE to Grade  $\leq 2$  (with the exception of alopecia and lymphopenia).
- Grade 5 toxicity (death)

### 4.7.1.2 Procedures for Subject Discontinuation

For subjects who discontinue participation in the study before the planned completion of therapy, it is intended that full end of study (EOS) assessments (see Section ) will be performed and the subjects will be followed-up for AEs until the event or its sequelae resolve or stabilise at a level acceptable to the investigator and the sponsor's clinical monitor or his/her designated representative (see Section 8). If a subject drops out or is discontinued from the study after IMP administration and before planned study completion (either during the Core Trial period or if the subject is receiving additional treatment cycles), an ED visit will take place within 14 days after ED. The investigator determines this timepoint taking into account the expected safety profile and the current clinical status of the subject. This visit will correspond to an EOCT visit (or EOAC visit if the subject dropout during additional cycles). Thereafter, follow-up for survival information will take place every three months. For those subjects with ED, every effort should be made to perform a safety follow-up at the latest 5 weeks after the last IMP administration.

The reason for and date of withdrawal from the study must be recorded in the electronic case report form (eCRF).

If the discontinuation is based on subject decision, every attempt will be made to determine:

- The reason for discontinuation;
- Whether the subjects also decided to withdraw consent for the sponsor to collect and use the data collected up to the discontinuation point.

Data collected prior to subject discontinuation may be kept in study records and shared with the sponsor for study analyses (see Section 13), unless a subject formally specifies their decision to withdraw consent for the use of data collected before discontinuation.

Subjects withdrawn from the study for unacceptable AE(s) will be followed until resolution or stabilisation of the AEs.

#### 4.7.1.3 Replacement Rules

In phase I, a cohort will be considered as completed once three subjects of the cohort complete Cycle 2 or discontinue early during Cycle 2 (except for Cohort 1). Except for Cohort 1, if subjects discontinue for any reason other than a DLT (e.g. disease progression) before end of Cycle 2, they may be replaced.

#### 4.7.2 Discontinuation of a Cohort or a Site or Study Termination

A specific site or a given cohort can be discontinued or the complete study terminated prematurely at any time, if the sponsor judges it necessary for any reason. In that case, all scheduled procedures and assessments for subjects who are still in the study will be performed. Some possible reasons for the closure of a study site may include:

- failure of the investigator staff to comply with the protocol or with the GCP guidelines.
- safety concerns
- inadequate subject recruitment.

In case of premature discontinuation of a site or complete study termination, the sponsor will notify the impacted investigator(s) in writing, depending on the reason(s) for discontinuation, whether the ongoing subjects should undergo the remaining IMP dose administration(s).

### 5 SELECTION OF SUBJECTS

## 5.1 Inclusion Criteria

## 5.1.1 Phase I

Eligible subjects must fulfil all the following inclusion criteria:

- (1) Signed informed consent form prior to all study procedures.
- (2) Aged 18 years or older.
- (3) Histologically or cytologically confirmed unresectable, locally advanced or metastatic disease and has received prior lines of standard-of-care chemotherapy/treatment and has no further suitable treatment options and a documented decision by a multidisciplinary oncology board including a specialist of the concerned pathology.
- (4) Subjects have:
  - (a) PDAC, or
  - (b) CRC (colorectal adenocarcinoma), or
  - (c) GC (gastric adenocarcinoma), or
  - (d) Gastrointestinal stromal tumours (GIST), or
  - (e) SCCHN, or
  - (f) ES.
- (5) Tumours showing:
  - (a) uptake of <sup>177</sup>Lu-3BP-227 (screening formulation) in known primary or metastatic sites as judged by the investigator to be greater than background; or
  - (b) uptake of <sup>111</sup>In-3BP-227 in known primary or metastatic sites (for subjects who participated in Study D-FR-01087-002) as judged by the investigator to be greater than background.
- (6) Measurable disease (based on RECIST version 1.1).
- (7) Criterion 7 is removed by protocol amendment.
- (8) Documentation of progressive disease in the 6 months prior to study start (treatment).
- (9) Eastern Cooperative Oncology Group performance status of 0 or 1 (unless disability is related to surgery in ES and agreed by the sponsor).
- (10) Adequate organ function as evidenced by:
  - (a) Leukocytes  $\geq 3000/\mu L$
  - (b) Absolute neutrophil count  $\geq 1500/\mu L$
  - (c) Platelets  $\geq$ 75,000/µL
  - (d) Hb > 9 g/dL or > 10 g/dL (if history of cardiac disease)
  - (e) Total serum bilirubin  $\leq 2$  times upper normal institutional limits (ULN)
  - (f) AST/ALT  $\leq 2.5 \times ULN$  (or  $\leq 5 \times ULN$ , if subject has liver metastases)
  - (g)  $eGFR \ge 55 \text{ mL/min.}$
- (11) Estimated life expectancy of >3 months.
- (12) Female subjects must not be pregnant or lactating at study entry and during the course of the study and must not become pregnant for at least 6 months following the last study treatment. Women of childbearing potential must agree to use a highly effective method of contraception (see note below).
- (13) Male subjects must not father children during the study and for at least 6 months after the last study treatment and in addition must agree to use a condom for this period to

protect his partner from contamination with the IMP. For males with partners who are of child bearing potential, effective contraception is a combination of male condom with either cap, diaphragm or sponge with spermicide (double barrier methods), but these are not considered to be highly effective. A man is considered to be infertile if he has had bilateral orchidectomy or successful vasectomy. Effective contraception includes a female partner of childbearing potential if she is using highly efficacious contraception (see note below), but the male subject must agree to use a condom to protect his partner as described above.

(14) Must be willing and able to comply with study restrictions and to remain at the clinic for the required time during the study period and willing to return to the clinic for the follow-up evaluation, as specified in the protocol.

*Note: Highly effective methods of contraception that result in a low failure rate (i.e., <1% per* year) when used consistently and correctly include combined (oestrogen and progestogen containing) hormonal contraception associated with inhibition of ovulation (oral, intravaginal, or transdermal), progestogen-only hormonal contraception associated with inhibition of (oral, injectable. or *implantable*), intrauterine ovulation device. intrauterine hormone-releasing system, bilateral tubal occlusion, vasectomised partner (the vasectomised partner has received medical assessment of the surgical success at least 6 months prior to the first study treatment and provided that partner is the sole sexual partner of the female subject of childbearing potential trial participant), or sexual abstinence;

True abstinence, when in line with the preferred and usual lifestyle of the subject, is considered a highly effective method only if defined as refraining from heterosexual intercourse during the entire period of study treatment and for 6 months after the last dose of <sup>177</sup>Lu-3BP-227. The reliability of sexual abstinence needs to be evaluated in relation to the duration of the clinical study and the preferred and usual lifestyle of the subject. Periodic abstinence (e.g. calendar, ovulation, symptothermal, and post-ovulation method) and withdrawal are not acceptable methods of contraception;

Female subject is considered of childbearing potential i.e. fertile, following menarche and until becoming postmenopausal unless permanently sterile. Permanent sterilisation methods include hysterectomy, bilateral salpingectomy and bilateral oophorectomy. A postmenopausal state is defined as no menses for 12 months without an alternative medical cause. A high follicle stimulating hormone (FSH) level in the postmenopausal range may be used to confirm a postmenopausal state in women not using hormonal contraception or hormonal replacement therapy. However, in the absence of 12 months of amenorrhea, a single FSH measurement is insufficient.

For male subjects, acceptable methods of effective contraception include sexual abstinence, successful vasectomy, bilateral orchidectomy and barrier methods (i.e. condom). In addition, male subjects should not donate sperm and female subjects should not donate eggs for at least 2 years after the last study treatment).

### 5.1.2 Phase II

The inclusion criteria for phase II will be revised based on the scenario adopted and indication(s) selected for investigation in phase II. This will be documented as part of a protocol amendment.

### 5.2 Exclusion Criteria (Phase I/II)

Eligible subjects must not have any of the following:

- (1) Prior treatment received:
  - (a) Any antitumour treatment since last documented disease progression

- (b) Any chemotherapy within 3 weeks or nitrosourea within 6 weeks prior to first treatment IMP administration
- (c) Any curative radiotherapy within 4 weeks, or palliative radiotherapy within 7 days prior to first treatment IMP administration
- (d) Any monoclonal antibodies within 4 weeks or tyrosine kinases inhibitors within 2 weeks prior to the first treatment IMP administration
- (e) Any other IMP within 2 weeks prior to first treatment IMP administration, if the previous compound is a mechanism-based molecularly targeted agent whose  $t_{1/2}$  is not well-characterised.
- (2) Brain metastases.
- (3) Nephrectomy, renal transplant or concomitant nephrotoxic therapy putting the subject at high risk of renal toxicity during the study.
- (4) Only nonmeasurable metastatic bone lesions.
- (5) Existing or planned colostomy during study participation.
- (6) Any history of inflammatory bowel disease.
- (7) Any uncontrolled significant medical, psychiatric or surgical condition or laboratory finding, that would pose a risk to subject safety or interfere with study participation or interpretation of individual subject results.
- (8) Clinically significant abnormalities on ECG at screening including corrected QT interval (Fridericia's formula) >450 msec for males or 470 msec for females at screening.
- (9) Previously received external beam irradiation to a field that includes more than 30% of the bone marrow or kidneys.
- (10) Criterion 10 is removed by protocol amendment.
- (11) Any unresolved NCI-CTCAE Grade 2 or higher toxicity (except alopecia) from previous antitumour treatment and/or medical/surgical procedures/interventions.
- (12) Known allergy to IMP or its excipients administered in this study, including imaging contrast media.
- (13) Positive pregnancy test (female subjects).
- (14) Likely to be uncompliant or uncooperative during the study, in the judgment of the investigator.
- (15) Unable to understand the nature, scope and possible consequences of the study, in the judgment of the investigator.
- (16) Sponsor employees or investigator site personnel directly affiliated with this study, and their immediate families. Immediate family is defined as a spouse, parent, child or sibling, whether biological or legally adopted.

Eligibility criteria for phase II will be reviewed as soon as phase I results are available.

## 5.3 Rationale for Specific Inclusion/Exclusion Criteria

Not applicable

## 6 STUDY CONDUCT

### 6.1 Study Schedule of Assessments

The schedule of procedures and assessments is summarised and presented in Attachment 1 Section 19.1, related to the dose escalation part of the phase I. The total volume of blood to be drawn for each subject is detailed in Attachment 2, Section 19.2.

In the eventuality of phase I expansion cohorts and for phase II, the schedule of assessment will be adapted based on phase I results and documented through a protocol amendment.

### 6.2 Study Visits and Procedures

All screened subjects must be identifiable throughout the study. The investigator will maintain a list of all subjects screened, with subject numbers and names, to enable records to be found at a later date, if required. Records should be completed up to the time of premature termination or normal study completion. In the event that a subject is a screen failure or does not receive study treatment, the primary reason will be recorded. Subjects may be re-screened after discussion and agreement with the sponsor. The conditions for re-screening include re-assessment of inclusion criteria (6) and (10) at a later time point.

If the COVID-19 pandemic prevents subjects from coming to the site, subjects can have their study visit assessments performed remotely as judged appropriate by the investigator. This must be discussed with the sponsor before being implemented. In such a case, the investigator will perform a telemedicine visit and will make every effort, where applicable, to contact the subject's general practitioner or specialist physician to ensure all important medical information and safety event(s) occurring since the last visit are collected. Guidance on how to collect protocol-planned assessments will be provided to the investigator in a separate document. This document will be filed in the electronic trial master file. Independent ethics committees (IECs)/institutional review boards (IRBs) will be notified of the changes as applicable locally. Of note, as the adapted visit deviates from the regular protocol plan, the changes will be recorded as protocol deviations related to COVID-19.

#### 6.2.1 Screening

Subjects will attend the clinical unit for an outpatient visit following which they will enter a screening period of 3 weeks prior to the first treatment administration. The screening period can be extended by up to two weeks if this is required for logistical reasons and agreed with the sponsor.

The screening IMP will be administered minimum 1 week prior to Day 1 after all other required screening assessments have been completed.

Subjects may be hospitalised for 24 hours following the screening IMP administration at the discretion of the investigator.

### 6.2.1.1 Informed Consent

After a subject has received explanations and response(s) to their potential questions about the study by the investigator (or designee) and has been given reasonable time to consider study participation, a signed and dated informed consent form will be obtained prior to any study procedures.

Subjects will be offered to participate to an optional research biobanking program. Subjects who agree to participate will be requested to sign a separate informed consent form.

### 6.2.1.2 Screening Procedures

After informed consent is obtained, screened subjects will be allocated a study-specific subject number, which must comply with formatting specifications provided by the sponsor (see Section 4.6.1.1).

During this initial screening visit, and subsequent ones, screening tests and assessments will be performed to check compliance with inclusion/exclusion criteria and study requirements in order to confirm subjects' eligibility prior to enrolment.

The screening procedures/evaluations detailed in Attachment 1, Section 19.1 will be performed during one or several outpatient visits.

The baseline assessment is the assessment done closest to Day 1 of the treatment administration (e.g. haematology and biochemistry, see Attachment 1 Section 19.1).

Evaluations obtained as part of routine medical care or previous assessments performed during (or before) the screening period may be used in place of the study specific evaluations. Subjects will acknowledge and agree to the possible use of this information for the study by giving informed consent.

For subjects who received the screening IMP administration and found ineligible to participate in the study, a physical examination, vital sign monitoring and clinical laboratory test assessment (as described in Attachment 3 Section 19.3) will be performed before discharge. Additionally, a safety follow-up visit or contact, whichever is feasible, will be performed within 5 weeks (+2 weeks) after the IMP screening administration to assess the safety (see Table 7). At least, clinical laboratory tests (e.g. haematology and biochemistry, as described in Attachment 3 Section 19.3), AEs and concomitant medication/therapy will be collected. If an on-site visit is not feasible, the visit may be performed by phone call. The clinical laboratory tests may be done locally.

### 6.2.2 Treatment Period and Additional Cycles for Phase I Dose Escalation

The visit descriptions in this section are relative to the phase I dose escalation part of the study. In case of phase I dose expansion and for phase II, these visits and timepoints will be amended in light of the phase I results.

After each administration of IMP/<sup>177</sup>Lu-3BP-227, subjects may be hospitalised for at least 24 hours for observation at the investigator's discretion. Afterwards, subjects can be discharged when the radioactivity reaches an acceptable level as per local regulation. The subjects must return to the study centre for further evaluations per the schedule of assessment. All other visits will be outpatient. Subjects are to be observed on study and if the disease comes under control are to be kept on study; if the disease does not appear to come under control, the subject is to be removed from study and treated as clinically indicated.

### 6.2.2.1 Early Discontinuation or End of Core Trial Visit

Subjects who complete the core trial, i.e. subjects who receive two treatment administrations and who continue the study until 6 weeks after the second dose administration, will be considered as having completed the study.

Subjects who discontinue the study prior to the EOCT visit are considered as early discontinued subjects. The assessments required at the ED visit are the same as the assessments required at the EOCT visit.

Subjects with ongoing AEs or clinically significant laboratory test abnormalities (as determined by the investigator) will be monitored by the investigator as described in Section 4.7.1.

### 6.2.2.2 End of Additional Cycles

If, according to the investigator, a subject has clinical benefit, the subject may receive up to four additional cycles after the EOCT, provided they have an acceptable tolerability profile and the organ dose limits are not exceeded. This is at the subject's discretion and must be discussed and agreed upon with the sponsor. For these subjects, an EOAC visit will be done 6 weeks after last dose administration.

## 6.2.2.3 End of Study

End of study (EOS) is defined as the last visit or last scheduled procedure shown in the Schedule of Assessment for the last active subject in the study.

### 6.2.3 Long-term Follow-up

For both phase I and phase II, a long-term follow-up period will start after the EOCT, ED or EOAC visit and subjects will be followed up every 3 months ( $\pm 2$  weeks) until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first. During this period, if subjects do not receive any other chemotherapy or radiotherapy, they will be followed for the antitumour effect and long-term safety of the IMP. Otherwise, subjects will be followed for antitumour effect up to the administration of any other chemotherapy or radiotherapy and for the safety aspects (in particular for haematological and renal safety) for the remainder of the long-term follow-up period.

### Five Years After the EOCT, ED or EOAC Visit

In the long-term follow-up period, subjects will be followed up every 3 months ( $\pm 2$  weeks) until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first. The following safety assessments will be performed by the site: these assessments can be performed on-site or by remote visit:

- Survival status
- Haematology and biochemistry tests (haemoglobin, platelet count, white blood cell count (absolute counts), AST, ALT, total bilirubin (with conjugated bilirubin if total bilirubin is abnormal i.e. outside the laboratory normal range), creatinine and urea) performed until the start of new antitumour treatment or up to 2 years after the EOCT, ED or EOAC visit, whichever comes first. Laboratory tests can be performed locally; results should be collected from the local laboratory.
- All AEs and serious adverse events (SAEs) are to be reported up to 6 months after the EOCT, ED or EOAC visit or until new antitumour treatment starts, whichever comes first. Thereafter, only AEs and SAEs related to the IMP or study procedure should be reported for the remainder of the safety follow-up period.
- Subsequent antitumour treatment

All collected data will be recorded in the eCRF.

### 6.2.4 Unscheduled Visits

Unscheduled visits or assessments may be necessary at any point during the treatment period to assess individual safety or tolerability. For example, laboratory assessments should be performed as clinically indicated during the study, tumour assessment could be performed whenever disease progression is suspected. Should an unscheduled visit or procedure listed in the schedule of assessment be performed during the study, the data will be recorded in the eCRF.

#### 6.3 Study Restrictions

#### 6.3.1 Concomitant Medications

The IMP excretion is partially hepatic; it is at the investigator's discretion to treat subjects with laxatives to accelerate excretion from the intestine.

Any antitumour treatment other than the IMP is prohibited from screening to 6 weeks after the last dose administration. All antitumour treatments given prior the first IMP dosing should be recorded in the eCRF. Dose, trade name, number of cycle and duration of treatment will be recorded as well.

Any prior or concomitant medication or therapy given to a subject either 30 days prior to dosing either within five half-lives of the previous product (if longer than 30 days) and during the study will be indicated in the eCRF. Dose, trade name and duration of treatment will be recorded.

Hormone substitution or therapies impacting one of the respective pituitary axis should be listed in the concomitant medications.

The potential of <sup>177</sup>Lu-3BP-227 to be associated with drug-drug interactions has not been established. It is anticipated that the major route of excretion will be via the kidneys and so medications which might affect renal function should be avoided or used with due caution as assessed by the investigator. These medications include but are not limited to the following.

Drugs causing prerenal damage:

- Drugs that cause excessive gastrointestinal losses, either through diarrhoea or vomiting, also cause volume depletion and may precipitate acute kidney injury (AKI).
- Nonsteroidal anti-inflammatory drugs (NSAIDs), even in short courses, can cause AKI as a result of renal underperfusion.
- Angiotensin-converting enzyme (ACE) inhibitors can also cause a deterioration in renal function. However, this is a problem only in patients with compromised renal perfusion, particularly those with renal artery stenosis.
- Care should be taken when an ACE inhibitor and NSAID are prescribed together, as this combination may precipitate an acute deterioration in renal function.

Drugs causing intrarenal damage:

- Intrarenal damage may result in a direct toxic effect on the kidneys or hypersensitivity reactions.
- Most drugs that cause damage within the kidneys do so as a result of hypersensitivity reactions, which involve either glomerular or interstitial damage.
- Drugs that have been reported to cause glomerulonephritis include penicillamine, gold, captopril, phenytoin and some antibiotics, including penicillins, sulfonamides and rifampicin.
- Drugs that may cause interstitial nephritis include penicillins, cephalosporins, sulfonamides, thiazide diuretics, furosemide, NSAIDs and rifampicin.
- There are a number of drugs that cause direct toxicity to the renal tubules (acute tubular necrosis) e.g. aminoglycosides, amphotericin and ciclosporin.

Drugs causing postrenal damage (urinary tract obstruction):

- High-dose sulfonamides, acetazolamide or methotrexate may cause crystalluria and could therefore cause urinary tract obstruction.
- Anticholinergics (e.g. tricyclic antidepressants), and alcohol may cause urinary tract obstruction due to retention of urine in the bladder.

Other nephrotoxic drugs:

- Cephalosporins: cephaloridine, one of the first cephalosporins introduced, has been associated with direct renal toxicity and is no longer in clinical use. Other cephalosporins are much less likely to produce renal damage but third-generation cephalosporins (e.g. cefixime) have (very rarely) been reported to cause nephrotoxicity.
- Analgesics:
  - NSAIDs may cause AKI due to hypoperfusion and interstitial nephritis, as well as analgesic nephropathy (chronic interstitial nephritis and papillary necrosis).
  - Analgesic nephropathy has been most commonly seen with combination analgesic products that contain aspirin and/or paracetamol.

- Analgesic nephropathy is one of the few preventable causes of chronic kidney disease. Discontinuation of the drugs often results in stabilisation or even improvement in renal function, but continued use leads to further renal damage.
- Lithium: serum levels of lithium consistently above the therapeutic range have been associated with development of nephrogenic diabetes insipidus.

Any further antitumour treatment for the disease under study that may have been taken during the long-term follow-up period should be reported on the prior and concomitant medications page of the eCRF.

## 7 STUDY TREATMENT

## 7.1 Investigational Medicinal Product

### 7.1.1 Dosage Form and Strength

<u>3BP-227 (also called IPN01087)</u> is a DOTA-conjugate of a NTSR1 antagonist, with a CCI

CI

The radiolabelled IMP, <sup>177</sup>Lu-3BP-227, is intended to be manufactured in two ways either decentralized at local radiopharmacies or centralized by a central manufacturing organisation.

For all screening and treatment formulations of the IMP, the specific activity of  $^{177}$ Lu-3BP-227 stays constant at 25 µg 3BP-227 per 1 GBq  $^{177}$ Lu at time of infusion.

One vial can contain up to 17.5 GBq of <sup>177</sup>Lu-3BP-227 at the end of manufacturing, this amount takes into account decay to ensure the administration of the highest planned activity of 7.5 GBq of <sup>177</sup>Lu-3BP-227 from the vial.

## 7.1.1.1 Decentralized Supply Dosage Form and Strength

The 3BP-227 is radiolabelled with <sup>177</sup>Lu at a specific ratio, time and temperature. The screening and treatment IMP are in a solution of ethanol and normal saline containing DiethyleneTriaminePentaAcetate (DTPA) and ascorbic acid to complex free <sup>177</sup>Lu and to protect the drug product from radiolysis, respectively. The IMP consists of a sterile clear solution, free from visible particles, filled in 27-mL Type I borosilicate glass vials closed with a chlorobutyl rubber stopper and crimped with aluminium seal.

## 7.1.1.2 Centralized Supply Dosage Form and Strength

The 3BP-227 is radiolabelled with <sup>177</sup>Lu at a specific ratio, time and temperature. The screening and treatment IMP are formulated in a solution of sterile water for injection containing DiethyleneTriaminePentaAcetate (DTPA) to complex free <sup>177</sup>Lu and ascorbate buffer to protect the drug product from radiolysis and to extend the shelf life. The IMP consists of a sterile clear to slightly yellow solution, free of suspended particles, filled in 25 mL Type I borosilicate glass vials, closed with a chlorobutyl rubber stopper and crimped with aluminium seal.

### 7.1.2 Supplies, Packaging and Labelling

For some qualified sites, the IMP will be produced and supplied by the local radiopharmacy with a batch number, a date and time of expiry and a certificate of analysis.

For the other sites, the IMP will be produced and supplied by a central manufacturing organisation with batch number, expiry date and time and certificate of analysis.

The IMP will be packaged in a primary Type I borosilicate glass vial and a secondary lead/tungsten containment for shielding, labelled according to applicable regulations, requirements and national laws in force.

### 7.1.3 Storage

The investigator, or an approved representative (e.g. pharmacist), will ensure that the IMP is maintained all the time in a dedicated, adequate shielding container and stored in a secured area, under recommended temperature monitored storage conditions.

In case of decentralized manufacturing by the local radiopharmacy, the IMP should be stored at ambient temperatures.

In case of manufacturing by the central manufacturing organisation, the IMP should be stored between  $+2^{\circ}C$  and  $+8^{\circ}C$ .

Since the IMP shelf life will vary depending on the mode of manufacturing (decentralised or centralised) and potentially on supporting data generated during the conduct of the study, the exact shelf life is indicated on the IMP label.

### 7.1.4 Preparation, Dispensing and Administration Procedures

All IMP provided to the site pharmacist will be allocated and dispensed by appropriately trained staff. IMP will be administered under medical supervision.

Specific preparation/reconstitution instructions, if applicable, will be specified in a separate document.

### 7.1.4.1 Route of Administration

The screening IMP formulation consists of 1 GBq of <sup>177</sup>Lu-3BP-227 in a total volume of 10 mL that will be administered by i.v. infusion over 10 minutes.

The treatment IMP formulation consists of a 2.5 to 10 GBq of <sup>177</sup> Lu-3BP-227 in a total volume of 20 mL that will be administered by i.v. infusion over 20 minutes.

If infusion reactions are observed, the infusion rate should be slowed to around 30 minutes or stopped if the reaction is severe.

In both cases, a 100 mL saline solution will be administered concomitantly, over a period of 30 minutes and starting at the same time as the IMP administration. The same venous access will be used for IMP and saline solution. Infusion should be done in the contralateral arm of the PK sampling.

## 7.1.4.2 Dosing Regimen in Phase I

The total radioactivity amount per cohort will be fractionated into two administrations separated by 4 weeks. Each separate dose will be instantly prepared prior to infusion.

The absorbed organ doses, especially the doses to kidney, bone marrow and liver as a limiting organ, will be analysed after each treatment cycle so that the following administrations can be adapted and to not exceed cumulative absorbed organ dose limits.

The subject will be instructed to drink water (at least 1.5 L/24 hours) on the days following each IMP administration.

### 7.1.4.3 Dosing Regimen in Phase II

The dose regimen in phase II will determined based on the data collected from the dose escalation part and will not exceed the MTCA.

#### 7.1.5 Accountability

The investigator should ensure adequate records (allocation, disposition, shipment, dispensing and returned drugs) are maintained in an IMP accountability log.

Unused IMP will be destroyed on site per local destruction procedure after approval of the sponsor.

### 7.1.6 Spillage

All due precautions and site procedures should be implemented to prevent spillage or leakage of radiodiagnostics or radiotherapeutics. Infusion bags, i.v. lines and venous access should all be secured, and the connections thoroughly checked. The infusion line should be taped in a loop and taped to the subject to prevent direct tension between the line and the venous access.

Despite precautions, if spillage or leakage should occur, then the site procedures must be implemented to protect the subject, staff and members of the public from radiation exposure. The subject should be moved from the area of the spillage or leakage while the area is decontaminated. Details of the spillage or leakage should be recorded (including how the incident happened, the time of the incident, an estimate (if possible) of the amount of substance lost) and the measures taken. In addition, the incident is to be reported in the same manner as an adverse event using the Medical Dictionary for Drug Regulatory Activities (MedDRA) Preferred Term 'Product Leakage' and as appropriate Preferred Term 'Occupational exposure

to radiation' (if there is exposure to staff) and Preferred Term 'Exposure to radiation' (if there is exposure to the subject or members of the public).

## 7.2 **Product Complaints**

The sponsor collects product complaints on study drugs used in clinical trials to ensure the safety of study participants, monitor quality and facilitate process and product improvements.

The investigator is responsible for handling the following aspects of the product complaint process in accordance with the instructions provided for this study:

- Recording a complete description of the product complaint reported and any associated AEs using the study-specific complaint forms provided for this purpose.
- Faxing the completed product complaint form within 24 hours to the sponsor or designee.

If the investigator is asked to return the product for investigation, they should return a copy of the product complaint form with the product.

### 8 SAFETY ASSESSMENTS

The following safety parameters will be collected and reviewed:

- DLTS (phase I dose escalation only, for definition see Section 4.7.1.1)
- AEs
- physical examination
- vital signs
- 12-lead ECG
- 24-hour 3-lead Holter ECG (phase I dose escalation only)
- ECOG performance status
- clinical laboratory tests including haematology, blood biochemistry, hormone analysis, urinalysis and pregnancy test
- specific renal safety biomarker.

The timepoints referenced in the study schedule of assessments in Attachment 1, Section 19.1 refer to the phase I dose escalation part of the study. In case of dose expansion cohorts and for phase II, these timepoints will be revised based on the dose escalation safety data and documented through a protocol amendment.

Further routine medical assessments or any additional safety procedures may be performed during the study, if warranted and after mutual agreement by the sponsor and the investigator, or when clinically indicated.

Any clinically significant finding that results in a diagnosis should be recorded as an AE from the time informed consent is given.

The investigator will be responsible for a clinical assessment of the study participants during the whole participation of the subjects in the study, from informed consent up to discharge from the study, and for the set-up of a discharge plan if needed.

Every effort should be made to ensure that all safety evaluations are completed by the same individual who made the initial baseline determination.

The sponsor's medical monitor and the Global Patient Safety (GPS) physician will monitor safety data throughout the course of the study.

#### 8.1 Adverse Events

Monitoring of AEs will be done from the time that a subject gives informed consent and throughout the study and will be elicited by direct, nonleading questioning or by spontaneous reports.

### 8.1.1 Definition of an Adverse Event

An AE is the development of an undesirable medical condition or the deterioration of a pre-existing medical condition, whether or not considered causally related to the product. An undesirable medical condition can be symptoms (e.g. nausea, chest pain), signs (e.g. tachycardia, enlarged liver) or the abnormal results of an investigation (e.g. laboratory findings, ECG).

An AE can include an undesirable medical condition occurring at any time, even if no IMP has been administered. All AEs, irrespective of causality, are to be reported to the sponsor up to 6 months after the EOCT, ED or EOAC visit or until new antitumour treatment starts, whichever comes first. After this timepoint, up to the end of the 5-year follow-up period, AEs should only be reported if the event is evaluated as related to the IMP or study procedure by the investigator.

This definition includes events occurring from the time of the subject giving informed consent until the EOS (last visit in the long term follow-up period)/ED.

Natural disease progression or deterioration of the malignancy treated in the study will be recorded as part of the efficacy evaluation. The signs and symptoms of disease progression will be documented as part of the radiological or clinical progression. Events that cannot be differentiated from progressive disease (PD) should be captured as AE/SAEs. Death due to disease progression should be reported as an SAE irrespective of causality if reported during the study period and up to 6 months after the end of the cycle of the last dose of study drug. Serious adverse events including death/fatal outcome will be reported to sponsor if evaluated as related to study treatment or procedure up to the protocol defined follow-up period and EOS.

## 8.1.2 Adverse Events Categorisation, Recording, and Follow-up

For all AEs, sufficient information should be obtained by the investigator to determine the causality of the AE (i.e. IMP administration, study procedure or other illness). The investigator is required to assess causality and record that assessment in the eCRF.

All AEs, including SAEs, are to be accurately recorded on the AE page of the subject's eCRF. Each event will be graded for severity using the classifications of NCI-CTCAE version 5.0 [31]. For events not addressed in the NCI-CTCAE version 5.0 classifications, the following grading will apply:

- **Mild** (Grade 1): Mild; asymptomatic or mild symptoms; clinical or diagnostic observations only; intervention not indicated.
- **Moderate** (Grade 2): Moderate; minimal, local or noninvasive intervention indicated; limiting age-appropriate instrumental ADL.
- Severe (Grade 3): Severe or medically significant but not immediately life-threatening; hospitalisation or prolongation of hospitalisation indicated; disabling; limiting selfcare ADL.
- Life-threatening (Grade 4): Life-threatening consequences; urgent intervention indicated.
- **Death** (Grade 5): Death related to AE.

The relationship of an AE to IMP administration will be classified by the investigator according to the following:

- **Related**: Reports including good reasons and sufficient information (e.g. plausible time sequence, dose-response relationship, pharmacology) to assume a causal relationship with IMP administration in the sense that it is plausible, conceivable or likely.
- Not related: Reports including good reasons and sufficient information (e.g. implausible time sequence and/or attributable to concurrent disease or other drugs) to rule out a causal relationship with IMP administration.

### 8.1.2.1 Assessment of Expectedness

The expectedness of an AE shall be determined by the sponsor according to the current approved version of the IB.

### 8.1.2.2 Follow-up of Adverse Events

Any AEs already recorded and designated as "continuing" should be reviewed at each subsequent assessment.

If an AE is still present at the end of the study, reasonable follow-up clinical monitoring should be managed by the investigator or any appropriate physician until the event or its sequelae resolves or stabilises at an acceptable level, as judged by the investigator and the sponsor's medical monitor or his/her designated representative. The frequency of follow-up evaluation is left to the investigator's discretion.

### 8.1.2.3 Reporting of Adverse Events

Any AEs/SAEs, irrespective of causality, are to be reported to the sponsor up to 6 months after the EOCT, ED or EOAC visit or until new antitumour treatment starts, whichever comes first. After this timepoint, up to the end of the 5-year follow-up period, AEs/SAEs should only be reported if the event is evaluated as related to the IMP or study procedure by the investigator.

Any AE considered related to IMP administration that the investigator becomes aware of after completion of the EOS/ED visit must be reported to the sponsor and will be recorded in the eCRF. EOS for AE reporting is defined as the end of the 5-year follow-up or death.

### 8.1.3 Serious Adverse Event Assessment and Reporting to Sponsor

The investigator must pursue and obtain information adequate both to determine the outcome of the AE and to assess whether it meets criteria for classification as an SAE requiring immediate notification to the GPS department of the sponsor.

An SAE is any AE that:

- Results in death;
- Is life-threatening; that is, any event that places the subject at immediate risk of death from the event as it occurs. It does not include an event that, had it occurred in a more severe form, might have caused death;
- Results in inpatient hospitalisation or prolongation of existing hospitalisation, excluding admission for social or administrative reasons:
  - Hospitalisation is defined as any inpatient admission (even if less than 24 hours). For chronic or long-term inpatients, inpatient admission also includes transfer within the hospital to an acute/intensive care inpatient unit.
  - Prolongation of hospitalisation is defined as any extension of an inpatient hospitalisation beyond the stay anticipated/required in relation to the original reason for the initial admission, as determined by the investigator or treating physician. For protocol-specified hospitalisation in clinical studies, prolongation is defined as any extension beyond the length of stay described in the protocol. Prolongation in the absence of a precipitating, treatment emergent, clinical AE (i.e. not associated with the development of a new AE or worsening of a pre-existing condition) may meet criteria for "seriousness" but is not an adverse experience and thus is not subject to immediate reporting to the sponsor.
  - Preplanned or elective treatments/surgical procedures should be noted in the subject's screening documentation. Hospitalisation for a preplanned or elective treatment/surgical procedure should not be reported as an SAE unless there are complications or sequelae which meet the criteria for seriousness described above.
- Results in a persistent or significant disability/incapacity, where disability is a substantial disruption of a person's ability to conduct normal life functions;
- Results in a congenital anomaly/birth defect in the offspring of a subject who received the IMP;
- Is an important medical event that may not result in death, be life-threatening, or require hospitalisation when, based upon appropriate medical judgement, may jeopardise the subject and may require medical or surgical intervention to prevent one of the outcomes listed in this definition. Examples of such medical events include allergic bronchospasm requiring intensive treatment in an emergency room or at home, blood dyscrasias or

convulsions that do not result in inpatient hospitalisation, or the development of drug dependency or drug abuse.

In addition to the above criteria, any additional AE that the sponsor or an investigator considers serious should be immediately reported to the sponsor. This includes any suspected or confirmed coronavirus COVID-19 (SARS-CoV-2) infection (seriousness criteria should be "other medically significant" if no other seriousness criteria are present (e.g. hospitalisation)).

In case of suspected or confirmed COVID-19 infection, the IMP administration may be temporarily discontinued depending on the subject's clinical presentation. In some cases, the investigator may request a subject be retested before the IMP administration is resumed.

All SAEs, regardless of treatment group or suspected relationship to IMP, must be reported immediately (within 24 hours of the investigator's knowledge of the event) to the sponsor's pharmacovigilance contact using the contact details specified on the front page of the current document.

Email is the preferred method of SAE notification. If email is not available, facsimile transmission is the alternative method. In rare circumstances and in the absence of facsimile equipment, notification by telephone is acceptable. If the immediate report is submitted by telephone, this must be followed by a detailed written report using the SAE report form.

The following information is the minimum that must be provided to the sponsor's pharmacovigilance contact:

- study number
- investigational site identification
- subject number
- AE
- investigator's name and contact details.

The additional information included in the SAE form must be provided to the sponsor or representative as soon as it is available. The investigator should always provide an assessment of causality for each event reported to the sponsor. Upon receipt of the initial report, the sponsor will ask for the investigator's causality assessment, if it was not provided with the initial report.

The investigator should report a diagnosis or a syndrome rather than individual signs or symptoms. The investigator should also try to distinguish a primary AE considered the foremost untoward medical occurrence, from secondary AEs that occurred as complications.

#### 8.1.4 Suspected Unexpected Serious Adverse Reactions

Suspected unexpected serious adverse reactions (SUSARs) are serious events that are not listed in the current approved IB and that the investigator identifies as related to the IMP or procedure.

#### 8.1.5 Pregnancy

Information regarding pregnancies occurring during the study must be collected and reported to the sponsor as SAEs. The sponsor will request further information from the investigator regarding the course and outcome of the pregnancy.

The investigator must instruct all female subjects to inform him immediately should they become pregnant. The investigator should counsel the subject, discuss the risks of continuing with the pregnancy and the possible effects on the foetus. Monitoring of the subject should continue until conclusion of the pregnancy.

Investigators must instruct all male subjects to inform them immediately, should their sexual partner become pregnant during the study. The investigator is to report to the sponsor if they become aware of a pregnancy occurring in the partner of a subject participating in the study. If

the female partner gives her consent, the pregnancy outcome should be followed up and reported.

If there is an abnormal pregnancy outcome or an AE reporting in the foetus/neonate/child following exposure to the IMP during an Ipsen-sponsored clinical study, the information will be collected in two clinical study SAE report forms, one for the mother and one for the foetus/neonate/child.

Pregnancies (in female subjects and partners of male subjects) with a conception date within 90 days after subject's dosing must also be reported to the investigator for onward reporting to the sponsor.

### 8.1.6 Adverse Events of Special Interest

Not applicable.

### **8.1.7** *Deaths*

All AEs resulting in death during the study period must be reported as SAEs.

The convention for recording deaths is as follows:

- adverse event term: lead cause of death (e.g. multiple organ failure, pneumonia, myocardial infarction);
- outcome: fatal.

The only exception is if the cause of death is unknown (i.e. sudden or unexplained death), in which case the AE term may be reported as "death" or "sudden death".

### 8.1.8 Reporting to Competent Authorities, IECs/IRBs and other Investigators

The sponsor will ensure that processes are in place for submission of reports of SUSARs occurring during the study to the competent authorities, IECs/IRBs and other investigators concerned.

Reporting will be done in accordance with the applicable regulatory requirements.

The sponsor must report all SUSARs to the European Medicines Agency's EudraVigilance database within 15 days. Fatal and life-threatening SUSARs should be reported within 7 calendar days, with another 8 days for completion of the report.

The sponsor may prepare additional reports for other authorities (e.g. FDA).

#### 8.2 Specific Safety Assessments

The investigator is responsible for monitoring subjects' safety at all times during the study.

For each assessment performed, the investigator must document their review of the result(s) in the source document(s) and, in case of an abnormal result or value(s) falling outside of predefined normal ranges, he/she should evaluate the finding according to NCI-CTCAE.

An abnormal finding may lead to a retest at the discretion of the investigator.

### 8.2.1 *Physical Examinations*

Complete physical examinations will be conducted as timepoints presented in the schedule of assessments in Attachment 1, Section 19.1.

This complete physical examination should include an evaluation of the general health, head, eyes, ears, nose, neck and throat (HEENT), heart, chest, abdomen, extremities, skin, lymph nodes, cardiovascular status and neurological status.

Limited physical examination should be directed at the evaluation of symptoms or specific potential safety issues (in particular in case of haematological or renal toxicity).

Any changes from baseline (prior to the first treatment IMP administration) or new abnormalities in physical examination findings judged to be clinically significant by the

investigator will be recorded as AEs. Any physical examination findings (abnormalities) persisting at the end of the study will be followed by the investigator until resolution or until reaching a clinically stable endpoint.

### 8.2.2 Vital Signs

Vital signs (systolic and diastolic blood pressure, heart rate and body temperature) will be assessed as presented in the schedule of assessments in Attachment 1, Section 19.1.

Systolic and diastolic blood pressure and heart rate assessments will be performed with an automated device so that measurements are independent of the observer. These parameters will be recorded after at least 5 minutes rest in supine position for supine assessment.

### 8.2.3 Height and Weight

Height and weight will be assessed as presented in the schedule of assessments in Attachment 1, Section 19.1.

### 8.2.4 Electrocardiograms

12-lead computerised standard ECGs, with paper printout, will be recorded and assessed locally.

Recordings should be performed in supine position after at least 5 minutes of rest.

During the screening period, a triplicate 12-lead ECG will be recorded to check eligibility criteria by measuring the ECG parameters, in particular the QTc (Fridericia's correction, QTcF) interval in order to exclude subjects with high/abnormal QT values before the screening administration (minus 15 minutes). Afterwards, single 12-lead ECG recordings will be performed at the end of <sup>177</sup>Lu-3BP-227 screening infusion ( $\pm$ 15 minutes) and 4 hours after the end of <sup>177</sup>Lu-3BP-227 infusion ( $\pm$ 30 minutes).

At each treatment administration, a triplicate 12-lead ECG will be recorded on Day 1 before the infusion (baseline) (minus 15 minutes).

Afterwards, single 12-lead ECGs, will be recorded on Day 1 just after <sup>177</sup>Lu-3BP-227 infusion ( $\pm 15$  minutes), at 4 hours ( $\pm 30$  minutes) after the end of <sup>177</sup>Lu-3BP-227 infusion and on Day 2 at 24 hours ( $\pm 4$  hours) after the end of <sup>177</sup>Lu-3BP-227 infusion as well as at EOCT and EOAC visits.

The timepoints of postdose assessments and the number of replicates may be modified, or additional timepoints may be added, based on data collected during the study.

Table 7 summarizes the ECG assessments to be performed for QTc monitoring during screening and treatment administration at each cycle.

|                                   | Timepoint   | Replicate  |
|-----------------------------------|---|------------|
|                                   | Before infusion (minus 15 minutes)                    | Triplicate |
| Screening                         | End of infusion (±15 minutes)                         | Single     |
| administration                    | 4 hours ( $\pm 30$ minutes) after the end of infusion | Single     |
|                                   | Before infusion (minus 15 minutes)                    | Triplicate |
| Treatment                         | End of infusion (±15 minutes)                         | Single     |
| administration<br>(at each cycle) | 4 hours ( $\pm 30$ minutes) after the end of infusion | Single     |
|                                   | 24 hours (±4 hours) after the end of infusion         | Single     |

 Table 6
 Timepoints and Replicates of 12-lead ECG Recordings

For each timepoint, computerised standard ECGs will be recorded so that the following parameter can be calculated and reported on the ECG paper printout:

- sinus rhythm
- duration of RR interval or heart rate
- duration of PR interval
- duration of QRS interval
- duration of QT interval
- QT interval corrected by the Fridericia's methodology.

Automated ECG interval data will be interpreted by a qualified physician at the site as soon as possible after the time of ECG collection, and ideally while the subject is still present, for immediate subject management. The qualified physician will document their review and interpretation (including evaluation of clinical significance in case of abnormality) on every ECG printout.

The paper printouts will be kept in the source documents at site. The QTcF intervals, ECG interpretation and abnormalities will be reported in the eCRF for integration with other clinical study data.

In the phase I dose escalation part, starting before each administration of the study drug, a 24hour 3-lead continuous ECG Holter will be recorded to monitor cardiac safety during the treatment. Holter assessments will be reviewed centrally. Results will be provided to the site in a timely manner and retained as source data.

#### 8.2.5 *Performance Status*

The performance status of the subjects will be defined by the ECOG scoring and assessed as presented in the schedule of assessments in Attachment 1, Section 19.1. For ES subjects, ECOG assessment should take into account disabilities related to surgery (see inclusion criterion 9).

#### 8.2.6 Clinical Laboratory Tests

Blood and urine samples collection will be performed for standard clinical laboratory tests, including biochemistry, haematology, hormone analysis and urinalysis panels, as well as specific tests and pregnancy tests for women of childbearing potential, at timepoints indicated in the schedule of assessments in Attachment 1, Section 19.1.

Clinical laboratory tests will be analysed by local laboratory.

The results of laboratory tests performed during the screening phase must be obtained before dosing on Day 1 (within 48 hours before Cycle 1 Day 1 visit).

The investigator will review each safety laboratory test result, document the review, and if applicable, record any AEs according to NCI-CTCAE.

Abnormalities in laboratory test values should only be reported as AEs if any of the following apply:

- They result in a change in IMP schedule of administration (change in dosage, delay in administration, IMP discontinuation),
- They require intervention or a diagnosis evaluation to assess the risk to the subject,
- They are considered clinically significant by the investigator, or the laboratory test abnormality suggests a disease and/or organ toxicity that is new or has worsened from baseline based on sponsor review.

All clinically significant out of normal range laboratory tests occurring during the study may be repeated at appropriate intervals until they return to baseline or to a level deemed acceptable by the investigator or until the abnormality is explained by an appropriate diagnosis that does not require further follow-up.

NOTE: it is to be borne in mind that all the biological samples will contain the radioactive substance <sup>177</sup>Lu-3BP-227 and appropriate measures must be taken during handling, storage and shipment.

### 8.2.6.1 Blood Analyses

Venous blood samples will be taken at the timepoints described in the schedule of assessment (Attachment 1, Section 19.1). Samples collected at Day 1 are to be taken predose, samples collected at Day 2 are to be collected at 24 hours  $\pm 2$  hours.

An estimation of the amount of blood required is provided in Attachment 2, Section 19.2.

The parameters to be assessed are listed in Attachment 3, Section 19.3.

For the determination of renal function, eGFR will be calculated by using the MDRD formula preferably; other methods to estimate renal function are acceptable if they are validated measures such as creatinine clearance by Cockcroft & Gault. Hormone analyses will only be performed for subjects who do not have substitution or therapy impacting one of the respective pituitary axis (e.g. no cortisol sampling in subjects who receive corticosteroids, no thyroid-stimulating hormone and free thyroxine sampling in subjects who have thyroxine substitution.

### 8.2.6.2 Urinalysis

Freshly voided urine samples (at least 10 mL) will be collected at the timepoints described in the schedule of assessment (Attachment 1, Section 19.1).

A dipstick assessment of the parameters listed in Attachment 3, Section 19.3 will be performed. In case of positivity and in the absence of common causes for proteinuria, a proteinuria determination in a 24 hours urine collection sample will be performed, according to the investigator's discretion and within a timeframe compatible with sufficient radioactivity excretion from the subject.

In case of abnormal result on the dipstick, a request for a confirmatory analysis or additional assessments might be sent to the local laboratory, at the discretion of the investigator.

Microscopy will be performed, if indicated, but results will not be collected in the eCRF. If in the opinion of the investigator there are any clinically significant abnormalities in microscopy, they will be recorded as AEs in the eCRF.

# 8.2.6.3 Pregnancy Test

Serum and urine pregnancy tests will be performed as per the schedule of assessments (Attachment 1, Section 19.1).

# 8.2.7 Specific Renal Safety Biomarkers



### 9 PHARMACOKINETIC ASSESSMENTS

The PK of <sup>177</sup>Lu-3BP-227 and 3BP-227, as well as the dosimetry assessments and timepoints described below are related to the phase I part of the study. The PK and dosimetry assessments will also be included in phase II. However, the timepoints might be revised based on the phase I results and documented as part of a protocol amendment.

## 9.1 Pharmacokinetics of <sup>177</sup>Lu-3BP-227

### 9.1.1 Blood Sample Collection

Total activity concentration in whole blood will be determined using a gamma counter calibrated for <sup>177</sup>Lu, according to the dosimetry operational manual (DOM).

For each subject, total activity concentration in whole blood will be measured. Each subject will have 18 blood samples (2 mL each) collected at Cycle 1 and Cycle 2 on the following timepoints after the end of <sup>177</sup>Lu-3BP-227 infusion:

- Day 1: before the infusion (baseline), at the end of infusion (0), 5 minutes  $\pm$  1 minute, 30 minutes  $\pm$  5 minutes, 90 minutes  $\pm$  15 minutes and 4 hours  $\pm$  30 minutes post infusion
- Day 2: 24 hours  $\pm$  2 hours
- Day 3: 48 hours  $\pm$  2 hours
- Day 4 to Day 5: 72 to 96 hours.

For subjects receiving additional administrations (up to four additional cycles), blood samples will be taken during first and third additional cycles according to the timepoints described above. After the second and fourth additional administrations, a single blood collection will be performed at 24 hours (as close as possible to the SPECT/CT) for the radioactive assessment of <sup>177</sup>Lu-3BP-227.

Blood samples should be collected from the arm opposite to that of the study drug infusion, or from another site.

The accurate time of sample collection and the duration for measuring the activity concentration must be recorded. Any issues associated with sample collection or processing should be reported to the sponsor's monitor.

Complete instructions for sample collection, processing and handling will be provided in a DOM.

Note that nominal sample collection times may be changed during the study based on available data. These changes will be decided by the sponsor. In any case, the total number of assessments may be decreased but it will not be increased.

### 9.1.2 Urine Sample Collection

To determine the renal excretion of <sup>177</sup>Lu, total activity concentration in urine will be determined using a gamma counter calibrated for <sup>177</sup>Lu, according to the DOM.

The samples for urine total activity concentration analysis will be taken from urine collected during four different periods at Cycle 1 only: from the start of the <sup>177</sup>Lu-3BP-227 infusion to 6 hours, 6 to 12 hours, 12 to 24 hours and 24 to 48 hours after the end of infusion (from the start of the <sup>177</sup>Lu-3BP-227 infusion to 6 hours after the end of the infusion only for US sites). Subjects should void shortly before the <sup>177</sup>Lu-3BP-227 infusion (not to be collected) and shortly before the end of the last collection period.

The accurate time of urine collection and the total urine volume for each collection interval must be recorded. All problems associated with sample collection or processing should be reported to the sponsor's monitor.

Complete instructions for urine collection, processing and handling will be provided in the DOM.

### 9.2 Nuclear Medicine Imaging for Dosimetry

Imaging of the tumour uptake of <sup>177</sup>Lu-3BP-227 will be performed locally by the site investigator. All images will be centralized and analysed by independent readers for dosimetry analysis of the dose limiting organs and tumour lesions. For screening, the uptake of <sup>177</sup>Lu-3BP-227 at tumour lesions will be assessed locally and will take precedence to central assessment.

During the treatment period, to improve the determination of the biokinetics of <sup>177</sup>Lu-3BP-227 and perform an absolute quantification of radioactivity in target organs, whole body scans (planar scintigraphy) and SPECT/CT will be performed at Cycles 1 and 2 at the following timepoints just after the end of <sup>177</sup>Lu-3BP-227 infusion:

- Day 1: 4 (±2) hours
- Day 2: 24 (±6) hours
- Day 3: 48 (±6) hours
- Day 4: 72 to 96 hours
- Days 7 to 8: 138 to 168 hours

Within each cycle, a single SPECT/standard dose CT will be performed. For all other time points, SPECT/low dose CT will be performed. Details of the procedures will be provided in the Image Acquisition Guidelines.

For subjects receiving additional administrations (up to four additional cycles), dosimetry assessments will be performed after the first and third additional administrations with nuclear medicine imaging as described above. After the second and fourth additional administrations, a single whole body scan and SPECT/standard dose CT at 48 ( $\pm$ 6) hours will be performed.

Details on the procedures will be given in a separate DOM.

NOTE: The optimal schedule of the whole body scans (planar scintigraphy) and SPECT/CT has not been established. Some adaptations in timepoints or number of assessments might be required during the study and will be decided by the dosimetry expert and the sponsor. In any case, the number of whole-body scans (planar scintigraphy) and SPECT/CT will not increase and corresponds to the maximum number of assessments. The schedule adaptation can only lead to a decreased number of assessments.

### 9.3 Pharmacokinetics of 3BP-227

## 9.3.1 Blood Sample Collection

Blood will be sampled for the purpose of determining plasma levels of 3BP-227 using high-performance liquid chromatography (HPLC) with tandem mass spectrometric (MS/MS) detection, according to a separate protocol established with the dedicated analytical laboratory.

For each subject, 3BP-227 plasma concentration will be measured. Each subject will have 10 blood samples (2 mL each) collected at Cycle 1 only at the following time points:

- Day 1: before the infusion (baseline), at the end of infusion of <sup>177</sup>Lu-3BP-227 (0), 5 minutes ±1 minute, 30 minutes ±5 minutes, 90 minutes ± 15 minutes and 4 hours ±30 minutes, 6 hours ±30 minutes and 8 hours ±1 hour after the end of infusion.
- Day 2: 24 hours  $\pm 2$  hours after the end of infusion of <sup>177</sup>Lu-3BP-227.
- Day 3: 48 hours  $\pm 2$  hours after the end of infusion of <sup>177</sup>Lu-3BP-227.

Blood samples should be collected from the arm opposite to that of the study drug infusion, or from another site.

The accurate time of sample collection must be recorded. Any issues associated with sample collection or processing should be reported to the sponsor's monitor.

Complete instructions for sample collection, processing, handling and shipment will be provided in the laboratory manual.

Residual plasma used for 3BP-227 PK analysis may also be used for exploratory analysis. This could include using leftover plasma for protein binding analysis, metabolite profiling or analysis of excipients. Plasma samples remaining from the analysis may be retained by the sponsor for additional investigations (i.e. long term stability, reproducibility).

Note that nominal sample collection times may be changed during the study based on available data. These changes will be decided by the sponsor. In any case, the total number of assessments may be decreased but it will not be increased.

#### 9.3.2 Urine Sample Collection

To determine the renal excretion of 3BP-227, the concentration of 3BP-227 in urine will be determined using HPLC with MS/MS detection, according to a separate protocol established with the dedicated analytical laboratory.

The samples for urine 3BP-227 concentration analysis will be taken from urine collected during four different periods at Cycle 1 only: from the start of the infusion to 6 hours, 6 to 12 hours, 12 to 24 hours and 24 to 48 hours after the start of infusion (from the start of the infusion to 6 hours after the end of the infusion only in US sites). Subjects should void shortly before the <sup>177</sup>Lu-3BP-227 infusion (not to be collected) and shortly before the end of the last collection.

The accurate time of urine collection and the total urine volume for each collection interval must be recorded. Any issues associated with sample collection or processing should be reported to the sponsor's monitor.

Complete instructions for urine collection, processing, handling and shipment will be provided in the laboratory manual.

### 10 PHARMACODYNAMIC/EFFICACY ASSESSMENTS

The study schedule of assessments in Attachment 1, Section 19.1 is applicable for the phase I parts of the study. The need and timepoints for phase II will be assessed based on the phase I results. Eventual changes will be described in a protocol amendment.

#### **10.1** Imaging Assessments

Tumour response assessments will be performed by the site investigator (local) for the phase I dose escalation part and by independent reader (central) for the phase II. All contrast enhanced computed tomography/magnetic resonance imaging (ceCT/MRI) scans will be obtained and used for the tumour response assessments as described in RECIST version 1.1 (Attachment 4, Section 19.4) by the investigator and/or independent readers. Either ceCT or MRI images can be used for tumour assessment. However, the same method should be used throughout the study.

All <sup>18</sup>F-FDG-PET images will be obtained and used for the metabolic tumour response assessments as described in PERCIST version 1.0 [38,] (see Table 7 in Section 19.6) by the investigator and/or independent readers.

During phase I, even if analysed locally, all images will be transferred to the central imaging laboratory for storage and potential central analysis upon sponsor request, e. g for the analysis of the homogeneity of the tumour uptake by <sup>177</sup>Lu-3BP-227, using PERCIST version 1.0 or practical PERCIST. During phase II, the independent panel's assessment will take precedence in case of a discrepancy between the local and central reviews using PERCIST version 1.0.

### 10.1.1 Screening Imaging Assessments

The screening tumour assessments will be performed within 3 weeks before screening IMP infusion. If a historic ceCT/MRI scan is present that is not older than 1 month at the time of informed consent signature, this scan can be used. However, the site should ensure that all required anatomies (including brain) are covered and perform scanning for missing anatomies. Imaging parameters and approach used at screening should remain consistent throughout the study and follow-up. This examination will serve as inclusion criteria check and baseline assessment for RECIST v1.1 tumour response evaluation.

As bone scan, PET scan, or plain films are not considered adequate imaging techniques to measure bone lesions, these lesions are difficult to assess through RECIST version 1.1. These techniques can be used to confirm the presence or disappearance of bone lesions. Lytic bone lesions or mixed lytic-blastic lesions, with identifiable soft tissue components, that can be evaluated by cross sectional imaging techniques such as CT or MRI can be considered as measurable lesions if the soft tissue component meets the definition of measurability described in RECIST version 1.1. Blastic bone lesions are nonmeasurable. As a consequence, subjects with only bone lesions are not eligible for the study.

### 10.1.2 Phase I Imaging Assessments

Radiological assessments for tumour response will be performed as per schedule of assessment (Attachment 1, Section 19.1.) as well as when clinically indicated. Additionally, in the event of biological or clinical signs of disease progression further radiological assessments can be done based upon the investigator's judgment.

For PDAC, CRC, GC and GIST, chest/abdomen/pelvis scans are required and can be performed in two images if required; from chest to abdomen and from abdomen to pelvis. The scans should extend from the lateral ends of the clavicles (to ensure complete coverage of lung apices) down to the lesser trochanters or caudally thereof (to ensure complete coverage of inguinal lymph nodes). For abdominal/pelvic scans, the scan should begin cranially at the right dome of the
diaphragm and extend down to the lesser trochanters or caudally thereof (to ensure complete coverage of inguinal lymph nodes).

For SCCHN, scans from head to pelvis are required and can be performed in three images if required; from head to chest, from chest to abdomen and from abdomen to pelvis).

For ES subject, whole body scans will be performed at screening to assess the localization of the targeted lesions. In case of lesions outside of the usual acquisition areas (chest to pelvis), the number of bed positions will be adapted accordingly for further assessment.

<sup>18</sup>F-FDG-PET images will be assessed locally and the results for screening and EOCT visit will be collected in the eCRF.

After the EOCT, ED or EOAC visit, subjects will be followed up every 3 months ( $\pm 2$  weeks) to further evaluate the efficacy (by CT or MRI) until disease progression, administration of any other chemotherapy or radiotherapy, lost to follow-up, withdrawal of consent, death or a maximum of 2 years, whichever occurs first (see Section 6.2.3). After disease progression is confirmed, no further CT/MRI scans will be required for tumour/disease assessment. The survival status and safety of the subjects will continue to be monitored as indicated in Section 8.1.2.3 until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first.

## 10.1.3 Phase II Imaging Assessments

For phase II, serial radiologic assessments of tumours by CT or MRI will be performed by investigators at Baseline and every 8 weeks (anticipated duration of two cycles), until either disease progression, start of a new antitumour treatment, withdrawal of consent, or maximum cumulative radioactivity is reached. For each subject, the same method of assessment must be used throughout the study.

Tumour assessments should be completed until it has been determined that the subject has progressive disease (in accordance with RECIST version 1.1). In the event of the subject discontinuing study treatment for reasons other than disease progression, a tumour assessment should be completed as soon as possible relative to the date of study termination, to assess overall disease status.

<sup>18</sup>F-FDG-PET scans images will be assessed to measure metabolic tumour response using PERCIST version 1.0.

Complete instructions from acquisition to central review will be provided by the contract research organisation (CRO) in charge of the central review and assessment.

After the EOCT, ED or EOAC visit, subjects will be followed up every 3 months ( $\pm 2$  weeks) to further evaluate the efficacy (by CT or MRI) until disease progression, administration of any other chemotherapy or radiotherapy, lost to follow-up, withdrawal of consent, death or a maximum of 2 years, whichever occurs first (see Section 6.2.3).

## **10.2** Quality of Life Questionnaire

Quality of Life questionnaires will be completed by the subjects during phase II. The questionnaire types and the timepoints for assessment will be described as part of a protocol amendment at the end of phase I, based on the selected phase II study design and the selected indications for investigation.

It is important that the quality of life questionnaires are completed prior to the other required assessments and that the investigator does not influence the subject's responses to the questionnaires in any way.

## **10.3** Tumour Markers

Tumour markers will be measured in serum at specific timepoints as described in the study schedule of assessments in Attachment 1, Section 19.1.

According to the type of tumour, the following tumour markers will be analysed:

- Serum carcinoembryonic antigen (CEA) will be analysed in all subjects.
- Serum cancer antigen 19-9 (CA 19-9) will be analysed in all subjects except in subjects with SCCHN.
- Serum LDH and bone specific alkaline phosphatase will be analysed only in subjects with ES.

All measurements will be done by the local laboratory.

# 10.4 Genomic Profiling in Circulating Cell-free DNA and Germline DNA



# 10.5 Gene Mutation Status



**10.6 Tumour Biopsy** 

| CCI |  |  |
|-----|--|--|
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |
|     |  |  |

Study number D-FR-01087-001 - Protocol v 8.0 Date: 12 June 2020

# 11 **BIOBANKING**

Analysis of biobank samples will be performed outside the scope of the main study and will be reported separately. Therefore, this analysis is optional.



## 12 STATISTICAL ANALYSES

An interim analysis of the data will be performed between phase I and phase II.

## 12.1 Analysis Sets

For dose escalation only, all subjects who complete Cycle 2 or discontinue during Cycle 1 or Cycle 2 due to a DLT will be included in the MTCA and MTSA evaluations.

For the overall study, all subjects who received at least one dose of the IMP will be included in the safety evaluation. All subjects who receive at least one dose of IMP and who have postbaseline efficacy data will be included in the efficacy evaluations. All subjects with evaluable 3BP-227 PK data and no major protocol deviation with an impact on PK analysis will be included in the evaluation of 3BP-227 PK. All subjects with evaluable dosimetry images and no major protocol deviation with an impact on dosimetry analysis will be included in the evaluation of dosimetry. All subjects with evaluable blood or urine dosimetry data and no major protocol deviation with an impact on <sup>177</sup>Lu-3BP-227 PK analysis will be included in the evaluation of <sup>177</sup>Lu-3BP-227 PK. Major deviations will be described in a specific document.

# **12.2** Statistical Methodology for Phase I

## 12.2.1 Sample Size Calculation

## 12.2.1.1 Dose Escalation

As this is primarily a descriptive safety and tolerability study, the total number of subjects is not based on a formal statistical sample size calculation.

For the dose escalation part, the actual sample size required to adequately determine the MTCA/MTSA during dose escalation depends on the initial dose, rate of dose escalation and the observed dose-toxicity and dose/radiation exposure relationships. Simulation studies have been performed to quantify the operational characteristics (i.e. precision of the MTCA/MTSA, sample size, number of subjects being over/under dosed) of the adaptive dose-escalation design under a number of plausible dose-DLT relationship scenarios. Results are provided in Attachment 5, Section 19.5. Based on experience, the chosen sample size of three to five subjects per cohort is considered sufficient to fulfil the objectives of the study. It is anticipated that approximately 30 subjects will be required to establish the MTCA or MACA.

## 12.2.1.2 Dose Expansion

If safety evaluation of <sup>177</sup>Lu-3BP-227 and dose schedules cannot be fully explored during the phase I dose escalation part, a phase I expansion part enrolling up to 45 subjects will serve to accomplish this objective including, but not limited to, schedules of high loading doses followed by fractionated lower doses or evaluation of <sup>177</sup>Lu-3BP-227 in combination with other antitumoral treatments (to be defined). The expansion part will also serve to clarify any uncertainties of antitumour responses. The sample size will be specified in a protocol amendment.

## 12.2.2 Statistical Methods

Details of the statistical analysis will be specified in the statistical analysis plan (SAP). The statistical analyses will be conducted using Statistical Analysis System. Summary descriptive tables, by RL, will be provided.

Continuous data will be summarised with the following items: frequency, median, range, mean and coefficient of variation (CV) if relevant.

Categorical data will be presented in contingency tables with frequencies and percentages of each modality (including missing data modality). The 95% confidence interval will be calculated following the exact method.

To describe time dependent parameters (PFS, OS), Kaplan-Meier curves and life tables will be provided. The 95% confidence interval of the median will be given.

An overview of the main analysis strategy is provided in the following sections.

# 12.2.3 Demographic and Other Baseline Characteristics

Summary statistics will be presented for the total study population and by cohort. Frequency tables for qualitative data will be provided. Medical and surgical history findings will be summarised using MedDRA terms.

## 12.2.4 Maximum Tolerated Single and Cumulative Activity

Individual listings and treatment summaries of DLTs with NCI-CTCAE code and grade will be presented.

The incidence of subjects with DLTs during Cycles 1 or 2 will be summarised by treatment and, if possible, modelled as a function of the fractional doses for Cycles 1 and 2 using Bayesian logistic regression. The moderately-informative independent priors used during the interim analysis (see Attachment 5, Section 19.5 for details) as well as noninformative priors will be used for this analysis in order to assess sensitivity of the estimates. Parameter estimates and model predictions will be reported with 90% credibility sets. The MTCA and MTSA will be computed as a derived function of model parameters.

The posterior distribution of the MTCA and MTSA will be summarised in tabular and graphical formats.

The number of subjects with at least one DLT or NCI-CTCAE Grade  $\geq 2$  (excluding hair loss) will be tabulated by RL by cycle and overall. The type of DLT and NCI-CTCAE Grade  $\geq 2$  will also be presented.

## 12.2.5 Safety Evaluation

For the overall study, descriptive statistics will be calculated on the safety parameters. No formal statistical analyses of safety data are planned.

Summaries will be prepared by treatment group and, as needed, by timepoint.

All AEs will be coded according to the latest version of the MedDRA and NCI-CTCAE.

Study drug TEAE summaries will include the overall incidence (by system organ class (SOC) and preferred term (PT)), events by maximum intensity, events by relationship to study drug, events leading to discontinuation of study drug and SAEs.

Physical examination findings, vital signs, ECG 3-lead 24h Holter recordings and clinical laboratory parameters will be summarised descriptively at each timepoint. Actual and change from baseline data will be calculated and summarised where data are available. The investigator's interpretation of 12-lead ECGs will be listed.

Concomitant medications will be coded using the latest version of the World Health Organization (WHO) drug dictionary and will be summarised by treatment group and overall with the number and percentage of subjects receiving concomitant medication by drug class and preferred drug name.

The NCI-CTCAE version 5.0 classification [31] will be used to classify all TEAEs and laboratory abnormalities.

Maximum grade or severity will be tabulated by subject for each MedDRA SOC and PT. Analyses of AEs and SAEs will be performed in two different ways: regardless of the relationship to the study treatment and related to the study treatment. Moreover, all AEs without SAEs and SAEs only will be tabulated.

For haematological and biochemical toxicities, the worst NCI-CTCAE grade by subject and by cycle will be tabulated and listed. For white blood cells, neutrophils, platelets and haemoglobin, with associated Grade 3 or 4 toxicities, nadir and day to nadir will be calculated.

A TEAE is defined as any AE that occurs during the active phase of the study if:

- It was not present prior to receiving the first dose of <sup>177</sup>Lu-3B-227; or
- It was present prior to receiving the first dose of <sup>177</sup>Lu-3B-227 but the intensity increased during the active phase of the study; or
- It was present prior to receiving the first dose of <sup>177</sup>Lu-3B-227, the intensity is the same but the drug relationship became related during the active phase of the study.

# 12.2.6 Pharmacokinetics, Biodistribution and Dosimetry

# 12.2.6.1 Biodistribution and Pharmacokinetic Analysis of <sup>177</sup>Lu-3BP-227

The biodistribution and PK analyses of the radiopharmaceutical <sup>177</sup>Lu-3BP-227 will be performed under the responsibility of the sponsor's Clinical Pharmacokinetics and Pharmacometrics department.

The following biodistribution (i.e. organ PK) parameters will be evaluated:

- maximal uptake (%) at the target lesion
- maximal uptake (%) in discernible organs and blood
- the  $C_{max}$  of  $^{177}Lu$ -3BP-227 in discernible thoracic and abdominal organs, target lesion and blood
- the  $T_{max}$  of  $^{177}Lu$ -3BP-227 in discernible thoracic and abdominal organs, target lesion and blood
- the AUC of <sup>177</sup>Lu-3BP-227 in discernible thoracic and abdominal organs, target lesion and blood
- terminal  $t_{1/2}$  of activity concentrations of <sup>177</sup>Lu-3BP-227 in blood.

Data analysis and all biodistribution parameters will be further described in a SAP.

Biodistribution data for target lesion and all abdominal and thoracic organs analysed will be listed for each study subject and each sampling timepoint. In addition, arithmetic mean, arithmetic standard deviation, CV and number of data points will be provided per sampling point. The biodistribution parameters will be estimated for each subject. Total radioactivity cumulatively excreted in the urine in the interval 0 to 6 hours, 6 to 12 hours, 12 to 24 hours and 24 to 48 hours postinfusion will be determined for each subject.

Measured total <sup>177</sup>Lu-3BP-227 radioactivity concentrations in whole blood and urine will be listed for each study subject and each sampling time point. In addition, means, standard deviation, CV and number of data points will be provided per sampling point. Based on the individual <sup>177</sup>Lu-3BP-227 blood time-activity curves, the concentration-time profiles of <sup>177</sup>Lu-3BP-227 and the AUC in blood, will be estimated for each subject.

# 12.2.6.2 Radiation Dosimetry Analysis of <sup>177</sup>Lu-3BP-227

The radiation dosimetry analysis of the radiopharmaceutical <sup>177</sup>Lu-3BP-227 will be performed by an independent expert under the responsibility of the sponsor's Clinical Pharmacokinetics and Pharmacometrics department.

Further details on dosimetric assessments and on dosimetric parameters will be provided in a Dosimetry Assessment Plan.

The dosimetric assessments will be performed and reported according to the criteria set out by the European Association of Nuclear Medicine Dosimetry Committee guidance document: good practice of clinical dosimetry reporting [32].

Calculations will be conducted for the following parameters (only in organs showing uptake):

- organs receiving the highest absorbed dose
- specific absorbed dose to the target lesion (Gy/GBq)
- specific absorbed dose per organ (Gy/GBq)
- cumulative absorbed organ doses (Gy).

Cumulative absorbed organ doses (Gy) and organs of highest radioactivity uptake will be identified visually. Regions of interest will be placed over these organs to determine the relative activity in the respective organs. Time activity curves (describing % IA/ROI of the activity amount injected versus time, considering renal excretion activity) will be derived. The absorbed doses of the dose-limiting organs (kidney, bone marrow and liver) will be evaluated and reported to the investigator before the next administration of <sup>177</sup>Lu-3BP-227 can be initiated to enable activity adaptations in the event the next dose may exceed the organ limits of 23 Gy (kidney), 2 Gy (bone marrow) and 30 Gy (liver). The dosimetry results for all other organs will be finalized for the final study report with the restriction that the dosimetry data must be available for the SRC meeting. These data can be reviewed at any time if a major safety issue occurs.

The proportion of renally excreted radioactivity, whole blood radioactivity, and dosimetric whole body images will be used to calculate the dosimetry of <sup>177</sup>Lu-3BP-227.

# 12.2.6.3 Pharmacokinetic Analysis of 3BP-227

The PK analysis of 3BP-227 will be performed under the responsibility of the sponsor's Clinical Pharmacokinetics and Pharmacometrics department.

Analysis of PK data by a noncompartmental approach will be documented in a separate SAP. Individual plasma and urine concentrations of 3BP-227 will be listed and summarised by time points using descriptive statistics for continuous variables (number of available observations, mean, median, standard deviation, minimum, maximum, geometric mean, and geometric coefficient of variation assuming lognormally distributed data). Linear and semilogarithmic plots of individual and mean plasma concentration-time profiles as well as spaghetti plots will be reported.

Any suspicious concentration will be investigated and kept in the PK analysis if possible. All excluded concentrations will be justified in the report.

If 3BP-227 levels are measurable in plasma and urine, PK parameters of 3BP-227 (including, but not limited to,  $C_{max}$ , AUC,  $t_{1/2}$ , Cl, Vd, Ae,  $CL_R$ ) will be derived using the noncompartmental approach on the individual plasma concentration-time profiles of 3BP-227 and on the individual urine concentrations.

An attempt to build an integrated model taking into account PK, dosimetry as well as efficacy and safety data will be made if warranted by the data. The exploratory analysis will be captured in a separate data analysis plan and reported in a standalone report.

# 12.2.7 Pharmacodynamic/Efficacy Evaluation

For phase I, tumour response will be evaluated only by the site investigator, using the revised RECIST guideline version 1.1 (see Attachment 4, Section 19.4). Only subjects with measurable disease at baseline, who have received at least two administrations of <sup>177</sup>Lu-3BP-227 and reached the end of Cycle 2 or EOCT visit would be considered evaluable for response.

All measurable lesions up to a maximum of two lesions per organ and five lesions in total, representative of all involved organs, should be identified as target lesions and recorded and measured at baseline. Target lesions should be selected on the basis of their size (lesions with the longest diameter), be representative of all involved organs and permit reproducible repeated measurements. On occasion, if the largest lesion does not permit reproducible measurement,

the next largest lesion which can be measured reproducibly will be selected. A sum of the diameters (longest for non-nodal lesions, short axis for nodal lesions) for all target lesions will be calculated and reported as the baseline sum diameters. If lymph nodes are included in the sum, then only the short axis will be added into the sum. The baseline sum of diameters will be used as reference to further characterise any objective tumour regression in the measurable dimension of the disease.

All other lesions (or sites of disease) including any measurable lesions over and above the five target lesions should be identified as nontarget lesions and should also be recorded at baseline. Measurements of these lesions are not required, but the presence, absence, or unequivocal progression of each should be noted throughout follow-up.

Subjects will be evaluated as CR, PR, SD, PD, or inevaluable according to RECIST version 1.1. Based on this classification, the following endpoint will be calculated as defined by RECIST version 1.1 and below.

- Objective Response Rates (ORR): proportion of subjects with a best overall response (BOR) characterised as either a Complete Response (CR) or Partial Response (PR) relative to the total number of evaluable subjects
- Disease control rate (DCR): proportion of subjects with a BOR characterised as CR, PR or SD relative to the total number of evaluable subjects

The PFS and OS will also be evaluated as defined per RECIST version 1.1.

Details of the response evaluation will be given in a separate image review charter. Details of the analysis will be specified in the SAP.

Efficacy data will be summarised using descriptive statistics and will be graphically displayed, if appropriate. The correlation between pharmacodynamic parameters and selected safety, efficacy, or PK parameters may be graphically displayed. Further statistical analyses may be conducted. Details of the response evaluation will be given in a separate image review charter. Details of the analysis will be specified in the SAP.

An attempt to build an integrated model taking into account PK, dosimetry as well as efficacy and safety data will be made, if warranted by the data. The exploratory analysis will be captured in a separate data analysis plan and reported in a standalone report.

# 12.2.8 Interim Analysis/Safety Review Committee

During the dose escalation part and before each decision on further dose escalation, safety and dose limit organ exposure data will be summarised and presented to the SRC. This SRC will be set-up for the dose escalation decisions (after the end of the third subject's Cycle 1) and second dosing cycle (after the end of the third subject's Cycle 2).

At the time of SRC meeting, all cumulative available information will be reviewed.

A continual reassessment method (CRM) in conjunction with a Bayesian approach modelling of DLT rates and radiation exposure to target organs may be performed during the planned review meetings to generate additional relevant information for the adaptive dose selection decisions. PK and pharmacodynamic data may be also incorporated in the model. The SRC will review all available data and make the final decision as to dose escalation, de-escalation, or cohort expansion during the adaptive dose-escalation phase. This group will also determine when to implement predefined stopping rules.

An interim analysis of the escalation cohort data may be done when the MTCA or MACA have been determined.

A specific charter will be developed to define roles and responsibilities, as well as the data set to be reviewed by the SRC.

Full analysis will be performed on the entire dataset at the EOCT and at the end of the long-term follow-up.

# 12.2.9 Clinical Study Report

A final clinical study report (CSR) will be prepared according to the International Council for Harmonisation (ICH) guideline on structure and contents of CSRs. A final CSR will be prepared where any subject has signed informed consent, regardless of whether the study is completed or prematurely terminated. Where appropriate an abbreviated report may be prepared. The CSR will comply with any applicable regulatory requirements, national laws in force and will be in English.

As indicated in Section 12.2.8, a full analysis based on all data recorded until the last subject completed their EOCT visit will be performed. This analysis will be included in a CSR. An addendum to the report will be prepared, including all data recorded from the follow-up period and additional cycles that occurred after the first data cut-off.

# **12.3** Statistical Methodology for Phase II

# 12.3.1 Sample Size Calculation

In case the antitumour activity is driven by a type of tumour, tumour-specific phase II cohort(s) will be initiated utilising an Optimal Simon Two-Stage design [33] to further evaluate the antitumour activity of <sup>177</sup>Lu-3BP-227. This approach tends to minimise the average number of subjects exposed to a new drug while addressing the scientific question.

The corresponding hypotheses (H) to be tested are:  $H_0$ : ORR  $\leq$ ORR<sub>0</sub> versus  $H_1$ : ORR>ORR<sub>0</sub> where ORR is the true objective response rate following <sup>177</sup>Lu-3BP-227 treatment and ORR<sub>0</sub> is the minimum objective response rate that would be considered to further clinical development.

For the purpose of the a priori sample size calculation, the following values are used for the PDAC cohort:  $ORR_0$  is 17% and ORR is 35%. These values reflect the results from PDAC therapy [34] currently registered and are considered standard-of-care, and the level of expectation for the IMP prior to phase I.

Similarly, for the CRC cohort (blended population with regard to somatic mutations in BRAF and RAS genes), ORR<sub>0</sub> is 15% and ORR is 30% [35;36]. The value for ORR<sub>0</sub> reflects an estimated average result from different clinical studies that investigated CRC therapies in different populations [35, 36]; it is anticipated that based on lack of available and effective therapies an enrichment of poor-prognosis subjects with metastatic and recurrent CRC and prevalent mutations in BRAF and RAS will occur in the study population, while it is anticipated that it is unlikely for subjects with wild-type disease for which effective therapies exist over multiple lines of treatment to be enrolled. Subjects with somatic BRAF and RAS mutations are anticipated to have less exposure to prior treatment for metastatic disease compared with subjects with prevalent wild type BRAF or RAS tumours.

The assumptions forming the understanding of the expected efficacy of standard-of-care treatment and leading to sample size estimations may be updated based on results emerging from phase I and evolving scientific knowledge.

In the first stage of the design, for the PDAC cohort, 16 subjects will be treated and evaluated for the ORR. If there are three or fewer responders in these 16 subjects, the cohort will be stopped for futility. Otherwise, 28 additional subjects will be treated and evaluated for a total of 44 subjects. The null hypothesis will be rejected if 12 or more responders are observed in the 44 subjects. This design yields a nominal one-sided type I error rate of 5% and a power of 80%, if the true response rate is 35%.

Similarly, in the first stage of the design, for the CRC cohort 19 subjects will be treated and evaluated for the ORR. If there are three or fewer responders in these 19 subjects, the cohort

will be stopped for futility. Otherwise, 36 additional subjects will be treated and evaluated for a total of 55 subjects. The null hypothesis will be rejected if 13 or more responders are observed in the 55 subjects. This design yields a nominal one-sided type I error rate of 5% and a power of 80%, if the true response rate is 30%.

Nonevaluable subjects will be replaced to meet the above numbers of evaluable subjects. A subject will be considered as nonevaluable and therefore replaced, if no evaluation post treatment is available or in case of drop-out before the 4-month visit for PDAC cohort and 8-month visit for CRC cohort with an insufficient follow-up duration (i.e. status being still "Stable Disease" as per RECIST version 1.1). In case of overenrollment, the first 16 evaluable subjects in the PDAC cohort and 19 subjects in the CRC cohort will be analysed at Stage 1 to make the decision about starting Stage 2. Assuming a non-evaluability rate of 20% for the PDAC cohort [34] and the CRC cohort [36], the number of subjects to be enrolled is expected to be approximately 55 and 70, respectively.

One, two or three further cohorts may be initiated (subject to results emerging from ongoing preclinical studies and antitumour efficacy seen during dose escalation and amending the current protocol) likely to enrol approximately 120 subjects with GC, SCCHN and/or ES.

In the case of acceptable tolerability and evident antitumour activity across all enrolled subjects in phase I, a phase II basket trial design will be utilised to study the antitumour activity of <sup>177</sup>Lu-3BP-227 in subjects with NTSR1 expressing tumours. The sample size will be described as part of a protocol amendment.

Subjects in each phase of the study will have a 5-year safety follow-up after the EOCT, ED or EOAC visit. The dose, the number and length of cycles in phase II will be refined according to phase I results.

However, if the antitumour activity is driven by a type of tumour, tumour-specific phase II cohort(s) will be initiated utilising an Optimal Simon's Two Stage design (see Phase II).

# 12.3.2 Analysis Sets

All subjects who receive at least one dose of the study drug will be included in the safety evaluation. All subjects who receive at least one dose of study drug and who have postbaseline efficacy data for the ORR will be included in the efficacy evaluations of the primary endpoint. All subjects who receive at least one dose of study drug and who have any postbaseline secondary efficacy data will be included in the efficacy evaluations of the secondary endpoints.

## 12.3.3 Demographic and Other Baseline Characteristics

Summary statistics will be presented for all treated subjects. Frequency tables for qualitative data will be provided. Medical and surgical history findings will be summarised using MedDRA terms.

## 12.3.4 Efficacy Evaluation

Tumour response will be assessed in imaging studies of CT or MRI scans after cycle 2 and subsequently every 8 weeks for the first 6 months and every 12 weeks thereafter.

The primary endpoint is ORR and will be calculated as defined in RECIST version1.1 and in Section 12.2.7. As part of secondary efficacy endpoints, DCR, TTP, Time to response (TTR), DOR, PFS and OS will also be determined as defined by RECIST version 1.1, metabolic response will also be determined as defined by PERCIST version 1.0.

In the particular context of the Simon Two-Stage approach, ORR will be analysed at the end of Stage 1 (and no later than after the 16-week visit of the last evaluable subject of the Stage 1 cohort for each PDAC and CRC cohorts). If the observed number of responders is below a predefined threshold, this part of the study will be stopped for futility. Otherwise, additional subjects will be treated to complete the planned enrolment. At the end of Stage 2, the null

hypothesis will be rejected depending on the total observed number of responders based on a predefined threshold.

At the end of the phase II, descriptive summaries will be provided for all primary and secondary efficacy endpoints. For the primary endpoint, final analysis will take into account the sequential sampling procedure of the design and the underlying binomial distribution assumed by the Simon Two-Stage design.

## 12.3.5 Safety Evaluation

Safety analyses similar to phase I (see Section 12.2.5) will be performed for phase II, except for DLT assessments and 24-hour 3-lead ECG Holter measurements.

# 13 DATA HANDLING AND RECORD KEEPING

# **13.1** Data Collection

In compliance with Good Clinical Practice (GCP), the source data, i.e. medical records/medical notes, etc. should be clearly marked and permit easy identification of a subject's participation in the specified clinical study.

Source data identification and location, whether standalone documents or direct eCRF records, will be specified in a standalone document signed by the investigator(s).

The investigator must record all data relating to protocol procedures, study drug administration, laboratory data, safety, PK and pharmacodynamic data on the source documents and report requested data on eCRFs provided for the study (see Section 13.2).

To ensure accurate, complete and reliable data, the sponsor or its representative will provide instructional material to the study site(s), as appropriate. Training will be given during a start-up/initiation meeting for instructions on the completion/data entry of any source data documents and eCRFs.

The investigators or their designees must verify that all data entries in the eCRF are accurate and consistent with source data records . If certain information is not available for a particular timepoint and/or subject, specific instructions should be followed, e.g. to document that the procedure was either not done or not applicable.

For all subjects who received any administration of <sup>177</sup>Lu-3BP-227 the full obtained data set needs to be recorded in the eCRF.

For all screening failures who did not receive the screening IMP administration, a minimum of data including the primary reason for screening failure will be recorded in the eCRF.

## **13.2** Data Reporting

Electronic data capture (EDC) will be utilised for collecting subject data. The study site is required to have a computer and internet connection available for study site entry of clinical data. All entries in the eCRF will be made under the electronic signature of the person performing the action. This electronic signature consists of an individual and confidential username and password combination. It is declared to be the legally binding equivalent of the handwritten signature. Only sponsor authorised users will have access to the eCRF as appropriate to their study responsibilities. Users must have successfully undergone software application training prior to entering data into the eCRF.

## 13.3 Data Management

Details of all data management procedures, from the initial planning to the archiving of final data sets/documents following database freeze/lock will be documented in appropriate standalone data management and validation plan(s).

Data management will be conducted by a CRO approved by the sponsor. All data management procedures will be completed in accordance with the contracted CRO's standard operating procedures. Prior to data becoming available for processing at the assigned data management CRO, they will be monitored.

The sponsor will ensure that an appropriate eCRF is developed to capture the data accurately and that suitable queries are raised to resolve any missing or inconsistent data. The investigator will receive the data from the clinical study in an electronic format (PDF files), which will be an exact copy of the eCRF and will include the full audit trail, for archiving purposes and future reference.

Any queries generated during the data management process will also be tracked by the contracted data management CRO. It is the central study monitor's responsibility to ensure that all queries are resolved by the relevant parties.

The CRO will also ensure, via SAE reconciliation, that SAE data collected in the eCRF are consistent with SAE data held in the sponsor's GPS department (and vice versa).

The coding of AE, medical history and prior/concomitant medication/nondrug therapy terms will be performed by the sponsor's Central Group. Prior/Concomitant medications will be coded using the latest version of the WHO drug dictionary and AEs/medical history/nondrug therapy terms will be coded using the latest version of MedDRA.

## 13.4 Record Keeping

The investigator will keep records of all original source data. This might include laboratory tests, medical records and clinical notes.

During the prestudy and initiation visits, the monitor must ensure the archiving facilities are adequate and archiving/retention responsibilities of the investigator have been discussed.

Study documents should be retained until at least 2 years after the last approval of a marketing application in an ICH region and until there are no pending or planned marketing applications in an ICH region (that is at least 15 years) or at least 2 years have elapsed since the formal discontinuation of clinical development of the product. However, these documents should be retained for a longer period if required by applicable regulatory requirements or by an agreement with the sponsor. The investigator should take measures to prevent accidental or premature destruction of these documents. The final archiving arrangements will be confirmed by the monitor when closing out the study site. The sponsor will inform the investigator, in writing, as to when these documents no longer need to be retained.

If the principal investigator relocates or retires, or otherwise withdraws responsibility for maintenance and retention of study documents, the sponsor must be notified (preferably in writing) so that adequate provision can be made for their future maintenance and retention.

# 14 **REGULATORY AND ETHICAL CONSIDERATIONS**

## **14.1 Regulatory Considerations**

The study will be conducted in compliance with IECs/IRBs, informed consent regulations, the Declaration of Helsinki and ICH Guidelines related to GCP. Any episode of noncompliance will be documented. The EDC system will comply with the FDA, 21 Code of Federal Regulations Part 11, Electronic Records, Electronic Signatures, and FDA, Guidance for Industry: Computerized Systems Used in Clinical Trials.

In addition, the study will adhere to all applicable international and local regulatory requirements.

All or some of the obligations of the sponsor will be assigned to a CRO.

An identification code assigned to each subject will be used in lieu of the subject's name to protect the subject's identity when reporting AEs and/or other trial-related data (see Section 4.6.1.1).

## 14.2 Ethical Review Considerations

The following documents should be submitted to the relevant ethics committee(s) (EC) for review and approval to conduct the study (this list may not be exhaustive):

- protocol/amendment(s) approved by the sponsor
- currently applicable IB or package labelling
- relevant investigator's curriculum vitae
- subject information and informed consent document(s) and form(s)
- subject emergency study contact cards
- recruitment procedures/materials (advertisements), if any.

The EC(s) will review all submission documents as required, and a written favourable opinion for the conduct of the study should be made available to the investigator before initiating the study. This document must be dated and clearly identify the version number(s) and date(s) of the documents submitted/reviewed and should include a statement from the EC that they comply with GCP requirements.

The study may begin at the investigative site(s) only after receiving this dated and signed documentation of the EC approval or favourable opinion.

During the study, any update to the following documents will be sent to the EC either for information, or for review and approval, depending on how substantial the modifications are: (1) the IB; (2) reports of SAEs; (3) all protocol amendments and revised informed consent(s), if any.

At the end of the study, the EC will be notified about the study completion.

# 14.3 Subject Information Sheet and Consent

The investigator is responsible for ensuring that the subject understands the potential risks and benefits of participating in the study, including answering, orally and/or in writing, to any questions the subject may have throughout the study and sharing any new information that may be relevant to the subject's willingness to continue his or her participation in the study in a timely manner.

The subject information sheet and consent document will be used to explain the potential risks and benefits of study participation to the subject in simple terms before the subject is entered into the study and to document that the subject is satisfied with his or her understanding of the study and desires to participate. The investigator is ultimately responsible for ensuring the EC-approved informed consent is appropriately signed and dated by each subject prior to the performance of any study procedures. Informed consent obtained under special circumstances may occur only if allowed by local laws and regulations.

The study has the option for subjects to consent to the collection of serum and whole blood samples for biobanking for future exploratory analysis and storage for up to 15 years (where local regulations allow). A specific informed consent is required for the collection of these samples and will be explained after the subject has given written informed consent for the main study.



# 14.4 Final Report Signature

The coordinating investigator or designee will be proposed to review and sign the clinical study report for this study, indicating agreement with the analyses, results, and conclusion of the report.

## 15 INSURANCE AND FINANCE

### 15.1 Insurance

The sponsor declares that it has taken out a product liability insurance covering all subjects screened and enrolled in this study in respect to risks involved in the study.

# 15.2 Financial Agreement

Since this study is to be performed in partnership with a CRO, separate financial agreements between the sponsor and the CRO on one side, and the CRO and the investigator site on the other side, will be signed prior to initiating the study, outlining overall sponsor and investigator responsibilities in relation to the study.

# 16 QUALITY CONTROL AND QUALITY ASSURANCE

To ensure accurate, complete, and reliable data, the sponsor or its representatives will provide instructional material to the study sites, as appropriate. A startup training session will be done prior to screening start to instruct the investigators and study coordinators. This session will give instruction on the protocol, the completion of the eCRF and all study procedures.

# **16.1 Protocol Amendments and Protocol Deviations and Exceptions**

## 16.1.1 Protocol Amendments

In the event that an amendment to this protocol is required, it will be classified into one of the following three categories:

- *nonsubstantial amendments* are those that are not considered 'substantial' (e.g. administrative changes) and as such only need to be notified to the IECs or regulatory authorities for information purposes;
- *substantial amendments* are those considered 'substantial' to the conduct of the clinical study where they are likely to have a significant impact on the:
  - Safety or physical or mental integrity of the subjects;
  - Scientific value of the study;
  - Conduct or management of the study, or
  - Quality or safety of the study drug used in the study.

Substantial amendments must be submitted to and approved by the IECs and relevant regulatory authorities, according to local regulations, prior to implementing changes.

• *urgent amendments* are those that require urgent safety measures to protect the study subjects from immediate hazard and as such may be implemented immediately by the sponsor with subsequent IECs and regulatory authority notification, forthwith.

The principal investigator and the sponsor will sign the protocol amendment.

# 16.1.2 Protocol Deviations and Exceptions

Protocol deviations are defined and classified as either major or minor for a given study. Major deviations (or a combination of minor becoming major) may or may not impact the analysis population. All minor and major protocol deviations will be identified and recorded by the investigator site personnel and should be traceable.

## Major protocol deviation definition:

Any changes in the study design, study conduct and/or procedures that are not in accordance with the protocol and any study materials originally approved by the IEC and which may affect the subject's rights, safety or wellbeing, or the completeness, accuracy and reliability of the study data. A major protocol deviation is any significant divergence from the protocol, i.e. nonadherence on the part of the subject, the investigator, or the sponsor to protocol specific inclusion/exclusion criteria, primary objective evaluation criteria and/or GCP guidelines.

Generally, a protocol deviation qualifies as major if:

- (1) The deviation has harmed or posed a significant or substantive risk of harm to the research subject.
- (2) The deviation compromises the scientific integrity of the data collected for the study.
- (3) The deviation is a wilful or knowing breach of human subject protection regulations, policies, or procedures on the part of the investigator(s).
- (4) The deviation involves a serious or continuing noncompliance with any applicable human subject protection regulations, policies, or procedures.

(5) The deviation is inconsistent with Ipsen's research, medical and ethical principles.

# Minor protocol deviation definition:

Any changes in the study design, study conduct and/or procedures that are not in accordance with the protocol and any study materials originally approved by the IEC but that do not have an important impact on the subject's rights, safety or well-being, or the completeness, accuracy and reliability of the study data. A minor protocol deviation is any significant divergence from the protocol that does not impact the study results.

As a matter of policy, the sponsor will not grant exceptions to protocol-specific entry criteria to allow subjects to enter a study. If under extraordinary circumstances such action is considered ethically, medically and scientifically justified for a particular subject, prior approval from the sponsor and the responsible IRB/IEC, in accordance with the standard operating procedure (SOP), is required before the subject is allowed to enter the study.

If investigative centre personnel learn that a subject who did not meet the protocol eligibility criteria was entered in a study (a protocol deviation), they must immediately inform the sponsor. Such subjects will be discontinued from the study, except in exceptional instance, following review and written approval by the sponsor and the responsible IRB/IEC, according to the applicable SOP. Retention of these subjects in the study will be discussed between sponsor and investigator, taking into account subject safety and data reliability. The IRB/IEC will be informed if subject safety/protection is ignorantly impacted.

# 16.1.3 Information to Study Personnel

The investigator is responsible for giving information about the study to all staff members involved in the study or in any element of subject management, both before starting any study procedures and during the course of the study (e.g. when new staff become involved).

The investigator must assure that all study staff members are qualified by education, experience, and training to perform their specific responsibilities. These study staff members must be listed on the clinical unit authorisation form, which includes a clear description of each staff member's responsibilities. This list must be updated throughout the study, as necessary.

The study monitor is responsible for explaining the protocol to all study staff, including the investigator, and for ensuring their compliance with the protocol. Additional information will be made available during the study when new staff become involved in the study and as otherwise agreed upon with either the investigator or the study monitor.

# 16.2 Monitoring

The investigator is responsible for the validity of all data collected at the site.

The sponsor is responsible for monitoring these data to verify that the rights and wellbeing of subjects are protected, study data are accurate (complete and verifiable to source data), and that the study is conducted in compliance with the protocol, GCP, and regulatory requirements.

Before the study initiation visit, the sponsor-assigned study monitor will write a monitoring plan indicating the monitoring procedures and at which occasions during the study monitoring visits will be performed.

Periodic visits will be made to the study site throughout the study at mutually agreeable times. Any appropriate communication tools will be set-up to ensure the sponsor and/or its representative is/are available for consultation, so they can stay in contact with the study site personnel.

Adequate time and space for monitoring visits should be made available by the investigator.

The investigator will allow direct access to all relevant files (for all subjects) and clinical study supplies (dispensing and storage areas), for the purpose of verifying entries made in the eCRF, and assist with the monitor's activities, if requested.

Quality of the paper-based or electronic data will be reviewed to detect errors in data collection and, if necessary, to verify the quality of the data.

The eCRF is expected to be completed on an ongoing basis to allow regular review by the study monitor, both remotely by the internet and during site visits. The study monitor will use functions of the EDC system to address any queries raised while reviewing the data entered by the study site personnel in a timely manner.

Whenever a subject name is revealed on a document required by the sponsor (e.g. laboratory print outs) the name must be blacked out permanently by the site personnel, leaving the date of birth visible and annotated with the subject number as identification.

# 16.3 Investigator's Regulatory Obligations

All clinical work under this protocol will be conducted according to GCP rules. This includes that the study may be audited at any time by quality assurance personnel designated by the sponsor, or by regulatory bodies. The investigator must adhere to the GCP principles in addition to any applicable local regulations.

If requested, the investigator will provide the sponsor, applicable regulatory agencies, and applicable EC with direct access to any original source documents.

The investigator(s) should demonstrate due diligence in recruitment and screening of potential study subjects. The enrolment rate should be sufficient to complete the study as agreed with the sponsor. The sponsor should be notified of any projected delays, which may impact the completion of the study.

# 16.3.1 Audit and Inspection

Authorised personnel from external CAs and the sponsor's authorised quality assurance personnel may carry out inspections and audits.

# 16.3.2 Data Quality Assurance

Monitored eCRF, transferred from the investigator site to the assigned data management group, will be reviewed (secondary monitoring) for completeness, consistency and protocol compliance.

Reasons should be given in the relevant eCRF for any missing data and other protocol deviations. Any electronic queries and items not adequately explained will require additional electronic manual queries to be raised to the investigator for clarification/correction. The investigator must ensure that queries are dealt with promptly. All data changes and clarifications can be viewed in the audit trail function of the eCRF.

## **17 PUBLICATION POLICY**

The sponsor encourages acknowledgement of all individuals/organisations involved in the funding or conduct of the study, including medical writers or statisticians subject to the consent of each individual and entity concerned, including acknowledgement of the sponsor.

The results of this study may be published or communicated to scientific meetings by the investigators involved in the study. For multicentre studies, a plan for scientific publication and presentation of the results may be agreed and implemented by the study investigators or a steering committee. The sponsor requires that reasonable opportunity be given to review the content and conclusions of any abstract, presentation, or paper before the material is submitted for publication or communicated. This condition also applies to any amendments that are subsequently requested by referees or journal editors. The sponsor will undertake to comment on the draft documents within the time period agreed in the contractual arrangements, including clinical trial agreements, governing the relationship between the sponsor and authors (or the author's institution). Requested amendments will be incorporated by the author, provided they do not alter the scientific value of the material.

If patentability could be adversely affected by publication, this will be delayed until (i) a patent application is filed for the content of the publication in accordance with applicable provisions of the clinical trial agreement concerned, (ii) the sponsor consents to the publication, or (iii) the time period as may be agreed in the contractual arrangements, including clinical trial agreements, governing the relationship between the sponsor and authors (or authors' institution) after receipt of the proposed publication by the sponsor, whichever of (i), (ii) or (iii) occurs first.

The author undertakes to reasonably consider the sponsor's request for delay to the proposed publication should the sponsor reasonably deem premature to publish the results obtained at that particular stage of the study.

## **18 REFERENCES**

- 1. Kirschner AS, Ice RD Beierwaltes WH. Radiation dosimetry of <sup>131</sup>I-19-iodocholesterol: the pitfalls of using tissue concentration data -- reply. J Nucl Med 1975;16(3):248-9.
- 2. Schulz J, Rohracker M, Stiebler M et al. Proof of therapeutic efficacy of a novel 177Lulabeled neurotensin receptor 1-antagonist in a colon carcinoma xenograft model. J Nucl Med 2017;58(6):936-41.
- 3. Baum RP, Singh A, Schuchardt C et al. 177Lu-3BP-227 for Neurotensin Receptor 1 Targeted Therapy of Metastatic Pancreatic Adenocarcinoma: First Clinical Results. J Nucl Med. 2018 May;59(5):809-814.
- 4. Körner M, Waser B, Strobel O et al. Neurotensin receptors in pancreatic ductal carcinomas. EJNMMI Res 2015;5:17.
- 5. Wang JG, Li NN, Li HN et al. Pancreatic cancer bears overexpression of neurotensin and neurotensin receptor subtype-1 and SR 48692 counteracts neurotensin induced cell proliferation in human pancreatic ductal carcinoma cell line PANC-1. Neuropeptides 2011;45(2):151-6.
- 6. Reubi JC, Waser B, Schaer JC, Laissue JA. Neurotensin receptors in human neoplasms: high incidence in Ewing's sarcomas. Int J Cancer 1999; 82(2):213-8.
- 7. Kwekkeboom DJ, de Herder WW, Kam BL et al. Treatment with the radiolabeled somatostatin analog [<sup>177</sup> Lu-DOTA<sup>0</sup>,Tyr<sup>3</sup>] octreotate: toxicity, efficacy, and survival. J Clin Oncol 2008;26(13):2124-30.
- 8. Strosberg J, El-Haddad G, Wolin E et al. Phase 3 trial of <sup>177</sup>Lu-DOTATATE for Midgut Neuroendocrine Tumors. N Engl J Med 2017;376(2):125-35.
- 9. Guédé D, Reigner B, Vandenhende F et al. Bayesian adaptive designs in single ascending dose trials in healthy volunteers. Br J Clin Pharmacol 2014;78(2):393–400.
- 10. D'Onofrio M, Ciaravino V, De Robertis R et al. Percutaneous ablation of pancreatic cancer. World J Gastroenterol 2016;22(44):9661-73.
- 11. American Cancer Society. Pancreatic Cancer Survival Rates, by Stage. Last updated May 2016, [available at: https://www.cancer.org/content/cancer/en/cancer/pancreatic-cancer/detection-diagnosis-staging/survival-rates.html, accessed on 2017 June 26].
- 12. Rahib L, Smith BD, Aizenberg R et al. Projecting cancer incidence and deaths to 2030: the unexpected burden of thyroid, liver, and pancreas cancers in the United States. Cancer Res 2014;74(11):2913–21.
- 13. Kleeff J, Michl P. Targeted therapy of pancreatic cancer: biomarkers are needed. Lancet Oncol 2017;18(4):421-2.
- 14. Previdi MC, Carotenuto P, Zito D et al. Noncoding RNAs as novel biomarkers in pancreatic cancer: what do we know? Future Oncol 2017;13(5):443-53.
- 15. Van Cutsem E, Cervantes A, Adam R et al. ESMO consensus guidelines for the management of patients with metastatic colorectal cancer. Ann Oncol 2016;27(8):1386-422.
- 16. Fakih MG. Metastatic colorectal cancer: current state and future directions. J Clin Oncol 2015;33(16):1809-24.
- 17. King GT, Lieu CH, Messersmith WA. Frontline Strategies for Metastatic Colorectal Cancer: New Sides to the Story. AJHO 2016;12(10):4-11.
- 18. Howlader N, AM N, M K, et al. SEER Cancer Statistics Review, 1975-2012, National Cancer Institute, Bethesda, MD. [available at: https://seer.cancer.gov/archive/csr/1975\_2013/, accessed on 2017 June 26].

- 19. Khatri VP, Petrelli NJ, Belghiti J. Extending the frontiers of surgical therapy for hepatic colorectal metastases: is there a limit? J Clin Oncol 2005;23(33):8490-9.
- 20. Clark JW, Grothey A. Systemic chemotherapy for metastatic colorectal cancer: Completed clinical trials. Last updated May 2017, [available at: http://www.uptodate.com/contents/systemic-chemotherapy-for-metastatic-colorectalcancer-completed-clinical-trials, accessed 2017 June 26].
- 21. Bodei L, Mueller-Brand J Baum RP et al. The joint IAEA, EANM, and SNMMI practical guidance on peptide receptor radionuclide therapy (PRRNT) in neuroendocrine tumours. Eur J Nucl Med Mol Imaging 2013;40(5):800-16.
- 22. Breeman W, de Blois E, Bakker W et al. Radiolabeling DOTA-Peptides with 90Y and 177Lu to a high specific activity. In: Radionucleide Peptide Cancer Therapy 1st edition. Michigan, USA: Taylor and Francis; 2006:119-25.
- 23. Food and Drug Administration, Center for Drug Evaluation and Research (CDER). Guidance for Industry. Estimating the maximum safe starting dose in initial clinical trials for therapeutics in adult healthy volunteers. July 2005 [available at: https://www.fda.gov/downloads/drugs/guidances/ucm078932.pdf, accessed 2017 Jul 04].
- 24. Wild D, Fani M, Fischer R et al. Comparison of somatostatin receptor agonist and antagonist for peptide receptor radionuclide therapy: a pilot study. J Nucl Med 2014;55(8):1248–52.
- 25. Gui X, Guzman G, Dobner PR et al. Increased neurotensin receptor-1 expression during progression of colonic adenocarcinoma. Peptides 2007;29(9):1609-15.
- 26. Shimizu S, Tsukada J, Sugimoto T et al. Identification of a novel therapeutic target for head and neck squamous cell carcinomas: a role for the neurotensin-neurotensin receptor 1 oncogenic signaling pathway. Int J Cancer 2008;123(8):1816-23.
- 27. Zhou Z, Xie J, Cai Y et al. The significance of NTR1 expression and its correlation with β-catenin and EGFR in gastric cancer. Diagn Pathol 2015;10:128.
- 28. Cassady JR. Clinical radiation nephropathy. Int J Radiat Oncol Biol Phys 1995;31(5):1249-56.
- 29. Emami B, Lyman J, Brown A et al. Tolerance of normal tissue to therapeutic irradiation. Int J Radiat Oncol Biol Phys 1991;21(1):109–22.
- National Council on Radiation Protection and measurements. Misadministration of radioactive material in medicine – Scientific background, 1991, [available at: http://public.eblib.com/choice/publicfullrecord.aspx?p=3382330, accessed: 2017 Jul 04].
- National Cancer Institute. Common Terminology Criteria for Adverse Events (CTCAE), Version 5.0, 2018 [available at: https://evs.nci.nih.gov/ftp1/CTCAE/CTCAE\_5.0/ accessed 28 February 2018]
- 32. Lassmann M, Chiesa C, Flux G et al. EANM Dosimetry Committee guidance document: good practice of clinical dosimetry reporting. Eur J Nucl Med Mol Imaging 2011;38(1):192-200.
- 33. Simon R. Optimal two-stage designs for phase II clinical trials. Control Clin Trials 1989;10(1):1-10.
- Wang-Gillam A, Li CP, Bodoky G et al. Nanoliposomal irinotecan with fluorouracil and folinic acid in metastatic pancreatic cancer after previous gemcitabine-based therapy (NAPOLI-1): a global, randomised, open-label, phase 3 trial. Lancet 2016;387(10018):545-57.

- 35. Mayer RJ, Van Cutsem E, Falcone A et al. Randomized trial of TAS-102 for refractory metastatic colorectal cancer. N Engl J Med 2015;372(20):190919.
- 36. Cunningham D, Humblet Y, Siena S et al. Cetuximab monotherapy and cetuximab plus irinotecan in irinotecan-refractory metastatic colorectal cancer. N Engl J Med 2004;351(4):337-45.
- 37. Tibaldi FS, Beck BHL, Bedding AW et al. Implementation of a Phase 1 adaptive clinical trial in a treatment of type 2 diabetes. Drug Inf J 2008;42:455-65.
- 38. Wahl RL, Jacene H, Kasamon Y, Lodge MA. From RECIST to PERCIST: Evolving Considerations for PET response criteria in solid tumors. J Nucl Med 2009;50(1):122-50.

# **19 ATTACHMENTS**

**19.1** Attachment 1 – Study Schedule of Assessments for Phase I Dose Escalation

### Table 7 Schedule of Assessments for Phase I – For Subjects who are Screen Failures after IMP Screening Administration

| Procedures and assessments                         | IMP Screening<br>Discharge Visit [a] | IMP Screening Follow-<br>up Visit [b] |
|--|--------------------------------------|---------------------------------------|
| AEs  | Х                                    | Х                                     |
| Physical examination                               | Х                                    | Х                                     |
| Vital signs  | Х                                    | Х                                     |
| Haematology and biochemistry [c]                   | Х                                    | Х                                     |
| Urinalysis   | Х                                    | Х                                     |
| Prior/concomitant medication/therapy               | X                                    | X                                     |
| <sup>177</sup> Lu-3BP-227 screening administration | Х                                    |                                       |

a before site discharge

b 5 weeks after the IMP Screening Discharge Visit +2 weeks. If the subject is not able to come for an onsite visit, the visit may be performed by phone call. The clinical laboratory tests may be done locally. If the subject is not available for a visit or a call, the investigator can contact his oncologist/general practitioner to get the data requested in protocol.

c additional safety assessments including haematology and biochemistry can be done if clinically indicated

| Procedures and  | CORE TRIAL [v]         |           |   |    |  |       |     |    |     | EOCT               | Long-term |   |    |         |      |    |    |                    |        |           |
|---|------------------------|-----------|---|----|--|-------|-----|----|-----|--------------------|-----------|---|----|---------|------|----|----|--------------------|--------|-----------|
| assessments   | Screening              |           |   |    |  |       |     |    | Tre | atment             | Period    | I |    |         |      |    |    |                    | or     | Follow-up |
|   |                        |           |   |    |  | Cycle | e 1 |    | -   |                    |           |   |    | <br>Cyc | le 2 |    |    |                    | ED [a] | [b]       |
|   | D-21 to D-1<br>[c]     | D1<br>[d] | D1     D2     D3     D4     D5     D7     D 15     D 22     D29     D1     D2     D3     D4     D5     D7     D 15     D 22     D 29       ij |    |  |       |     |    |     | D43                |           |   |    |         |      |    |    |                    |        |           |
|   | Visit window<br>(days) |           |   | +1 |  |       | +1  | ±1 | ±1  | +7<br>(+28)<br>[e] |           |   | +1 |         | +1   | ±1 | ±1 | +7<br>(+28)<br>[e] | +7     | ±14       |
| Informed consent  | Х                      |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    |        |           |
| Inclusion/exclusion criteria                              | Х                      | х         |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    |        |           |
| Subject demographics and height                           | Х                      |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    |        |           |
| Medical and disease history                               | Х                      |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    |        |           |
| ECOG performance status                                   | Х                      | х         |   |    |  |       |     |    |     |                    | Х         |   |    |         |      |    |    |                    | Х      |           |
| ceCT/MRI  | x [f]                  |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    | х      | Х         |
| <sup>18</sup> F-FDG-PET                                   |                        | x [g]     |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    | Х      |           |
| <sup>177</sup> Lu-3BP-227 screening<br>administration [h] | x                      |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    |        |           |
| Tumour biopsy   | x [i]                  |           |   |    |  |       |     |    |     |                    |           |   |    |         |      |    |    |                    | х      |           |

#### Table 8 Schedule of Assessments for Phase I – Dose Escalation

AE=adverse event; ceCT/MRI=contrast enhanced computed tomography/magnetic resonance imaging; cfDNA=cell-free deoxyribonucleic acid; D=Day; ECG=electrocardiogram; ECOG=Eastern Cooperative Oncology Group; ED=early discontinuation; EOCT=end of core trial; IMP=investigational medicinal product; NTSR1=neurotensin receptor 1; PET=positron emission tomography; PK=pharmacokinetic; SPECT=single photon emission computed tomography.

- a the ED visit will take place within 14 days after ED from the study and at the latest 5 weeks after the last IMP administration, the visit will correspond to an EOCT visit. For subjects who received the screening IMP administration and found ineligible to participate to the study, an ED visit will be performed within 5 weeks (+ 2 weeks) after the IMP screening administration to assess the safety. At least, clinical laboratory tests (e.g. haematology and biochemistry), AEs and concomitant medication/therapy will be collected. If the subject is not able to come for an on-site visit, the visit may be performed by phone call. The clinical laboratory tests may be done locally.
- b follow-up visits will take place every 3 months (±2 weeks) and will start after the EOCT, ED or EOAC visit. Efficacy will be assessed until disease progression, administration of any other chemotherapy or radiotherapy, lost to follow-up, withdrawal of consent, death or a maximum of 2 years, whichever occurs first (see Section 6.2.3). After disease progression is confirmed, no further CT/MRI scans will be required for tumour/disease assessment. The survival status and safety of the subjects will continue to be monitored as indicated in Section 8.1.2.3 until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first.
- c the screening period can be extended by two weeks if this is required for logistical reasons.
- d Day 1 of Cycle 1 is the day of the first treatment administration. Day 1 of Cycle 2 and each potential subsequent cycle may coincide with Day 29 of the previous cycle if all safety assessments are performed and allow for the next administration.
- e in case of toxicity requiring a delay, the next treatment administration will be delayed by up to 4 weeks.
- f whole body ceCT or ceMRI including brain unless a similar exam has already been performed within 1 month prior to Day -21.
- g Cycle 1 Day 1 <sup>18</sup>FDG-PET between Day -21 and Day 1.
- h screening administration will be done after confirmation of eligibility (i.e. after all other screening assessments have been performed), minimum 1 week before the first treatment administration.

| Procedures and   |                        |           | CORE TRIAL [V] |    |    |       |    |      |       |                    |           |    | EOCT | Long-term |     |           |     |     |                    |        |           |
|--|------------------------|-----------|----------------|----|----|-------|----|------|-------|--------------------|-----------|----|------|-----------|-----|-----------|-----|-----|--------------------|--------|-----------|
| assessments  | Samooning              |           |                |    |    |       |    | Trea | atmen | t Perio            | od        |    |      |           |     |           |     |     |                    | or     | Follow-up |
|  | Screening              |           |                |    | Су | cle 1 |    |      |       |                    |           |    |      |           | Сус | cle 2     |     |     |                    | ED [a] | [b]       |
|  | D-21 to D-1<br>[c]     | D1<br>[d] | D2             | D3 | D4 | D5    | D7 | D 15 | D 22  | D29                | D1<br>[d] | D2 | D3   | D4        | D5  | <b>D7</b> | D15 | D22 | D29                | D43    |           |
|  | Visit window<br>(days) |           |                | +1 |    |       | +1 | ±1   | ±1    | +7<br>(+28)<br>[e] |           |    | +1   |           |     | +1        | ±1  | ±1  | +7<br>(+28)<br>[e] | +7     | ±14       |
| <sup>177</sup> Lu-3BP-227 treatment<br>administration                        |                        | х         |                |    |    |       |    |      |       |                    | х         |    |      |           |     |           |     |     |                    |        |           |
| Blood sampling for<br><sup>177</sup> Lu-3BP-227 PK [j]                       |                        | Х         | х              | х  | 2  | x     |    |      |       |                    | х         | х  | х    | 2         | K   |           |     |     |                    |        |           |
| Planar scintigraphy [k]  | х                      | Х         | Х              | Х  | 2  | ĸ     | Х  |      |       |                    | х         | Х  | х    | 2         | ĸ   | х         |     |     |                    |        |           |
| SPECT/CT scan [k]  | х                      | Х         | х              | х  | 2  | x     | х  |      |       |                    | х         | х  | х    | 2         | ĸ   | х         |     |     |                    |        |           |
| Blood sampling for<br>3BP-227 PK [l]   |                        | Х         | х              | х  |    |       |    |      |       |                    |           |    |      |           |     |           |     |     |                    |        |           |
| Urine sampling for<br><sup>177</sup> Lu-3BP-227 PK [m] and<br>3BP-227 PK [t] |                        | Х         | X              | Х  |    |       |    |      |       |                    |           |    |      |           |     |           |     |     |                    |        |           |

i in case subject consents to have a biopsy and in case it can be accomplished with reasonable safety, a tumour biopsy will be taken during screening from the primary or metastatic lesion, whichever is accessible, ideally on lesions which are positive for NTSR1 on <sup>177</sup>Lu 3BP-227 SPECT/CT following IMP screening administration, as soon as <sup>177</sup>Lu uptake has been confirmed. If not, archival tissue from a previous tumour biopsy can be used for exploratory analysis (tumour microenvironment analysis and transcriptomics).

j eighteen blood samples will be collected during the treatment period. Blood samplings will be performed just before the <sup>177</sup>Lu-3BP-227 infusion (baseline), at the end of infusion (0), 5 minutes ±1 minute, 30 minutes ±5 minutes, 90 minutes ±15 minutes and 4 hours ±30 minutes, 24 hours ±2 hours, 48 hours ±2 hours and 72 to 96 hours post infusion.

k eleven whole body scans (planar scintigraphy) and SPECT/CT acquisitions will be performed during the treatment period. Whole body scans (planar scintigraphy) and SPECT/CT scan will be performed at the following timepoints just after the end of <sup>177</sup>Lu-3BP-227 infusion: Day 1: 4 (±2) hours, Day 2: 24 (±6) hours, Day 3: 48 (±6) hours, Day 4: 72 to 96 hours, and Days 7 to 8: 138 to 168 hours. Within each cycle, a single SPECT/standard dose CT will be performed. A SPECT/low dose CT will be performed at all other timepoints. Details of the procedures will be provided in the Image Acquisition Guidelines. At screening, planar scintigraphy (1 or 2 timepoint(s) at the investigator's discretion) and optional SPECT/CT scans (up to 2 at the investigator's discretion) will be performed after screening administration of <sup>177</sup>Lu-3BP-227.

1 ten blood samples will be collected at cycle 1. Blood samplings will be performed just before <sup>177</sup>Lu-3BP-227 infusion (baseline), at the end of infusion of <sup>177</sup>Lu-3BP-227 (0), 5 minutes ±1 minute, 30 minutes ±5 minutes, 90 minutes ±15 minutes and 4 hours ±30 minutes, 6 hours ±30 minutes, 8 hours ±1 hour, 24 hours ±2 hours and 48 hours ±2 hours after the end of infusion of <sup>177</sup>Lu-3BP-227.

m four urine samples will be collected during the treatment period at the following time periods and only for Cycle 1: from the start of the IMP infusion to 6 hours, 6 to 12 hours, 12 to 24 hours, and 24 to 48 hours (from the start of the IMP infusion to 6 hours after the end of the infusion only for US sites) after the end of <sup>177</sup>Lu-3BP-227 infusion.

| Procedures and CORE TRIAL [v] F         |                        |                |  |    |     |       |    |      | EOCT   | Long-term |       |    |    |     |     |       |     |     |     |        |           |
|---|------------------------|----------------|--|----|-----|-------|----|------|--------|-----------|-------|----|----|-----|-----|-------|-----|-----|-----|--------|-----------|
| assessments                             | Sereening              |                |  |    |     |       |    | Tre  | eatmei | nt Per    | iod   |    |    |     |     |       |     |     |     | or     | Follow-up |
|   | Screening              |                |  |    | Cyc | ele 1 |    |      |        | -         |       |    |    |     | Cyc | ele 2 |     |     |     | ED [a] | [b]       |
|   | D-21 to D-1<br>[c]     | D1 [d]         | D2   | D3 | D4  | D5    | D7 | D 15 | D 22   | D29       | D1[d] | D2 | D3 | D4  | D5  | D7    | D15 | D22 | D29 | D43    |           |
|   | Visit window<br>(days) |                | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ |    |     |       |    |      |        |           |       |    | +7 | ±14 |     |       |     |     |     |        |           |
| AEs                                     | Х                      | Х              | х  | Х  | Х   | Х     | х  | Х    | Х      | х         | Х     | х  | Х  | х   | х   | х     | Х   | Х   | х   | Х      | x [z]     |
| Physical examination                    | х                      | x [ <b>y</b> ] |  |    |     |       |    | Х    |        | х         | Х     |    |    |     |     |       | Х   |     | х   | х      |           |
| Vital signs                             | x [n]                  | x [n]          | Х  | Х  |     |       | х  | Х    | Х      | х         | x [n] | х  | х  |     |     | х     | Х   | Х   | х   | х      |           |
| Body weight                             | х                      | x [ <b>y</b> ] |  |    |     |       |    | Х    |        | х         | Х     |    |    |     |     |       | Х   |     | Х   | Х      |           |
| ECG (12-lead) [o]                       | Х                      | Х              | х  |    |     |       |    |      |        |           | Х     | х  |    |     |     |       |     |     |     | Х      |           |
| 24-hour 3-lead Holter ECG<br>[p]        |                        | Х              |  |    |     |       |    |      |        |           | х     |    |    |     |     |       |     |     |     |        |           |
| Haematology and biochemistry [w]        | Х                      | x [y]          |  |    |     |       | х  | х    | х      | х         | х     |    |    |     |     | х     | х   | х   | Х   | х      | х         |
| Urinalysis                              | х                      | x [ <b>y</b> ] | Х  | Х  |     |       |    |      |        | х         | Х     |    |    |     |     |       |     |     | х   | Х      |           |
| CCI                                     |                        |                |  |    |     |       |    |      |        |           |       |    |    |     |     |       |     |     |     |        |           |
| Prior/concomitant<br>medication/therapy | X                      | X              | Х  | х  | х   | х     | х  | x    | х      | х         | х     | х  | х  | х   | х   | х     | х   | х   | Х   | Х      | x         |

n prior to and at the end of <sup>177</sup>Lu-3BP-227 infusion (0) as well as 30±5 minutes, 90±15 minutes, 4 hours±30 minutes after the end of <sup>177</sup>Lu-3BP-227 infusion.

o during the screening period, a triplicate 12-lead ECG will be recorded before the screening administration (minus 15 minutes), and a single 12-lead ECG will be recorded at the end of <sup>177</sup>Lu-3BP-227 infusion (±15 minutes) and 4 hours (±30 minutes) after the end of infusion. At each treatment administration, a triplicate 12-lead ECG will be recorded on Day 1 before the infusion (baseline) (minus 15 minutes) and a single 12-lead ECG recordings at the end of <sup>177</sup>Lu-3BP-227 infusion (±15 minutes), at 4 hours after the end of <sup>177</sup>Lu-3BP-227 infusion (±30 minutes) and on Day 2 at 24 hours after the end of <sup>177</sup>Lu-3BP-227 infusion (±4 hours) as well as at EOCT. All 12-lead computerised standard ECGs will be recorded in the supine position after at least 5 minutes of rest.

p starting before <sup>177</sup>Lu-3BP-227 infusion, a 24-hour 3-lead continuous ECG Holter will be recorded.

| Procedures and   |  | CORE TRIAL [v]  |   |  |   |  |  |   |   |   |                     |                          | EOCT                                      | Long-term  |  |   |  |   |  |   |   |
|--|--|---|---|--|---|--|--|---|---|---|---------------------|--------------------------|---|--|--|---|--|---|--|---|---|
| assessments  | Screening  |   |   |  |   |  |  | Tre   | eatmen  | t Peri  | od                  |                          |   |  |  |   |  |   |  | or  | Follow-up   |
|  | Sereening  |   |   |  | Сус   | le 1   |  | 1   |   |   |                     |                          |   |  | Cyc  | le 2  |  |   |  | ED [a]  | [b]   |
|  | D-21 to D-1<br>[c]   | D1 [d]  | D2  | D3   | D4  | D5   | D7   | D 15  | D 22  | D29   | D1[d]               | D2                       | D3  | D4   | D5   | D7  | D15  | D22   | D29  | D43   |   |
|  | Visit window<br>(days)   |   |   | +1   |   |  | +1   | ±1  | ±1  | +7<br>(+28)<br>[e]  |                     |                          | +1  |  |  | +1  | ±1   | ±1  | +7<br>(+28)<br>[e]   | +7  | ±14   |
| Pregnancy test [q]   | Х  | Х   |   |  |   |  |  |   |   |   | х                   |                          |   |  |  |   |  |   |  | х   |   |
| Hormone analysis [r]   | Х  |   |   |  |   |  |  |   |   |   |                     |                          |   |  |  |   |  |   |  | Х   |   |
| CCI  |  | v   |   |  |   |  |  |   |   |   | v                   |                          |   |  |  |   |  |   |  | v   | _   |
|  |  | Λ   |   |  |   |  |  |   |   |   | Λ                   |                          |   |  |  |   |  |   |  | Λ   |   |
| Survival status  |  |   |   |  |   |  |  |   |   |   |                     |                          |   |  |  |   |  |   |  |   | X   |
| <ul> <li>q seruin pregnancy test with sample for hormone anal CCI</li> <li>t four urine samples will be hours, and 24 to 48 hours</li> <li>u CCI</li> <li>v if an additional cohort of therapy is to be the same additional safety assessm</li> <li>x urine samples will be con 177Lu-3BP-227 infusion</li> <li>y on Day 1, assessments to Any AEs/SAEs, irrespect first. After this timepoin investigator.</li> </ul> | vsis (NTSR1 or<br>pe collected duri<br>s after the start of<br>s subjects is recr<br>as the schedule<br>hents including l<br>llected (and froz<br>at the latest) and<br>b be done predos<br>tive of causality<br>t, up to the end of | ng the treat<br>of <sup>177</sup> Lu-3B<br>uited to rec<br>of assessm<br>haematolog<br>ten) at the ful<br>EOCT/ED<br>se and up to<br>to ful the 5-yea | ment peri<br>P-227 inf<br>eive 3 cyc<br>ents for C<br>y and bio<br>ollowing<br>cearly m<br>48 hours<br>reported t<br>r follow-1 | cles of<br>Cycle 2<br>Cycle 2<br>Cycle 2<br>Cycle 2<br>Cycle 3<br>Cycle | the follo<br>from the follo<br>from the follo<br>therapy<br>stry can<br>bints: C<br>e IMP i<br>ponsor<br>od, AE | y as des<br>be don<br>y as des<br>be don<br>ycle 1:<br>nfusion<br>up to 6<br>s/SAE | scribed<br>ne if cl<br>Day 1<br>n.<br>6 montl<br>s shoul | in the<br>infusion<br>in the<br>inically<br>: early<br>as after<br>d only | nd only<br>on to 6 J<br>Study I<br>/ indica<br>mornin | Tor Cy<br>nours a<br>Design<br>ted.<br>g, befo<br>pCT, El<br>orted if | re the i<br>D or E0 | from<br>escala<br>infusi | the s<br>of th<br>ation<br>visit<br>evalu | tart o<br>e infu<br>part 1<br>paseli<br>or ur<br>uated | f the<br>ision<br>Phase<br>ne), I<br>as re | IMP<br>only<br>I), t<br>Øay 3<br>w ar<br>ated | infusio<br>for US<br>he scho<br>3: early<br>titumo | on to 6<br>S sites),<br>edule o<br>v morni<br>our trea<br>IMP o | hours, 6<br>f assessn<br>ng (48 ho<br>tment sta<br>r study p | to 12 hour<br>nents for th<br>ours after t<br>urts, which<br>rocedure b | s, 12 to 24<br>e third cycle of<br>ne end of<br>ever comes<br>y the |

| Procedures and assessments                          |       |    | EOAC or ED [g] |    |    |    |     |     |       |     |
|---|-------|----|----------------|----|----|----|-----|-----|-------|-----|
|   | D1    | D2 | D3             | D4 | D5 | D7 | D15 | D22 | D29   | D43 |
| Visit window (days)                                 |       |    | +1             |    |    | +1 | ±1  | ±1  | +7    | +7  |
|   |       |    |                |    |    |    |     |     | (+28) |     |
| ECOG performance status                             | Х     |    |                |    |    |    |     |     |       | Х   |
| ceCT/MRI [a]  |       |    |                |    |    |    |     |     | х     | Х   |
| <sup>18</sup> F-FDG-PET                             |       |    |                |    |    |    |     |     |       | Х   |
| <sup>177</sup> Lu-3BP-227 treatment administration  | Х     |    |                |    |    |    |     |     |       |     |
| Planar scintigraphy [b]                             | Х     | Х  | Х              | 2  | X  | Х  |     |     |       |     |
| SPECT/CT scan [b]                                   | Х     | Х  | Х              | 2  | X  | Х  |     |     |       |     |
| Blood sampling for <sup>177</sup> Lu 3BP-227 PK [c] | Х     | Х  | Х              | 2  | X  |    |     |     |       |     |
| AEs   | Х     | Х  | Х              | Х  | Х  | Х  | Х   | х   | Х     | Х   |
| Physical examination                                | х     |    |                |    |    |    |     |     | х     | Х   |
| Vital signs   | x [d] | Х  | Х              |    |    | Х  | Х   | х   | Х     | Х   |
| Body weight   | Х     |    |                |    |    |    | Х   |     | Х     | Х   |
| ECG (12-lead) [e]                                   | Х     |    |                |    |    |    |     |     |       | Х   |
| Haematology and biochemistry                        | Х     | Х  |                |    |    | Х  | Х   | х   | Х     | Х   |
| Urinalysis  | Х     |    |                |    |    |    |     |     | Х     | Х   |
| Prior/concomitant medication/therapy                | х     | Х  | Х              | Х  | Х  | Х  | Х   | Х   | Х     | Х   |
| Pregnancy test [f]                                  | х     |    |                |    |    |    |     |     |       | X   |
| Tumour markers in serum                             | х     |    |                |    |    |    |     |     |       | Х   |

#### Table 9 Schedule of Assessments for Additional Cycles in Case of Clinical Benefit and Good Tolerability

AE=adverse event; ceCT/MRI=contrast enhanced computed tomography/magnetic resonance imaging; D=Day; ECG=electrocardiogram; ECOG=Eastern Cooperative Oncology Group; EOAC=end of additional cycles; IMP=investigational medicinal product; PET=positron emission tomography; PK=pharmacokinetic; SPECT=single photon emission computed tomography.

a ceCT/MRI will be done every 2 cycles during additional administrations, to confirm the clinical benefit after 2 additional administrations (i.e. at Cycle 4 and EOAC). In case of early discontinuation, the ceCT/MRI should be performed at the ED visit, to confirm any disease progression.

b full dosimetry assessment will be performed after the first and third additional administrations as described for Cycles 1 and 2. After the second and fourth additional administrations, only a single SPECT/standard dose CT at 48 (±6) hours will be performed

c After the first and the third additional administration, blood samplings will be performed just before <sup>177</sup>Lu-3BP-227 infusion (baseline), at the end of infusion of <sup>177</sup>Lu-3BP-227 (0), 5 minutes ±1 minute, 30 minutes ±5 minutes, 90 minutes ±15 minutes and 4 hours ±30 minutes, 24 hours ±2 hours and 48 hours ±2 hours, and 72 to 96 hours ±2 hours. After the second and fourth additional administrations a single blood collection will be performed at 24 hours only (as close as possible to the SPECT/CT).

d prior to and at the end of  $^{177}$ Lu-3BP-227 infusion (0) as well as  $30\pm5$  minutes,  $90\pm15$  minutes, 4 hours $\pm30$  minutes after the end of  $^{177}$ Lu-3BP-227 infusion.

- e at each cycle, a triplicate 12-lead ECG will be recorded on Day 1 before the infusion (baseline) (minus 15 minutes). A single 12-lead computerised standard ECG, with paper printout, will be recorded in supine position after at least 5 minutes of rest during each cycle on Day 1 at the end of <sup>177</sup>Lu-3BP-227 infusion (±15 minutes) and at 4 hours (±30 minutes) after the end of <sup>177</sup>Lu-3BP-227 infusion as well as at EOAC.
- f a urine pregnancy test will be performed on Day 1 prior to IMP administration.

g the ED visit will take place within 14 days after ED from the study and at the latest 5 weeks after the last IMP administration.

## **19.2** Attachment 2 – Blood Sampling Summary

This table gives an indication of the number of (veni) punctures and blood volumes for all blood sampling (screening, safety laboratories and bioanalytical assays) during the study. Since this is a multicentre study using local labs, blood volumes for some assessments may vary between labs depending on their internal processes.

Fewer venipunctures and blood draws may actually occur if needed for safety purposes, but this will not require a protocol amendment.

| Purpose                              | Maximum<br>blood volume per<br>sample (mL) | Maximum<br>number of<br>blood samples | Maximum total<br>volume (mL) |
|--------------------------------------|--|---------------------------------------|------------------------------|
| Screening                            |  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 1                                     | 25                           |
| CCI                                  |  |                                       |                              |
| Core trial                           |  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 10                                    | 250                          |
| Radiopharmaceutical PK blood         | 2  | 18                                    | 36                           |
| 3BP-227 PK                           | 2  | 11                                    | 22                           |
| Tumour markers in serum              | 3  | 2                                     | 6                            |
|                                      |  |                                       |                              |
| EOCT/ED                              | -  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 1                                     | 25                           |
| Tumour markers in serum              | 2  | 1                                     | 2                            |
|                                      |  |                                       |                              |
| Additional cycles (number per cycle) | -  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 6                                     | 150                          |
| Radiopharmaceutical PK blood         | 2  | 9                                     | 18                           |
| Tumour markers in serum              | 2  | 1                                     | 2                            |
| EOAC                                 |  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 1                                     | 25                           |
| Tumour markers in serum              | 2  | 1                                     | 2                            |
| LTFU                                 |  |                                       |                              |
| Clinical laboratory tests [a]        | 25   | 8                                     | 200                          |
| Total volume over study period       |  | Approxima                             | tely 820 mL                  |

βHCG=beta human chorionic gonadotrophin; ED=early discontinuation; EOCT=end of core trial; LTFU=long-term follow-up; PK=pharmacokinetic.

a clinical laboratory tests include blood samples for the analysis of haematology, biochemistry, serum βHCG (pregnancy test) and hormone analysis as applicable according to the schedule of assessments.

b

|                     | •         |       |       |            |      |      |
|---------------------|-----------|-------|-------|------------|------|------|
| Haematology         | Screening | Core  | EOCT/ | Additional | EOAC | LTFU |
|                     |           | trial | ED    | cycles     |      |      |
| RBC count           | Х         | Х     | Х     | Х          | Х    |      |
| Haematocrit         | Х         | Х     | Х     | Х          | Х    |      |
| Hb                  | Х         | Х     | Х     | Х          | Х    | Х    |
| MCV                 | Х         | Х     | Х     | Х          | Х    |      |
| WBC count           | Х         | Х     | Х     | Х          | Х    | Х    |
| Absolute counts of: | Х         | Х     | Х     | Х          | Х    | Х    |
| Neutrophils         |           |       |       |            |      |      |
| Lymphocytes         |           |       |       |            |      |      |
| Monocytes           |           |       |       |            |      |      |
| Eosinophils         |           |       |       |            |      |      |
| Basophils           |           |       |       |            |      |      |
| Platelets count     | x         | х     | x     | x          | x    | x    |

# **19.3** Attachment **3** – Clinical Laboratory Tests

ED=early discontinuation; EOAC=end of additional cycles; EOCT=end of core trial; Hb= haemoglobin; LTFU=long-term follow-up; MCV=mean corpuscular volume; RBC=red blood cell; WBC=white blood cell.

| Clinical chemistry   | Screening | Core  | EOCT/ | Additional | EOAC | LTFU |
|--|-----------|-------|-------|------------|------|------|
|  |           | trial | ED    | cycles     |      |      |
| ALT  | Х         | Х     | X     | Х          | Х    | Х    |
| Albumin  | Х         | Х     | Х     | Х          | Х    |      |
| ALP  | Х         | Х     | Х     | Х          | Х    |      |
| AST  | Х         | Х     | Х     | Х          | Х    | Х    |
| Calcium  | Х         | Х     | Х     | Х          | Х    |      |
| Chloride   | Х         | Х     | Х     | Х          | Х    |      |
| Conjugated bilirubin (direct) [a]  | Х         | Х     | Х     | Х          | Х    | Х    |
| Creatinine   | Х         | Х     | Х     | Х          | Х    | Х    |
| CRP  | Х         | Х     | Х     | Х          | Х    |      |
| eGFR   | Х         | Х     | Х     | Х          | Х    |      |
| Glucose  | Х         | Х     | Х     | Х          | Х    |      |
| Potassium  | Х         | Х     | Х     | Х          | Х    |      |
| Sodium   | Х         | Х     | Х     | Х          | Х    |      |
| Total bilirubin  | Х         | Х     | Х     | Х          | Х    | Х    |
| Total cholesterol  | Х         | Х     | Х     | Х          | Х    |      |
| Total protein  | Х         | Х     | Х     | Х          | Х    |      |
| TG   | Х         | Х     | Х     | Х          | Х    |      |
| Urea   | Х         | Х     | Х     | Х          | Х    | Х    |
| Uric acid  | Х         | Х     | Х     | Х          | Х    |      |
| Serum tumour markers:  |           | х     | х     | Х          | Х    |      |
| <ul> <li>CEA - an subjects</li> <li>CA 19-9 – all subjects except<br/>SCCHN</li> </ul> |           |       |       |            |      |      |
| <ul> <li>Serum LDH and BSAP for ES</li> </ul>  |           | 1     |       | 1          | 1    | 1    |

ALP=alkaline phosphatase; ALT=alanine aminotransferase; AST=aspartate aminotransferase; BSAP=Bone Specific Alkaline Phosphatase; CA 19-9=cancer antigen 19-9; CEA=carcinoembryonic antigen; eGFR=estimated glomerular filtration rate; ED=early discontinuation; EOAC=end of additional cycles; EOCT=end of core trial; ES=Ewing Sarcoma; LDH=Lactate dehydrogenase; LTFU=long-term follow-up; SCCHN=squamous-cell carcinoma of head and neck; TG=triglycerides.

a only to be performed if the total bilirubin is abnormal i.e. outside the laboratory normal range

| Urinalysis                         | Screening | Core  | EOCT/ | Additional | EOAC | LTFU |
|------------------------------------|-----------|-------|-------|------------|------|------|
|                                    |           | trial | ED    | cycles     |      |      |
| Urinalysis (dipstick)              |           |       |       |            |      |      |
| Bilirubin                          | Х         | Х     | Х     | Х          | Х    |      |
| Blood                              | Х         | Х     | Х     | Х          | Х    |      |
| Glucose                            | Х         | Х     | Х     | Х          | Х    |      |
| Ketones                            | Х         | Х     | Х     | Х          | Х    |      |
| Leucocytes                         | Х         | х     | Х     | Х          | Х    |      |
| Nitrite                            | Х         | х     | Х     | Х          | Х    |      |
| pH                                 | Х         | х     | Х     | Х          | Х    |      |
| Protein                            | Х         | х     | Х     | Х          | Х    |      |
| Proteinuria                        | Х         | х     | Х     | Х          | Х    |      |
| Specific gravity                   | Х         | х     | Х     | Х          | Х    |      |
| Urobilinogen                       | х         | х     | х     | Х          | Х    |      |
| Urine collection                   |           |       |       |            |      |      |
| Proteinuria (only in case dipstick | Х         | Х     | Х     | Х          | Х    |      |
| is positive, according to          |           |       |       |            |      |      |
| investigator's discretion and      |           |       |       |            |      |      |
| sufficient radioactivity excretion |           |       |       |            |      |      |
| from the subject)                  |           |       |       |            |      |      |

ED=early discontinuation; EOAC=end of additional cycles; EOCT=end of core trial; LTFU=long-term follow-up.

| Hormone analysis | Screening | Core<br>trial | EOCT/<br>ED | Additional<br>cycles | EOAC | LTFU |
|------------------|-----------|---------------|-------------|----------------------|------|------|
| Cortisol         | х         |               | х           |                      |      |      |
| fT4 and TSH      | Х         |               | Х           |                      |      |      |
| IGF-1            | Х         |               | Х           |                      |      |      |
| РТН              | x         |               | x           |                      |      |      |

ED=early discontinuation; EOAC=end of additional cycles; EOCT=end of core trial; fT4=free thyroxine; IGF1=insulin-like growth factor 1; LTFU=long-term follow-up; PTH=parathyroid hormone; TSH=thyroid-stimulating hormone.

| Pregnancy test | Screening   | Core<br>trial | EOCT/<br>ED | Additional cycles | EOAC  | LTFU |
|----------------|-------------|---------------|-------------|-------------------|-------|------|
|                | urine/serum | urine         | urine       | urine             | urine |      |

ED=early discontinuation; EOAC=end of additional cycles; EOCT=end of core trial; LTFU=long-term follow-up.

#### 19.4 Attachment 4 – RECIST Version 1.1 Guidelines

EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247



# New response evaluation criteria in solid tumours: **Revised RECIST guideline (version 1.1)**

E.A. Eisenhauer<sup>a,\*</sup>, P. Therasse<sup>b</sup>, J. Bogaerts<sup>c</sup>, L.H. Schwartz<sup>d</sup>, D. Sargent<sup>e</sup>, R. Ford<sup>f</sup>, J. Dancey<sup>9</sup>, S. Arbuck<sup>h</sup>, S. Gwyther<sup>i</sup>, M. Mooney<sup>9</sup>, L. Rubinstein<sup>9</sup>, L. Shankar<sup>9</sup>, L. Dodd<sup>9</sup>, R. Kaplan<sup>j</sup>, D. Lacombe<sup>c</sup>, J. Verweij<sup>k</sup>

\*National Cancer Institute of Canada – Clinical Trials Group, 10 Stuart Street, Queen's University, Kingston, ON, Canada <sup>b</sup>GlaxoSmithKline Biologicals, Rixensart, Belgium <sup>c</sup>European Organisation for Research and Treatment of Cancer, Data Centre, Brussels, Belgium <sup>d</sup>Memorial Sloan Kettering Cancer Center, New York, NY, USA <sup>e</sup>Mayo Clinic, Rochester, MN, USA <sup>f</sup>RadPharm, Princeton, NJ, USA <sup>g</sup>Division of Cancer Treatment and Diagnosis, National Cancer Institute, Bethesda, MD, USA <sup>h</sup>Schering-Plough, Kenilworth, NJ, USA <sup>1</sup>East Surrey Hospital, Redhill, Surrey, UK <sup>1</sup>National Cancer Research Network, Leeds, UK <sup>k</sup>Erasmus University Medical Center, Rotterdam, The Netherlands

### ARTICLEINFO

#### Article history: Received 17 October 2008

Accepted 29 October 2008

Keywords: Response criteria Solid tumours Guidelines

#### ABSTRACT

Background: Assessment of the change in tumour burden is an important feature of the clinical evaluation of cancer therapeutics: both tumour shrinkage (objective response) and disease progression are useful endpoints in clinical trials. Since RECIST was published in 2000, many investigators, cooperative groups, industry and government authorities have adopted these criteria in the assessment of treatment outcomes. However, a number of questions and issues have arisen which have led to the development of a revised RECIST guideline (version 1.1). Evidence for changes, summarised in separate papers in this special issue, has come from assessment of a large data warehouse (>6500 patients), simulation studies and literature reviews.

Highlights of revised RECIST 1.1: Major changes include: Number of lesions to be assessed: based on evidence from numerous trial databases merged into a data warehouse for analysis purposes, the number of lesions required to assess tumour burden for response determination has been reduced from a maximum of 10 to a maximum of five total (and from five to two per organ, maximum). Assessment of pathological lymph nodes is now incorporated: nodes with a short axis of  $\ge$  15 mm are considered measurable and assessable as target lesions. The short axis measurement should be included in the sum of lesions in calculation of tumour response. Nodes that shrink to <10 mm short axis are considered normal. Confirmation of response is required for trials with response primary endpoint but is no longer required in randomised studies since the control arm serves as appropriate means of interpretation of data. Disease progression is clarified in several aspects: in addition to the previous definition of progression in target disease of 20% increase in sum, a 5 mm absolute increase is now required as well to guard against over calling PD when the total sum is very

\* Corresponding author: Tel.: +1 613 533 6430; fax: +1 613 533 2411.

E-mail address: eeisenhauer@ctg.queensu.ca (E.A. Eisenhauer). 0959-8049/\$ - see front matter © 2008 Elsevier Ltd. All rights reserved. doi:10.1016/j.ejca.2008.10.026

#### EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247

229

small. Furthermore, there is guidance offered on what constitutes 'unequivocal progression' of non-measurable/non-target disease, a source of confusion in the original RECIST guideline. Finally, a section on detection of new lesions, including the interpretation of FDG-PET scan assessment is included. *Imaging guidance*: the revised RECIST includes a new imaging appendix with updated recommendations on the optimal anatomical assessment of lesions.

Future work: A key question considered by the RECIST Working Group in developing RECIST 1.1 was whether it was appropriate to move from anatomic unidimensional assessment of tumour burden to either volumetric anatomical assessment or to functional assessment with PET or MRI. It was concluded that, at present, there is not sufficient standardisation or evidence to abandon anatomical assessment of tumour burden. The only exception to this is in the use of FDG-PET imaging as an adjunct to determination of progression. As is detailed in the final paper in this special issue, the use of these promising newer approaches requires appropriate clinical validation studies.

© 2008 Elsevier Ltd. All rights reserved.

#### 1. Background

#### 1.1. History of RECIST criteria

Assessment of the change in tumour burden is an important feature of the clinical evaluation of cancer therapeutics. Both tumour shrinkage (objective response) and time to the development of disease progression are important endpoints in cancer clinical trials. The use of tumour regression as the endpoint for phase II trials screening new agents for evidence of anti-tumour effect is supported by years of evidence suggesting that, for many solid tumours, agents which produce tumour shrinkage in a proportion of patients have a reasonable (albeit imperfect) chance of subsequently demonstrating an improvement in overall survival or other time to event measures in randomised phase III studies (reviewed in [1-4]). At the current time objective response carries with it a body of evidence greater than for any other biomarker supporting its utility as a measure of promising treatment effect in phase II screening trials. Furthermore, at both the phase II and phase III stage of drug development, clinical trials in advanced disease settings are increasingly utilising time to progression (or progression-free survival) as an endpoint upon which efficacy conclusions are drawn, which is also based on anatomical measurement of tumour size.

However, both of these tumour endpoints, objective response and time to disease progression, are useful only if based on widely accepted and readily applied standard criteria based on anatomical tumour burden. In 1981 the World Health Organisation (WHO) first published tumour response criteria, mainly for use in trials where tumour response was the primary endpoint. The WHO criteria introduced the concept of an overall assessment of tumour burden by summing the products of bidimensional lesion measurements and determined response to therapy by evaluation of change from baseline while on treatment.<sup>5</sup> However, in the decades that followed their publication, cooperative groups and pharmaceutical companies that used the WHO criteria often 'modified' them to accommodate new technologies or to address areas that were unclear in the original document. This led

to confusion in interpretation of trial results6 and in fact, the application of varying response criteria was shown to lead to very different conclusions about the efficacy of the same regimen.7 In response to these problems, an International Working Party was formed in the mid 1990s to standardise and simplify response criteria. New criteria, known as RECIST (Response Evaluation Criteria in Solid Tumours), were published in 2000.8 Key features of the original RECIST include definitions of minimum size of measurable lesions, instructions on how many lesions to follow (up to 10; a maximum five per organ site), and the use of unidimensional, rather than bidimensional, measures for overall evaluation of tumour burden. These criteria have subsequently been widely adopted by academic institutions, cooperative groups, and industry for trials where the primary endpoints are objective response or progression. In addition, regulatory authorities accept RECIST as an appropriate guideline for these assessments.

#### 1.2. Why update RECIST?

Since RECIST was published in 2000, many investigators have confirmed in prospective analyses the validity of substituting unidimensional for bidimensional (and even three-dimensional)-based criteria (reviewed in [9]). With rare exceptions (e.g. mesothelioma), the use of unidimensional criteria seems to perform well in solid tumour phase II studies.

However, a number of questions and issues have arisen which merit answers and further clarity. Amongst these are whether fewer than 10 lesions can be assessed without affecting the overall assigned response for patients (or the conclusion about activity in trials); how to apply RECIST in randomised phase III trials where progression, not response, is the primary endpoint particularly if not all patients have measurable disease; whether or how to utilise newer imaging technologies such as FDG-PET and MRI; how to handle assessment of lymph nodes; whether response confirmation is truly needed; and, not least, the applicability of RECIST in trials of targeted non-cytotoxic drugs. This revision of the RECIST guidelines includes updates that touch on all these points.
### 1.3. Process of RECIST 1.1 development

The RECIST Working Group, consisting of clinicians with expertise in early drug development from academic research organisations, government and industry, together with imaging specialists and statisticians, has met regularly to set the agenda for an update to RECIST, determine the evidence needed to justify the various changes made, and to review emerging evidence. A critical aspect of the revision process was to create a database of prospectively documented solid tumour measurement data obtained from industry and academic group trials. This database, assembled at the EORTC Data Centre under the leadership of Jan Bogaerts and Patrick Therasse (co-authors of this guideline), consists of >6500 patients with >18,000 target lesions and was utilised to investigate the impact of a variety of questions (e.g. number of target lesions required, the need for response confirmation, and lymph node measurement rules) on response and progression-free survival outcomes. The results of this work, which after evaluation by the RECIST Working Group led to most of the changes in this revised guideline, are reported in detail in a separate paper in this special issue.<sup>10</sup> Larry Schwartz and Robert Ford (also co-authors of this guideline) also provided key databases from which inferences have been made that inform these revisions.<sup>21</sup>

The publication of this revised guideline is believed to be timely since it incorporates changes to simplify, optimise and standardise the assessment of tumour burden in clinical trials. A summary of key changes is found in Appendix I. Because the fundamental approach to assessment remains grounded in the anatomical, rather than functional, assessment of disease, we have elected to name this version RECIST 1.1, rather than 2.0.

### 1.4. What about volumetric or functional assessment?

This raises the question, frequently posed, about whether it is 'time' to move from anatomic unidimensional assessment of tumour burden to either volumetric anatomical assessment or to functional assessment (e.g. dynamic contrast enhanced MRI or CT or (18)F-fluorodeoxyglucose positron emission tomographic (FDG-PET) techniques assessing tumour metabolism). As can be seen, the Working Group and particularly those involved in imaging research, did not believe that there is at present sufficient standardisation and widespread availability to recommend adoption of these alternative assessment methods. The only exception to this is in the use of FDG-PET imaging as an adjunct to determination of progression, as described later in this guideline. As detailed in paper in this special issue12, we believe that the use of these promising newer approaches (which could either add to or substitute for anatomical assessment as described in RECIST) requires appropriate and rigorous clinical validation studies. This paper by Sargent et al. illustrates the type of data that will be needed to be able to define 'endpoints' for these modalities and how to determine where and when such criteria/modalities can be used to improve the reliability with which truly active new agents are identified and truly inactive new agents are discarded in comparison to RECIST criteria in phase II screening trials. The RECIST Working Group looks forward

to such data emerging in the next few years to allow the appropriate changes to the next iteration of the RECIST criteria.

### 2. Purpose of this guideline

This guideline describes a standard approach to solid tumour measurement and definitions for objective assessment of change in tumour size for use in adult and paediatric cancer clinical trials. It is expected these criteria will be useful in all trials where objective response is the primary study endpoint, as well as in trials where assessment of stable disease, tumour progression or time to progression analyses are undertaken, since all of these outcome measures are based on an assessment of anatomical tumour burden and its change on study. There are no assumptions in this paper about the proportion of patients meeting the criteria for any of these endpoints which will signal that an agent or treatment regimen is active: those definitions are dependent on type of cancer in which a trial is being undertaken and the specific agent(s) under study. Protocols must include appropriate statistical sections which define the efficacy parameters upon which the trial sample size and decision criteria are based. In addition to providing definitions and criteria for assessment of tumour response, this guideline also makes recommendations regarding standard reporting of the results of trials that utilise tumour response as an endpoint.

While these guidelines may be applied in malignant brain tumour studies, there are also separate criteria published for response assessment in that setting.<sup>13</sup> This guideline is not intended for use for studies of malignant lymphoma since international guidelines for response assessment in lymphoma are published separately.<sup>24</sup>

Finally, many oncologists in their daily clinical practice follow their patients' malignant disease by means of repeated imaging studies and make decisions about continued therapy on the basis of both objective and symptomatic criteria. It is not intended that these RECIST guidelines play a role in that decision making, except if determined appropriate by the treating oncologist.

### 3. Measurability of tumour at baseline

### 3.1. Definitions

At baseline, tumour lesions/lymph nodes will be categorised measurable or non-measurable as follows:

### 3.1.1. Measurable

Tumour lesions: Must be accurately measured in at least one dimension (longest diameter in the plane of measurement is to be recorded) with a minimum size of:

- 10 mm by CT scan (CT scan slice thickness no greater than 5 mm; see Appendix II on imaging guidance).
- 10 mm caliper measurement by clinical exam (lesions which cannot be accurately measured with calipers should be recorded as non-measurable).
- 20 mm by chest X-ray.

Malignant lymph nodes: To be considered pathologically enlarged and measurable, a lymph node must be  $\geq$ 15 mm in short axis when assessed by CT scan (CT scan slice thickness recommended to be no greater than 5 mm). At baseline and in follow-up, only the short axis will be measured and followed (see Schwartz et al. in this Special Issue<sup>15</sup>). See also notes below on 'Baseline documentation of target and non-target lesions' for information on lymph node measurement.

### 3.1.2. Non-measurable

All other lesions, including small lesions (longest diameter <10 mm or pathological lymph nodes with  $\ge$  10 to <15 mm short axis) as well as truly non-measurable lesions. Lesions considered truly non-measurable include: leptomeningeal disease, ascites, pleural or pericardial effusion, inflammatory breast disease, lymphangitic involvement of skin or lung, abdominal masses/abdominal organomegaly identified by physical exam that is not measurable by reproducible imaging techniques.

3.1.3. Special considerations regarding lesion measurability Bone lesions, cystic lesions, and lesions previously treated with local therapy require particular comment:

### Bone lesions:.

- Bone scan, PET scan or plain films are not considered adequate imaging techniques to measure bone lesions. However, these techniques can be used to confirm the presence or disappearance of bone lesions.
- Lytic bone lesions or mixed lytic-blastic lesions, with identifiable soft tissue components, that can be evaluated by cross sectional imaging techniques such as CT or MRI can be considered as measurable lesions if the soft tissue component meets the definition of measurability described above.
- Blastic bone lesions are non-measurable

### Cystic lesions:

- Lesions that meet the criteria for radiographically defined simple cysts should not be considered as malignant lesions (neither measurable nor non-measurable) since they are, by definition, simple cysts.
- 'Cystic lesions' thought to represent cystic metastases can be considered as measurable lesions, if they meet the definition of measurability described above. However, if noncystic lesions are present in the same patient, these are preferred for selection as target lesions.

Lesions with prior local treatment:

 Tumour lesions situated in a previously irradiated area, or in an area subjected to other loco-regional therapy, are usually not considered measurable unless there has been demonstrated progression in the lesion. Study protocols should detail the conditions under which such lesions would be considered measurable.

### 3.2. Specifications by methods of measurements

### 3.2.1. Measurement of lesions

All measurements should be recorded in metric notation, using calipers if clinically assessed. All baseline evaluations

should be performed as close as possible to the treatment start and never more than 4 weeks before the beginning of the treatment.

### 3.2.2. Method of assessment

The same method of assessment and the same technique should be used to characterise each identified and reported lesion at baseline and during follow-up. Imaging based evaluation should always be done rather than clinical examination unless the lesion(s) being followed cannot be imaged but are assessable by clinical exam.

Clinical lesions: Clinical lesions will only be considered measurable when they are superficial and  $\geq 10$  mm diameter as assessed using calipers (e.g. skin nodules). For the case of skin lesions, documentation by colour photography including a ruler to estimate the size of the lesion is suggested. As noted above, when lesions can be evaluated by both clinical exam and imaging, imaging evaluation should be undertaken since it is more objective and may also be reviewed at the end of the study.

Chest X-ray: Chest CT is preferred over chest X-ray, particularly when progression is an important endpoint, since CT is more sensitive than X-ray, particularly in identifying new lesions. However, lesions on chest X-ray may be considered measurable if they are clearly defined and surrounded by aerated lung. See Appendix II for more details.

CT, MRI: CT is the best currently available and reproducible method to measure lesions selected for response assessment. This guideline has defined measurability of lesions on CT scan based on the assumption that CT slice thickness is 5 mm or less. As is described in Appendix II, when CT scans have slice thickness greater than 5 mm, the minimum size for a measurable lesion should be twice the slice thickness. MRI is also acceptable in certain situations (e.g. for body scans). More details concerning the use of both CT and MRI for assessment of objective tumour response evaluation are provided in Appendix II.

Ultrasound: Ultrasound is not useful in assessment of lesion size and should not be used as a method of measurement. Ultrasound examinations cannot be reproduced in their entirety for independent review at a later date and, because they are operator dependent, it cannot be guaranteed that the same technique and measurements will be taken from one assessment to the next (described in greater detail in Appendix II). If new lesions are identified by ultrasound in the course of the study, confirmation by CT or MRI is advised. If there is concern about radiation exposure at CT, MRI may be used instead of CT in selected instances.

Endoscopy, laparoscopy: The utilisation of these techniques for objective tumour evaluation is not advised. However, they can be useful to confirm complete pathological response when biopsies are obtained or to determine relapse in trials where recurrence following complete response or surgical resection is an endpoint.

Tumour markers: Tumour markers alone cannot be used to assess objective tumour response. If markers are initially above

the upper normal limit, however, they must normalise for a patient to be considered in complete response. Because tumour markers are disease specific, instructions for their measurement should be incorporated into protocols on a disease specific basis. Specific guidelines for both CA-125 response (in recurrent ovarian cancer) and PSA response (in recurrent prostate cancer), have been published.<sup>16-28</sup> In addition, the Gynecologic Cancer Intergroup has developed CA125 progression criteria which are to be integrated with objective tumour assessment for use in first-line trials in ovarian cancer.<sup>19</sup>

232

Cytology, histology: These techniques can be used to differentiate between PR and CR in rare cases if required by protocol (for example, residual lesions in tumour types such as germ cell tumours, where known residual benign tumours can remain). When effusions are known to be a potential adverse effect of treatment (e.g. with certain taxane compounds or angiogenesis inhibitors), the cytological confirmation of the neoplastic origin of any effusion that appears or worsens during treatment can be considered if the measurable tumour has met criteria for response or stable disease in order to differentiate between response (or stable disease) and progressive disease.

### 4. Tumour response evaluation

## 4.1. Assessment of overall tumour burden and measurable disease

To assess objective response or future progression, it is necessary to estimate the overall tumour burden at baseline and use this as a comparator for subsequent measurements. Only patients with measurable disease at baseline should be included in protocols where objective tumour response is the primary endpoint. Measurable disease is defined by the presence of at least one measurable lesion (as detailed above in Section 3). In studies where the primary endpoint is tumour progression (either time to progression or proportion with progression at a fixed date), the protocol must specify if entry is restricted to those with measurable disease or whether patients having non-measurable disease only are also eligible.

## 4.2. Baseline documentation of 'target' and 'non-target' lesions

When more than one measurable lesion is present at baseline all lesions up to a maximum of five lesions total (and a maximum of two lesions per organ) representative of all involved organs should be identified as *target lesions* and will be recorded and measured at baseline (this means in instances where patients have only one or two organ sites involved a maximum of two and four lesions respectively will be recorded). For evidence to support the selection of only five target lesions, see analyses on a large prospective database in the article by Bogaerts et al.<sup>10</sup>.

Target lesions should be selected on the basis of their size (lesions with the longest diameter), be representative of all involved organs, but in addition should be those that lend themselves to reproducible repeated measurements. It may be the case that, on occasion, the largest lesion does not lend itself to reproducible measurement in which circumstance the next largest lesion which can be measured reproducibly should be selected. To illustrate this point see the example in Fig. 3 of Appendix II.

Lymph nodes merit special mention since they are normal anatomical structures which may be visible by imaging even if not involved by tumour. As noted in Section 3, pathological nodes which are defined as measurable and may be identified as target lesions must meet the criterion of a short axis of ≥15 mm by CT scan. Only the short axis of these nodes will contribute to the baseline sum. The short axis of the node is the diameter normally used by radiologists to judge if a node is involved by solid tumour. Nodal size is normally reported as two dimensions in the plane in which the image is obtained (for CT scan this is almost always the axial plane; for MRI the plane of acquisition may be axial, saggital or coronal). The smaller of these measures is the short axis. For example, an abdominal node which is reported as being 20 mm × 30 mm has a short axis of 20 mm and qualifies as a malignant, measurable node. In this example, 20 mm should be recorded as the node measurement (See also the example in Fig. 4 in Appendix II). All other pathological nodes (those with short axis  $\ge 10$  mm but <15 mm) should be considered non-target lesions. Nodes that have a short axis <10 mm are considered non-pathological and should not be recorded or followed

A sum of the diameters (longest for non-nodal lesions, short axis for nodal lesions) for all target lesions will be calculated and reported as the baseline sum diameters. If lymph nodes are to be included in the sum, then as noted above, only the short axis is added into the sum. The baseline sum diameters will be used as reference to further characterise any objective tumour regression in the measurable dimension of the disease.

All other lesions (or sites of disease) including pathological lymph nodes should be identified as non-target lesions and should also be recorded at baseline. Measurements are not required and these lesions should be followed as 'present', 'absent', or in rare cases 'unequivocal progression' (more details to follow). In addition, it is possible to record multiple nontarget lesions involving the same organ as a single item on the case record form (e.g. 'multiple enlarged pelvic lymph nodes' or 'multiple liver metastases').

### 4.3. Response criteria

This section provides the definitions of the criteria used to determine objective tumour response for target lesions.

4.3.1. Evaluation of target lesions

- Complete Response (CR): Disappearance of all target lesions. Any pathological lymph nodes (whether target or non-target) must have reduction in short axis to <10 mm.
- Partial Response (PR): At least a 30% decrease in the sum of diameters of target lesions, taking as reference the baseline sum diameters.

- Progressive Disease (PD): At least a 20% increase in the sum of diameters of target lesions, taking as reference the smallest sum on study (this includes the baseline sum if that is the smallest on study). In addition to the relative increase of 20%, the sum must also demonstrate an absolute increase of at least 5 mm. (Note: the appearance of one or more new lesions is also considered progression).
- Stable Disease (SD): Neither sufficient shrinkage to qualify for PR nor sufficient increase to qualify for PD, taking as reference the smallest sum diameters while on study.

### 4.3.2. Special notes on the assessment of target lesions

Lymph nodes. Lymph nodes identified as target lesions should always have the actual short axis measurement recorded (measured in the same anatomical plane as the baseline examination), even if the nodes regress to below 10 mm on study. This means that when lymph nodes are included as target lesions, the 'sum' of lesions may not be zero even if complete response criteria are met, since a normal lymph node is defined as having a short axis of <10 mm. Case report forms or other data collection methods may therefore be designed to have target nodal lesions recorded in a separate section where, in order to qualify for CR, each node must achieve a short axis <10 mm. For PR, SD and PD, the actual short axis measurement of the nodes is to be included in the sum of target lesions.

Target lesions that become 'too small to measure'. While on study, all lesions (nodal and non-nodal) recorded at baseline should have their actual measurements recorded at each subsequent evaluation, even when very small (e.g. 2 mm). However, sometimes lesions or lymph nodes which are recorded as target lesions at baseline become so faint on CT scan that the radiologist may not feel comfortable assigning an exact measure and may report them as being 'too small to measure'. When this occurs it is important that a value be recorded on the case report form. If it is the opinion of the radiologist that the lesion has likely disappeared, the measurement should be recorded as 0 mm. If the lesion is believed to be present and is faintly seen but too small to measure, a default value of 5 mm should be assigned (Note: It is less likely that this rule will be used for lymph nodes since they usually have a definable size when normal and are frequently surrounded by fat such as in the retroperitoneum; however, if a lymph node is believed to be present and is faintly seen but too small to measure, a default value of 5 mm should be assigned in this circumstance as well). This default value is derived from the 5 mm GT slice thickness (but should not be changed with varying CT slice thickness). The measurement of these lesions is potentially non-reproducible, therefore providing this default value will prevent false responses or progressions based upon measurement error. To reiterate, however, if the radiologist is able to provide an actual measure, that should be recorded, even if it is below 5 mm.

Lesions that split or coalesce on treatment. As noted in Appendix II, when non-nodal lesions 'fragment', the longest diameters of the fragmented portions should be added together to calculate the target lesion sum. Similarly, as lesions coalesce, a plane between them may be maintained that would aid in obtaining maximal diameter measurements of each individual lesion. If the lesions have truly coalesced such that they are no longer separable, the vector of the longest diameter in this instance should be the maximal longest diameter for the 'coalesced lesion'.

4.3.3. Evaluation of non-target lesions

This section provides the definitions of the criteria used to determine the tumour response for the group of non-target lesions. While some non-target lesions may actually be measurable, they need not be measured and instead should be assessed only qualitatively at the time points specified in the protocol.

- Complete Response (CR): Disappearance of all non-target lesions and normalisation of tumour marker level. All lymph nodes must be non-pathological in size (<10 mm short axis).
- Non-CR/Non-PD: Persistence of one or more non-target lesion(s) and/or maintenance of tumour marker level above the normal limits.
- Progressive Disease (PD): Unequivocal progression (see comments below) of existing non-target lesions. (Note: the appearance of one or more new lesions is also considered progression).

### 4.3.4. Special notes on assessment of progression of nontarget disease

The concept of progression of non-target disease requires additional explanation as follows:

When the patient also has measurable disease. In this setting, to achieve 'unequivocal progression' on the basis of the non-target disease, there must be an overall level of substantial worsening in non-target disease such that, even in presence of SD or PR in target disease, the overall tumour burden has increased sufficiently to merit discontinuation of therapy (see examples in Appendix II and further details below). A modest 'increase' in the size of one or more non-target lesions is usually not sufficient to quality for unequivocal progression status. The designation of overall progression solely on the basis of change in non-target disease in the face of SD or PR of target disease will therefore be extremely rare.

When the patient has only non-measurable disease. This circumstance arises in some phase III trials when it is not a criterion of study entry to have measurable disease. The same general concepts apply here as noted above, however, in this instance there is no measurable disease assessment to factor into the interpretation of an increase in non-measurable disease burden. Because worsening in non-target disease cannot be easily quantified (by definition: if all lesions are truly non-measurable) a useful test that can be applied when assessing patients for unequivocal progression is to consider if the increase in overall disease burden based on the change in non-measurable disease is comparable in magnitude to the increase that would be required to declare PD for measurable disease: i.e. an increase in tumour burden representing an additional 73% increase in 'volume' (which is equivalent to a 20% increase diameter in a measurable lesion). Examples include an increase in a pleural effusion from 'trace' to 'large', an increase in lymphangitic

disease from localised to widespread, or may be described in protocols as 'sufficient to require a change in therapy'. Some illustrative examples are shown in Figs. 5 and 6 in Appendix II. If 'unequivocal progression' is seen, the patient should be considered to have had overall PD at that point. While it would be ideal to have objective criteria to apply to non-measurable disease, the very nature of that disease makes it impossible to do so, therefore the increase must be substantial.

### 4.3.5. New lesions

The appearance of new malignant lesions denotes disease progression; therefore, some comments on detection of new lesions are important. There are no specific criteria for the identification of new radiographic lesions; however, the finding of a new lesion should be unequivocal: i.e. not attributable to differences in scanning technique, change in imaging modality or findings thought to represent something other than tumour (for example, some 'new' bone lesions may be simply healing or flare of pre-existing lesions). This is particularly important when the patient's baseline lesions show partial or complete response. For example, necrosis of a liver lesion may be reported on a CT scan report as a 'new' cystic lesion, which it is not.

A lesion identified on a follow-up study in an anatomical location that was not scanned at baseline is considered a new lesion and will indicate disease progression. An example of this is the patient who has visceral disease at baseline and while on study has a CT or MRI brain ordered which reveals metastases. The patient's brain metastases are considered to be evidence of PD even if he/she did not have brain imaging at baseline.

If a new lesion is equivocal, for example because of its small size, continued therapy and follow-up evaluation will clarify if it represents truly new disease. If repeat scans confirm there is definitely a new lesion, then progression should be declared using the date of the initial scan.

While FDG-PET response assessments need additional study, it is sometimes reasonable to incorporate the use of FDG-PET scanning to complement CT scanning in assessment of progression (particularly possible 'new' disease). New lesions on the basis of FDG-PET imaging can be identified according to the following algorithm:

- Negative FDG-PET at baseline, with a positive<sup>1</sup> FDG-PET at follow-up is a sign of PD based on a new lesion.
- b. No FDG-PET at baseline and a positive FDG-PET at follow-up:

If the positive FDG-PET at follow-up corresponds to a new site of disease confirmed by CT, this is PD.

If the positive FDG-PET at follow-up is not confirmed as a new site of disease on CT, additional follow-up CT scans are needed to determine if there is truly progression occurring at that site (if so, the date of PD will be the date of the initial abnormal FDG-PET scan).

If the positive FDG-PET at follow-up corresponds to a pre-existing site of disease on CT that is not progressing on the basis of the anatomic images, this is not PD.

### 4.4. Evaluation of best overall response

The best overall response is the best response recorded from the start of the study treatment until the end of treatment taking into account any requirement for confirmation. On occasion a response may not be documented until after the end of therapy so protocols should be clear if post-treatment assessments are to be considered in determination of best overall response. Protocols must specify how any new therapy introduced before progression will affect best response designation. The patient's best overall response assignment will depend on the findings of both target and non-target disease and will also take into consideration the appearance of new lesions. Furthermore, depending on the nature of the study and the protocol requirements, it may also require confirmatory measurement (see Section 4.6). Specifically, in non-randomised trials where response is the primary endpoint, confirmation of PR or CR is needed to deem either one the 'best overall response'. This is described further below.

### 4.4.1. Time point response

It is assumed that at each protocol specified time point, a response assessment occurs. Table 1 on the next page provides a summary of the overall response status calculation at each time point for patients who have measurable disease at baseline.

When patients have non-measurable (therefore non-target) disease only, Table 2 is to be used.

### 4.4.2. Missing assessments and inevaluable designation

When no imaging/measurement is done at all at a particular time point, the patient is not evaluable (NE) at that time point. If only a subset of lesion measurements are made at an assessment, usually the case is also considered NE at that time point, unless a convincing argument can be made that the contribution of the individual missing lesion(s) would not change the assigned time point response. This would be most likely to happen in the case of PD. For example, if a patient had a baseline sum of 50 mm with three measured lesions and at follow-up only two lesions were assessed, but those gave a sum of 80 mm, the patient will have achieved PD status, regardless of the contribution of the missing lesion.

### 4.4.3. Best overall response: all time points

The best overall response is determined once all the data for the patient is known.

Best response determination in trials where confirmation of complete or partial response IS NOT required: Best response in these trials is defined as the best response across all time points (for example, a patient who has SD at first assessment, PR at second assessment, and PD on last assessment has a best overall response of PR). When SD is believed to be best response, it must also meet the protocol specified minimum time from baseline. If the minimum time is not met when SD is otherwise the best time point response, the patient's best response depends on the subsequent assessments. For example, a patient who has SD at first assessment, PD at second and does not meet minimum duration for SD, will have a best response of PD. The same patient lost to follow-up after the first SD assessment would be considered inevaluable.

<sup>&</sup>lt;sup>1</sup> A 'positive' FDG-PET scan lesion means one which is FDG avid with an uptake greater than twice that of the surrounding tissue on the attenuation corrected image.

| Target lesions       | Non-target lesions             | New<br>lesions | Overall<br>response |  |
|----------------------|--------------------------------|----------------|---------------------|--|
| CR                   | CR                             | No             | CR                  |  |
| CR                   | Non-CR/non-PD                  | No             | PR                  |  |
| CR                   | Not evaluated                  | No             | PR                  |  |
| PR                   | Non-PD or<br>not all evaluated | No             | PR                  |  |
| SD                   | Non-PD or<br>not all evaluated | No             | SD                  |  |
| Not all<br>evaluated | Non-PD                         | No             | NE                  |  |
| PD                   | Any                            | Yes or No      | PD                  |  |
| Any                  | PD                             | Yes or No      | PD                  |  |
| Any                  | Any                            | Yes            | PD                  |  |

PD = progressive disease, and NE = inevaluable.

| Table 2 - Time point response: patients with non-target disease only. |                    |                            |  |
|---|--------------------|----------------------------|--|
| Non-target lesions  | New lesions        | Overall response           |  |
| CR  | No                 | CR                         |  |
| Non-CR/non-PD   | No                 | Non-CR/non-PD <sup>a</sup> |  |
| Not all evaluated   | No                 | NE                         |  |
| Unequivocal PD  | Yes or No          | PD                         |  |
| Any   | Yes                | PD                         |  |
| CR = complete respon<br>NE = inevaluable.                             | nse, PD = progress | sive disease, and          |  |

a 'Non-CR/non-PD' is preferred over 'stable disease' for non-target disease since SD is increasingly used as endpoint for assessment of efficacy in some trials so to assign this category when no lesions can be measured is not advised.

Best response determination in trials where confirmation of complete or partial response IS required: Complete or partial responses may be claimed only if the criteria for each are met at a subsequent time point as specified in the protocol (generally 4 weeks later). In this circumstance, the best overall response can be interpreted as in Table 3.

### 4.4.4. Special notes on response assessment

When nodal disease is included in the sum of target lesions and the nodes decrease to 'normal' size (<10 mm), they may still have a measurement reported on scans. This measurement should be recorded even though the nodes are normal in order not to overstate progression should it be based on increase in size of the nodes. As noted earlier, this means that patients with CR may not have a total sum of 'zero' on the case report form (CRF).

In trials where confirmation of response is required, repeated 'NE' time point assessments may complicate best response determination. The analysis plan for the trial must address how missing data/assessments will be addressed in determination of response and progression. For example, in most trials it is reasonable to consider a patient with time point responses of PR-NE-PR as a confirmed response.

Patients with a global deterioration of health status requiring discontinuation of treatment without objective evidence of disease progression at that time should be reported as 'symptomatic deterioration'. Every effort should be made to document objective progression even after discontinuation of treatment. Symptomatic deterioration is not a descriptor of an objective response: it is a reason for stopping study therapy. The objective response status of such patients is to be determined by evaluation of target and non-target disease as shown in Tables 1–3.

Conditions that define 'early progression, early death and inevaluability' are study specific and should be clearly described in each protocol (depending on treatment duration, treatment periodicity).

In some circumstances it may be difficult to distinguish residual disease from normal tissue. When the evaluation of complete response depends upon this determination, it is recommended that the residual lesion be investigated (fine

| Table 3 – Best overall response when confirmation of CR and PR required. |   |   |  |  |
|--|---|---|--|--|
| Overall response<br>First time point                                     | Overall response<br>Subsequent time point | BEST overall response   |  |  |
| CR   | CR  | CR  |  |  |
| CR   | PR  | SD, PD or PR <sup>a</sup>                                       |  |  |
| CR   | SD  | SD provided minimum criteria for SD duration met, otherwise, PD |  |  |
| CR   | PD  | SD provided minimum criteria for SD duration met, otherwise, PD |  |  |
| CR   | NE  | SD provided minimum criteria for SD duration met, otherwise NE  |  |  |
| PR   | CR  | PR  |  |  |
| PR   | PR  | PR  |  |  |
| PR   | SD  | SD  |  |  |
| PR   | PD  | SD provided minimum criteria for SD duration met, otherwise, PD |  |  |
| PR   | NE  | SD provided minimum criteria for SD duration met, otherwise NE  |  |  |
| NE   | NE  | NE  |  |  |

CR = complete response, PR = partial response, SD = stable disease, PD = progressive disease, and NE = inevaluable. a If a CR is truly met at first time point, then any disease seen at a subsequent time point, even disease meeting PR criteria relative to baseline, makes the disease PD at that point (since disease must have reappeared after CR). Best response would depend on whether minimum duration for SD was met. However, sometimes 'CR' may be claimed when subsequent scans suggest small lesions were likely still present and in fact the patient had PR, not CR at the first time point. Under these circumstances, the original CR should be changed to PR and the best response is PR.

needle aspirate/biopsy) before assigning a status of complete response. FDG-PET may be used to upgrade a response to a CR in a manner similar to a biopsy in cases where a residual radiographic abnormality is thought to represent fibrosis or scarring. The use of FDG-PET in this circumstance should be prospectively described in the protocol and supported by disease specific medical literature for the indication. However, it must be acknowledged that both approaches may lead to false positive CR due to limitations of FDG-PET and biopsy resolution/sensitivity.

For equivocal findings of progression (e.g. very small and uncertain new lesions; cystic changes or necrosis in existing lesions), treatment may continue until the next scheduled assessment. If at the next scheduled assessment, progression is confirmed, the date of progression should be the earlier date when progression was suspected.

### 4.5. Frequency of tumour re-evaluation

Frequency of tumour re-evaluation while on treatment should be protocol specific and adapted to the type and schedule of treatment. However, in the context of phase II studies where the beneficial effect of therapy is not known, follow-up every 6-8 weeks (timed to coincide with the end of a cycle) is reasonable. Smaller or greater time intervals than these could be justified in specific regimens or circumstances. The protocol should specify which organ sites are to be evaluated at baseline (usually those most likely to be involved with metastatic disease for the tumour type under study) and how often evaluations are repeated. Normally, all target and non-target sites are evaluated at each assessment. In selected circumstances certain non-target organs may be evaluated less frequently. For example, bone scans may need to be repeated only when complete response is identified in target disease or when progression in bone is suspected.

After the end of the treatment, the need for repetitive tumour evaluations depends on whether the trial has as a goal the response rate or the time to an event (progression/death). If 'time to an event' (e.g. time to progression, disease-free survival, progression-free survival) is the main endpoint of the study, then routine scheduled re-evaluation of protocol specified sites of disease is warranted. In randomised comparative trials in particular, the scheduled assessments should be performed as identified on a calendar schedule (for example: every 6–8 weeks on treatment or every 3–4 months after treatment) and should not be affected by delays in therapy, drug holidays or any other events that might lead to imbalance in a treatment arm in the timing of disease assessment.

### 4.6. Confirmatory measurement/duration of response

#### 4.6.1. Confirmation

In non-randomised trials where response is the primary endpoint, confirmation of PR and CR is required to ensure responses identified are not the result of measurement error. This will also permit appropriate interpretation of results in the context of historical data where response has traditionally required confirmation in such trials (see the paper by Bogaetts et al. in this Special Issue<sup>10</sup>). However, in all other circumstances, i.e. in randomised trials (phase II or III) or studies where stable disease or progression are the primary endpoints, confirmation of response is not required since it will not add value to the interpretation of trial results. However, elimination of the requirement for response confirmation may increase the importance of central review to protect against bias, in particular in studies which are not blinded.

In the case of SD, measurements must have met the SD criteria at least once after study entry at a minimum interval (in general not less than 6-8 weeks) that is defined in the study protocol.

### 4.6.2. Duration of overall response

The duration of overall response is measured from the time measurement criteria are first met for CR/PR (whichever is first recorded) until the first date that recurrent or progressive disease is objectively documented (taking as reference for progressive disease the smallest measurements recorded on study).

The duration of overall complete response is measured from the time measurement criteria are first met for CR until the first date that recurrent disease is objectively documented.

### 4.6.3. Duration of stable disease

Stable disease is measured from the start of the treatment (in randomised trials, from date of randomisation) until the criteria for progression are met, taking as reference the smallest sum on study (if the baseline sum is the smallest, this is the reference for calculation of PD).

The clinical relevance of the duration of stable disease varies in different studies and diseases. If the proportion of patients achieving stable disease for a minimum period of time is an endpoint of importance in a particular trial, the protocol should specify the minimal time interval required between two measurements for determination of stable disease.

Note: The duration of response and stable disease as well as the progression-free survival are influenced by the frequency of follow-up after baseline evaluation. It is not in the scope of this guideline to define a standard follow-up frequency. The frequency should take into account many parameters including disease types and stages, treatment periodicity and standard practice. However, these limitations of the precision of the measured endpoint should be taken into account if comparisons between trials are to be made.

### 4.7. Progression-free survival/proportion progression-free

#### 4.7.1. Phase II trials

This guideline is focused primarily on the use of objective response endpoints for phase II trials. In some circumstances, 'response rate' may not be the optimal method to assess the potential anticancer activity of new agents/regimens. In such cases 'progression-free survival' (PFS) or the 'proportion progression-free' at landmark time points, might be considered appropriate alternatives to provide an initial signal of biologic effect of new agents. It is clear, however, that in an uncontrolled trial, these measures are subject to criticism since an apparently promising observation may be related to biological factors such as patient selection and not the impact of the intervention. Thus, phase II screening trials utilising these endpoints are best designed with a randomised control. Exceptions may exist where the behaviour patterns of certain cancers are so consistent (and usually consistently poor), that a non-randomised trial is justifiable (see for example van Glabbeke et al.<sup>20</sup>). However, in these cases it will be essential to document with care the basis for estimating the expected PFS or proportion progression-free in the absence of a treatment effect.

### 4.7.2. Phase III trials

Phase III trials in advanced cancers are increasingly designed to evaluate progression-free survival or time to progression as the primary outcome of interest. Assessment of progression is relatively straightforward if the protocol requires all patients to have measurable disease. However, restricting entry to this subset of patients is subject to criticism: it may result in a trial where the results are less likely to be generalisable if, in the disease under study, a substantial proportion of patients would be excluded. Moreover, the restriction to entry will slow recruitment to the study. Increasingly, therefore, trials allow entry of both patients with measurable disease as well as those with non-measurable disease only. In this circumstance, care must be taken to explicitly describe the findings which would qualify for progressive disease for those patients without measurable lesions. Furthermore, in this setting, protocols must indicate if the maximum number of recorded target lesions for those patients with measurable disease may be relaxed from five to three (based on the data found in Bogaerts et al.<sup>10</sup> and Moskowitz et al.<sup>11</sup>). As found in the 'special notes on assessment of progression', these guidelines offer recommendations for assessment of progression in this setting. Furthermore, if available, validated tumour marker measures of progression (as has been proposed for ovarian cancer) may be useful to integrate into the definition of progression. Centralised blinded review of imaging studies or of source imaging reports to verify 'unequivocal progression' may be needed if important drug development or drug approval decisions are to be based on the study outcome. Finally, as noted earlier, because the date of progression is subject to ascertainment bias, timing of investigations in study arms should be the same. The article by Dancey et al. in this special issue<sup>21</sup> provides a more detailed discussion of the assessment of progression in randomised trials

### 4.8. Independent review of response and progression

For trials where objective response (CR + PR) is the primary endpoint, and in particular where key drug development decisions are based on the observation of a minimum number of responders, it is recommended that all claimed responses be reviewed by an expert(s) independent of the study. If the study is a randomised trial, ideally reviewers should be blinded to treatment assignment. Simultaneous review of the patients' files and radiological images is the best approach.

Independent review of progression presents some more complex issues: for example, there are statistical problems with the use of central-review-based progression time in place of investigator-based progression time due to the potential introduction of informative censoring when the former precedes the latter. An overview of these factors and other lessons learned from independent review is provided in an article by Ford et al. in this special issue.<sup>22</sup>

### 4.9. Reporting best response results

### 4.9.1. Phase II trials

When response is the primary endpoint, and thus all patients must have measurable disease to enter the trial, all patients included in the study must be accounted for in the report of the results, even if there are major protocol treatment deviations or if they are not evaluable. Each patient will be assigned one of the following categories:

- 1. Complete response
- 2. Partial response
- 3. Stable disease
- 4. Progression
- Inevaluable for response: specify reasons (for example: early death, malignant disease; early death, toxicity; tumour assessments not repeated/incomplete; other (specify)).

Normally, all eligible patients should be included in the denominator for the calculation of the response rate for phase II trials (in some protocols it will be appropriate to include all treated patients). It is generally preferred that 95% two-sided confidence limits are given for the calculated response rate. Trial conclusions should be based on the response rate for all eligible (or all treated) patients and should not be based on a selected 'evaluable' subset.

#### 4.9.2. Phase III trials

Response evaluation in phase III trials may be an indicator of the relative anti-tumour activity of the treatments evaluated and is almost always a secondary endpoint. Observed differences in response rate may not predict the clinically relevant therapeutic benefit for the population studied. If objective response is selected as a primary endpoint for a phase III study (only in circumstances where a direct relationship between objective tumour response and a clinically relevant therapeutic benefit can be unambiguously demonstrated for the population studied), the same criteria as those applying to phase II trials should be used and all patients entered should have at least one measurable lesion.

In those many cases where response is a secondary endpoint and not all trial patients have measurable disease, the method for reporting overall best response rates must be pre-specified in the protocol. In practice, response rate may be reported using either an 'intent to treat' analysis (all randomised patients in the denominator) or an analysis where only the subset of patients with measurable disease at baseline are included. The protocol should clearly specify how response results will be reported, including any subset analyses that are planned.

The original version of RECIST suggested that in phase III trials one could write protocols using a 'relaxed' interpretation of the RECIST guidelines (for example, reducing the number of lesions measured) but this should no longer be done since these revised guidelines have been amended in such a way that it is clear how these criteria should be applied for all trials in which anatomical assessment of tumour response or progression are endpoints.

238

EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247

|   | RECIST 1.0   | RECIST 1.1  | Rationale  | Reference in special issue<br>(if applicable) |
|---|--|---|--|---|
| Minimum size measurable<br>lesions              | CT: 10 mm spiral<br>20 mm non-spiral                         | CT 10 mm; delete reference to spiral scan   | Most scans used have 5 mm or less slice<br>thickness Clearer to give instruction based on<br>slice interval if it is greater than 5 mm   |   |
|   | Clinical: 20 mm  | Clinical: 10 mm (must be<br>measurable with calipers)   | Caliper measurement will make this reliable  |   |
|   | Lymph node: not mentioned                                    | CT:<br>≥15 mm short axis for target<br>≥10-<15 mm for non-target<br><10 mm is non-pathological  | Since nodes are normal structure need to define<br>pathological enlargement. Short axis is most<br>sensitive   | e Schwartz et al. <sup>25</sup>               |
| pecial considerations on<br>esion measurability | ₩  | Notes included on bone<br>lesions, cystic lesions   | Clarify frequently asked questions   |   |
| werall tumour burden                            | 10 lesions (5 per organ)                                     | 5 lesions (2 per organ)   | Data warehouse analysis shows no loss of<br>information if lesion number reduced from 10 to<br>5. A maximum of 2 lesions per organ yields<br>sufficient representation per disease site  | Bogaerts et al. <sup>20</sup>                 |
| esponse criteria target<br>isease               | CR lymph node not mentioned                                  | CR lymph nodes must be  | In keeping with normal size of nodes   | Schwartz et al.15                             |
|   | PD 20% increase over smallest sum on<br>study or new lesions | DD 20% increase over smallest<br>sum on study (including<br>baseline if that is smallest) and<br>at least 5 mm increase or new<br>lesions   | Clarification that if baseline measurement is<br>smaller than any on study measurement, it is<br>reference against which PD is assessed<br>5 mm absolute increase to guard against over<br>calling PD when total sum is very small and 200<br>increase is within measurement error | 6   |
| esponse criteria non-target<br>isease           | 'unequivocal progression' considered as PD                   | More detailed description of<br>'unequivocal progression' to<br>indicate that it should not<br>normally trump target disease<br>status. It must be<br>representative of overall<br>disease status change, not a<br>single lesion increase | Confusion with RECIST 1.0 where some were<br>considering PD if 'increase' in any non-target<br>lesion, even when target disease is stable or<br>responding   |   |
| New lesions                                     | -  | New section on New lesions  | To provide guidance on when a lesion is<br>considered new (and thus PD)  |   |
| overall response                                | Table integrated target and non-target lesions               | Two tables: one integrating<br>target and non-target and the<br>other of non-target only  | To account for the fact that RECIST criteria are<br>now being used in trials where PFS is the<br>endpoint and not all patients have measurable<br>(target) disease at baseline   | Dancey et al. <sup>21</sup>                   |

|                                  |  | Special notes:<br>How to assess and measure<br>lymph nodes<br>CR in face of residual tissue<br>Discussion of 'equivocal'<br>progression                        | Frequently asked questions on these topics   |                               |
|----------------------------------|--|--|--|-------------------------------|
| Confirmatory measure             | For CR and FR: criteria<br>must be met again 4<br>weeks after initial<br>documentation | Retain this requirement ONLY<br>for<br>non-randomised trials with<br>primary endpoint of response  | Data warehouse shows that response rates<br>rise when confirmation is eliminated, but<br>the only circumstance where this is<br>important is in trials where there is no<br>concurrent comparative control and where<br>this measure is the primary endpoint | Bogaerts et al. <sup>10</sup> |
| Progression-free survival        | General comments only  | More specific comments on<br>use of PFS (or proportion<br>progression-free) as<br>phase II endpoint<br>Greater detail on PFS<br>assessment in phase III trials | Increasing use of FFS in phase III trials<br>requires guidance on assessment of FD in<br>patients with non-measurable disease  | Dancey et al. <sup>21</sup>   |
| Reporting of response<br>results | 9 categories suggested for<br>reporting phase II results                               | Divided into phase II and phase<br>III<br>9 categories collapsed into 5<br>In phase III, guidance given<br>about reporting response                            | Simplifies reporting and clarifies how to<br>report phase II and III data consistently   |                               |
| Response in phase III<br>trials  | More relaxed guidelines<br>possible if protocol specified                              | This section removed and<br>referenced in section<br>above: no need to have<br>different criteria for phase II<br>and III                                      | Simplification of response assessment by<br>reducing number of lesions and eliminating<br>need for confirmation in randomised<br>studies where response is not the primary<br>endpoint makes separate 'rules'<br>unnecessary                                 |                               |
| Imaging appendix                 | Appendix I   | Appendix II: updated with<br>detailed guidance on<br>use of MRI, PET/CT<br>Other practical guidance<br>included  | Evolving use of newer modalities addressed.<br>Enhanced guidance in response to frequent<br>questions and from radiology review<br>experience  |                               |
| New appendices                   |  | Appendix I: comparison of<br>RECIST 1.0 and 1.1<br>Appendix III: frequently asked<br>questions   |  |                               |

EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247

Study number D-FR-01087-001 - Protocol v8.0 Date: 12 June 2020

### Conflict of interest statement

None declared.

240

### Acknowledgements

The RECIST Working Group would like to thank the following organisations which made data bases available to us in order to perform the analyses which informed decisions about changes to this version of the criteria: Amgen; AstraZeneca; Breast Cancer International Research Group (BCIRG); Bristol-Myers Squibb; European Organisation for Research and Treatment of Cancer (EORTC) Breast Cancer Group and Gastrointestinal Group; Erasmus University Medical Center, Rotterdam, The Netherlands; Genentech; Pfizer; RadPharm; Roche; Sanofi Aventis.

We would also like to thank the following individuals from academic, government, and pharmaceutical organisations for providing helpful comments on an earlier draft of these revised guidelines: Ohad Amit, Phil Murphy, Teri Crofts and Janet Begun, GlaxoSmithKline, USA; Laurence H. Baker, Southwest Oncology Group, USA; Karla Ballman, Mayo Clinic, USA; Charles Baum, Darrel Cohen, and Mary Ashford Collier, Pfizer, USA; Gary J. Becker, American Board of Radiology, Tucson, USA; Jean-Yves Blay, University Claude Pertrand, Lyon France; Renzo Canetta, Bristol-Myers Squibb, USA; David Chang, Amgen Inc., USA; Sandra Chica, Perceptive Informations Inc. (PAR-EXEL), USA; Martin Edelman, University of Maryland Greenbaum Cancer Centre, USA; Gwendolyn Fyfe, Genentech, USA; Bruce Giantonio, Eastern Cooperative Oncology Group, USA; Gary Gordon, Abbott Pharmaceuticals, USA; Ronald Gottlieb, Roswell Park Cancer Institute, USA; Simon Kao, University of Iowa College of Medicine, USA; Wasaburo Koizumi, Kitasato University, Japan; Alessandro Riva, Novartis Pharmaceuticals, USA; Wayne Rackhoff, Ortho Biotech Oncology Research and Development, USA; Nagahiro Saijo, President Japanese Society of Medical Oncology, Japan; Mitchell Schnall American College of Radiology Imaging Network, USA; Yoshik Shimamura, PAR-EXEL International Inc., Japan; Rajeshwari Sridhara, Centre for Drug Evaluation and Research, Food and Drug Administration, USA; Andrew Stone, Alan Barge, AstraZeneca, United Kingdom; Orhan Suleiman, Centre for Drug Evaluation and Research, Food and Drug Administration, USA; Daniel C. Sullivan, Duke University Medical Centre, USA; Masakazu Toi, Kyoto University, Japan; Cindy Welsh, Centre for Drug Evaluation and Research, Food and Drug Administration, USA.

Finally, the RECIST Working Group would like to thank individuals who were not permanent members of the group (which are all acknowledged as co-authors) but who attended working group meetings from time to time and made contributions to the total process over the past 7 years: Richard Pazdur, Food and Drug Administration, USA; Francesco Pignatti, European Medicines Agency, London, UK.

## Appendix II. Specifications for standard anatomical radiological imaging

These protocols for image acquisition of computed tomography (CT) and magnetic resonance imaging (MRI) are recommendations intended for patients on clinical trials where RECIST assessment will be performed. Standardisation of imaging requirements and image acquisition parameters is ideal to allow for optimal comparability of subjects within a study and results between studies. These recommendations are designed to balance optimised image acquisition protocols with techniques that should be feasible to perform globally at imaging facilities in all types of radiology practices. These guidelines are not applicable to functional imaging techniques or volumetric assessment of tumour size.

Scanner quality control is highly recommended and should follow standard manufacturer and facility maintenance schedules using commercial phantoms. It is likely that for RE-CIST unidimensional measurements this will be adequate to produce reproducible measurements. Imaging quality control for CT includes an analysis of image noise and uniformity and CT number as well as spatial resolution. The frequency of quality control analysis is also variable and should focus on clinically relevant scanning parameters. Dose analysis is always important and the use of imaging should follow the ALARA principle, As Low As Reasonably Achievable', which refers to making every reasonable effort to maintain radiation exposures as far below the dose limits as possible.

### Specific notes

Chest X-ray measurement of lesions surrounded by pulmonary parenchyma is feasible, but not preferable as the measurement represents a summation of densities. Furthermore, there is poor identification of new lesions within the chest on X-ray as compared with CT. Therefore, measurements of pulmonary parenchymal lesions as well as mediastinal disease are optimally performed with CT of the chest. MRI of the chest should only be performed in extenuating circumstances. Even if IV contrast cannot be administered (for example, in the situation of allergy to contrast), a non-contrast CT of the chest is still preferred over MRI or chest X-ray.

CT scans: CT scans of the chest, abdomen, and pelvis should be contiguous throughout all the anatomic region of interest. As a general rule, the minimum size of a measurable lesion at baseline should be no less than double the slice thickness and also have a minimum size of 10 mm (see below for minimum size when scanners have a slice thickness more than 5 mm). While the precise physics of lesion size and partial volume averaging is complex, lesions smaller than 10 mm may be difficult to accurately and reproducibly measure. While this rule is applicable to baseline scans, as lesions potentially decrease in size at follow-up CT studies, they should still be measured. Lesions which are reported as 'too small to measure' should be assigned a default measurement of 5 mm if they are still visible.

The most critical CT image acquisition parameters for optimal tumour evaluation using RECIST are anatomic coverage, contrast administration, slice thickness, and reconstruction interval.

a. Anatomic coverage: Optimal anatomic coverage for most solid tumours is the chest, abdomen and pelvis. Coverage should encompass all areas of known predilection for metastases in the disease under evaluation and

### EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247

should additionally investigate areas that may be involved based on signs and symptoms of individual patients. Because a lesion later identified in a body part not scanned at baseline would be considered as a new lesion representing disease progression, careful consideration should be given to the extent of imaging coverage at baseline and at subsequent follow-up time points. This will enable better consistency not only of tumour measurements but also identification of new disease.

b. IV contrast administration: Optimal visualisation and measurement of metastases in solid tumours requires consistent administration (dose and rate) of IV contrast as well as timing of scanning. Typically, most abdominal imaging is performed during the portal venous phase and (optimally) about the same time frame after injection on each examination (see Fig. 1 for impact of different phase of IV contrast on lesion measurement). Most solid tumours may be scanned with a single phase after administration of contrast. While triphasic CT scans are sometimes performed on other types of vascular tumours to improve lesion conspicuity, for consistency and uniformity, we would recommend triphasic CT for hepatocellular and neuroendocrine tumours for which this scanning protocol is generally standard of care, and the improved temporal resolution of the triphasic scan will enhance the radiologists' ability to consistently and reproducibly measure these lesions. The precise dose and rate of IV contrast is dependent upon the CT scanning equipment, CT acquisition protocol, the type of contrast used, the available venous access and the medical condition of the patient. Therefore, the method of administration of intravenous contrast agents is variable. Rather than try to institute rigid rules regarding methods for administering contrast agents and the volume injected, it is appropriate to suggest that an adequate volume of a suitable contrast agent should be given so that the metastases are demonstrated to best effect and a consistent method is used on subsequent examinations for any given patient (ideally, this would be specified in the protocol or for an institution). It is very important that the same technique be used at baseline and on follow-up examinations for a given patient. This will greatly enhance the reproducibility of the tumour measurements. If prior to enrolment it is known a patient is not able to undergo CT scans with IV contrast due to allergy or renal insufficiency, the decision as to whether a non-contrast CT or MRI (with or without IV contrast) should be used to evaluate the subject at baseline and follow-up should be guided by the tumour type under investigation and the anatomic location of the disease. For patients who develop contraindications to contrast after baseline contrast CT is done, the decision as to whether non-contrast CT or MRI (enhanced or non-enhanced) should be performed should also be based on the tumour type, anatomic location of the disease and should be optimised to allow for comparison to the prior studies if possible. Each case should be discussed with the radiologist to determine if substitution of these other approaches is possible and, if not, the patient should be considered not evaluable from that point forward. Care must be taken in measurement of target lesions on a different modality and interpretation of non-target disease or new lesions, since the same lesion may appear to have a different size using a new modality (see Fig. 2 for a comparison of CT and MRI of the same lesion). Oral contrast is recommended to help visualise and differentiate structures in the abdomen

c. Slice thickness and reconstruction interval: RECIST measurements may be performed at most clinically obtained slice thicknesses. It is recommended that CT scans be performed at 5 mm contiguous slice thickness or less and indeed this guideline presumes a minimum 5 mm thickness in recommendations for measurable lesion definition. Indeed, variations in slice thickness can have an impact on lesion measurement and on detection of new lesions. However, consideration should also be given for minimising radiation exposure. With these parameters, a minimum 10 mm lesion is considered measurable at baseline. Occasionally, institutions may perform medically acceptable scans at slice thicknesses greater than 5 mm. If this occurs, the minimum size of measurable lesions at baseline should be twice the slice



Fig. 1 – Difference in measurement/visualisation with different phases of IV contrast administration. Hypervascular metastases imaged in the arterial phase (left) and the portal venous phase (right). Note that the number of lesions visible differs greatly between the two phases of contrast administration as does any potential lesion measurement. Consistent CT scan acquisition, including phase of contrast administration, is important for optimal and reproducible tumour

### EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247



Fig. 2 - CT versus MRI of same lesions showing apparent 'progression' due only to differing method of measurement.

thickness of the baseline scans. Most contemporary CT scanners are multidetector which have many imaging options for these acquisition parameters.<sup>23</sup> The equipment vendor and scanning manual should be reviewed if there are any specific system questions.

d. Alternative contrast agents: There are a number of other, new contrast agents, some organ specific.<sup>24</sup> They may be used as part of patient care for instance, in liver lesion assessment, or lymph node characterisation<sup>25</sup>, but should not as yet be used in clinical trials.

FDG-PET has gained acceptance as a valuable tool for detecting, staging and restaging several malignancies. Criteria for incorporating (or substituting) FDG-PET into anatomical assessment of tumour response in phase II trials are not yet available, though much research is ongoing. Nevertheless, FDG-PET is being used in many drug development trials both as a tool to assess therapeutic efficacy and also in assessment of progression. If FDG-PET scans are included in a protocol, by consensus, an FDG uptake period of 60 min prior to imaging has been decided as the most appropriate for imaging of patients with malignancy.26 Whole-body acquisition is important since this allows for sampling of all areas of interest and can assess if new lesions have appeared thus determining the possibility of interval progression of disease. Images from the base of the skull to the level of the mid-thigh should be obtained 60 min post injection. PET camera specifications are variable and manufacturer specific, so every attempt should be made to use the same scanner, or the same model scanner, for serial scans on the same patient. Whole-body acquisitions can be performed in either 2- or 3-dimensional mode with attenuation correction, but the method chosen should be consistent across all patients and serial scans in the clinical trial.

PET/CT scans: Combined modality scanning such as with PET-CT is increasingly used in clinical care, and is a modality/technology that is in rapid evolution; therefore, the recommendations in this paper may change rather quickly with time. At present, low dose or attenuation correction CT portions of a combined PET-CT are of limited use in anatomically based efficacy assessments and it is therefore suggested that they should not be substituted for dedicated diagnostic contrast enhanced CT scans for anatomically based RECIST measurements. However, if a site can document that the CT performed as part of a PET-CT is of identical diagnostic quality to a diagnostic CT (with IV and oral contrast) then the CT portion of the PET-CT can be used for RECIST measurements. Note, however, that the PET portion of the CT introduces additional data which may bias an investigator if it is not routinely or serially performed.

Ultrasound examinations should not be used in clinical trials to measure tumour regression or progression of lesions because the examination is necessarily subjective and operator dependent. The reasons for this are several: Entire examinations cannot be reproduced for independent review at a later date, and it must be assumed, whether or not it is the case, that the hard-copy films available represent a true and accurate reflection of events. Furthermore, if, for example, the only measurable lesion is in the para-aortic region of the abdomen and if gas in the bowel overlies the lesion, the lesion will not be detected because the ultrasound beam cannot penetrate the gas. Accordingly, the disease staging (or restaging for treatment evaluation) for this patient will not be accurate.

While evaluation of lesions by physical examination is also of limited reproducibility, it is permitted when lesions are superficial, at least 10 mm size, and can be assessed using calipers. In general, it is preferred if patients on clinical trials have at least one lesion that is measurable by CT. Other skin or palpable lesions may be measured on physical examination and be considered target lesions.

Use of MRI remains a complex issue. MRI has excellent contrast, spatial and temporal resolution; however, there are many image acquisition variables involved in MRI, which greatly impact image quality, lesion conspicuity and measurement. Furthermore, the availability of MRI is variable globally. As with CT, if an MRI is performed, the technical specifications of the scanning sequences used should be optimised for the evaluation of the type and site of disease. Furthermore, as with CT, the modality used at follow-up should be the same as was used at baseline and the lesions should be measured/assessed on the same pulse sequence. Generally, axial imaging of the abdomen and pelvis with T1 and T2 weighted imaging along with gadolinium enhanced imaging should be performed. The field of view, matrix, number of excitations, phase encode steps, use of fat suppression and fast sequences should be optimised for the spe

cific body part being imaged as well as the scanner utilised. It is beyond the scope of this document or appendix to prescribe specific MRI pulse sequence parameters for all scanners, body parts and diseases. Ideally, the same type of scanner should be used and the image acquisition protocol should be followed as closely as possible to prior scans. Body scans should be performed with breath-hold scanning techniques if possible.

Selection of target lesions: In general, the largest lesions representative of involved organs (up to a maximum of two per organ and five total) are selected to follow as target lesions. However, in some cases, the largest lesions may not be easily measured and are not suitable for follow-up because of their configuration. In these cases, identification of the largest most reproducible lesions is advised. Fig. 3 provides an illustrative example where the largest lesion is not the most reproducible and another lesion is better to select and follow:

### Measurement of lesions

The longest diameter of selected lesions should be measured in the plane in which the images were acquired. For body CT, this is the axial plane. In the event isotropic reconstructions are performed, measurements can be made on these reconstructed images; however, it should be cautioned that not all radiology sites are capable of producing isotropic reconstructions. This could lead to the undesirable situation of measurements in the axial plane at one assessment point and in a different plane at a subsequent assessment. There are some tumours, for instance paraspinal lesions, which are better measured in the coronal or sagittal plane. It would be acceptable to measure these lesions in these planes if the reconstructions in those planes were isotropic or the images were acquired with MRI in those planes. Using the same plane of evaluation, the maximal diameter of each target lesion should always be measured at subsequent follow-up time points even if this results in measuring the lesion at a different slice level or in a different orientation or vector compared with the baseline study. Software tools that calculate the maximal diameter for a perimeter of a tumour may be employed and may even reduce variability.

The only exception to the longest diameter rule is lymph node measurement. Because malignant nodes are identified by the length of their short axis, this is the guide used to determine not only whether they are pathological but is also the dimension measured for adding into the sum of target lesions. Fig. 4 illustrates this point: the large arrow identifies a malignant node: the shorter perpendicular axis is  $\geq 15$  mm and will be recorded. Close by (small arrow) there is a normal node: note here the long axis is greater than 10 mm but the short axis is well below 10 mm. This node should be considered non-pathological.

If a lesion disappears and reappears at a subsequent time point it should continue to be measured. However, the patient's response at the point in time when the lesion reappears will depend upon the status of his/her other lesions. For example, if the patient's tumour had reached a CR status and the lesion reappeared, then the patient would be considered PD at the time of reappearance. In contrast, if the tumour status was a PR or SD and one lesion which had disappeared then reappears, its maximal diameter should be added to the sum of the remaining lesions for a calculated response: in other words, the reappearance of an apparently 'disappeared' single lesion amongst many which remain is not in itself en-



Fig. 3 – Largest lesion may not be most reproducible: most reproducible should be selected as target. In this example, the primary gastric lesion (circled at baseline and at follow-up in the top two images) may be able to be measured with thin section volumetric CT with the same degree of gastric distention at baseline and follow-up. However, this is potentially challenging to reproduce in a multicentre trial and if attempted should be done with careful imaging input and analysis. The most reproducible lesion is a lymph node (circled at baseline and at follow-up in the bottom two images).



Fig. 4 – Lymph node assessment: large arrow illustrates a pathological node with the short axis shown as a solid line which should be measured and followed. Small arrow illustrates a non-pathological node which has a short axis <10 mm.

ough to qualify for PD: that requires the sum of all lesions to meet the PD criteria. The rationale for such a categorisation is based upon the realisation that most lesions do not actually 'disappear' but are not visualised because they are beyond the resolving power of the imaging modality employed.

The identification of the precise boundary definition of a lesion may be difficult especially when the lesion is embed-

ded in an organ with a similar contrast such as the liver, pancreas, kidney, adrenal or spleen. Additionally, peritumoural oedema may surround a lesion and may be difficult to distinguish on certain modalities between this oedema and actual tumour. In fact, pathologically, the presence of tumour cells within the oedema region is variable. Therefore, it is most critical that the measurements be obtained in a reproducible manner from baseline and all subsequent follow-up timepoints. This is also a strong reason to consistently utilise the same imaging modality.

When lesions 'fragment', the individual lesion diameters should be added together to calculate the target lesion sum. Similarly, as lesions coalesce, a plane between them may be maintained that would aid in obtaining maximal diameter measurements of each individual lesion. If the lesions have truly coalesced such that they are no longer separable, the vector of the longest diameter in this instance should be the maximal longest diameter for the 'merged lesion'.

### Progression of non-target lesions

To achieve 'unequivocal progression' there must be an overall level of substantial worsening in non-target disease that is of a magnitude that, even in the presence of SD or PR in target disease, the treating physician would feel it important to change therapy. Examples of unequivocal progression are shown in Figs. 5 and 6.



Fig. 5 - Example of unequivocal progression in non-target lesions in liver.



Fig. 6 - Example of unequivocal progression in non-target lesion (nodes).

## Appendix III. Frequently asked questions

| Question  | Answer   |  |  |
|---|--|--|--|
| What should be done if several unique lesions at<br>baseline become confluent at a follow-up<br>evaluation?   | Measure the longest diameter of the confluent mass and record to add into the sum of the longest diameters   |  |  |
| How large does a new lesion have to be to count<br>as progression? Does any small subcentimetre<br>lesion qualify, or should the lesion be at least<br>measurable?  | New lesions do not need to meet 'measurability criteria' to be considered valid. If it is<br>clear on previous images (with the same technique) that a lesion was absent then its<br>definitive appearance implies progression. If there is any doubt (because of the<br>techniques or conditions) then it is suggested that treatment continue until next<br>scheduled assessment when, generally, all should be clear. Either it gets bigger and the<br>date of progression is the date of the first suspicion, or it disappears and one may then<br>consider it an artefact with the support of the radiologists  |  |  |
| How should one lesion be measured if on<br>subsequent exams it is split into two?   | Measure the longest diameter of each lesion and add this into the sum  |  |  |
| Does the definition of progression depend on the status of all target lesions or only one?  | As per the RECIST 1.1 guideline, progression requires a 20% increase in the sum of<br>diameters of all target lesions AND a minimum absolute increase of 5 mm in the sum   |  |  |
| Are RECIST criteria accepted by regulatory agencies?  | Many cooperative groups and members of pharma were involved in preparing RECIST<br>1.0 and have adopted them. The FDA was consulted in their development and supports<br>their use, though they don't require it. The European and Canadian regulatory<br>authorities also participated and the RECIST criteria are now integrated in the European<br>note for guidance for the development of anticancer agents. Many pharmaceutical<br>companies are also using them. RECIST 1.1 was similarly widely distributed before<br>publication  |  |  |
| What is the criterion for a measurable lesion if<br>the CT slice thickness is >5 mm?  | RECIST 1.1 recommends that GT scans have a maximum slice thickness of 5 mm and the minimum size for a measurable lesion is twice that: 10 mm (even if slice thickness is <5 mm). If scanners with slice thickness >5 mm are used, the minimum lesion size must have a longest diameter twice the actual slice thickness  |  |  |
| What should we record when target lesions<br>become so small they are below the 10 mm<br>'measurable' size?   | Target lesion measurability is defined at baseline. Thereafter, actual measurements,<br>even if <10 mm, should be recorded. If lesions become very small, some radiologists<br>indicate they are 'too small to measure'. This guideline advises that when this occurs, if<br>the lesion is actually still present, a default measurement of 5 mm should be applied. If<br>in fact the radiologist believes the lesion has gone, a default measurement of 0 mm<br>should be recorded  |  |  |
| If a patient has several lesions which have<br>decreased in size to meet PR criteria and one<br>has actually disappeared, does that patient have<br>PD if the 'disappeared' lesion reappears?   | Unless the sum meets the PD criteria, the reappearance of a lesion in the setting of PR (or SD) is not PD. The lesion should simply be added into the sum.<br>If the patients had had a CR, clearly reappearance of an absent lesion would qualify for PD  |  |  |
| When measuring the longest diameter of target<br>lesions in response to treatment, is the same<br>axis that was used initially used subsequently,<br>even if there is a shape change to the lesion that<br>may have produced a new longest diameter?        | The longest diameter of the lesion should always be measured even if the actual axis is different from the one used to measure the lesion initially (or at different time point during follow-up)<br>The only exception to this is lymph nodes: as per RECIST 1.1 the short axis should always be followed and as in the case of target lesions, the vector of the short axis may change on follow-up  |  |  |
| Target lesions have been selected at baseline<br>and followed but then one of these target<br>lesions then becomes non-evaluable (i.e.<br>different technique used)<br>What is the effect this has on the other target<br>lesions and the overall response? | What may be done in such cases is one of the following:<br>(a) If the patient is still being treated, call the centre to be sure that future evaluations are<br>done with the baseline technique so at least SOME courses are fully evaluable<br>(b) If that is not possible, check if there IS a baseline exam by the same technique which<br>was used to follow patientsin which case if you retrieve the baseline measures from<br>that technique you retrieve the lesion evaluability<br>(c) If neither (a) nor (b) is possible then it is a judgement call about whether you delete<br>the lesion from all forms or consider the impact of the lesion overall is so important that<br>its being non-evaluable makes the overall response interpretation inevaluable without<br>it. Such a decision should be discussed in a review panel<br>It is NOT recommended that the lesion be included in baseline sums and then excluded<br>from follow-up sums since this biases in favour of a response<br>(continued on next page) |  |  |

| Question  | Answer   |
|---|--|
| What if a single non-target lesion cannot be reviewed, for whatever reason; does this negate the overall assessment?  | Sometimes the major contribution of a single non-target lesion may be in<br>the setting of CR having otherwise been achieved: failure to examine one<br>non-target in that setting will leave you unable to claim CR. It is also<br>possible that the non-target lesion has undergone such substantial<br>progression that it would override the target disease and render patient<br>PD. However, this is very unlikely, especially if the rest of the measurable<br>disease is stable or responding  |
| A patient has a 32% decrease in sum cycle 2, a 28% decrease cycle<br>4 and a 33% decrease cycle 6. Does confirmation of PR have to<br>take place in sequential scans or is a case like this confirmed PR? | It is not infrequent that tumour shrinkage hovers around the 30% mark<br>In this case, most would consider PR to have been confirmed looking at<br>this overall case. Had there been two or three non-PR observations<br>between the two time point PR responses, the most conservative<br>approach would be to consider this case SD  |
| In the setting of a breast cancer neoadjuvant study, would<br>mammography not be used to assess lesions? Is CT preferred in<br>this setting?  | Neither CT nor mammography are optimal in this setting. MRI is the preferred modality to follow breast lesions in a neoadjuvant setting  |
| A patient has a lesion measurable by clinical exam and by CT<br>scan. Which should be followed?   | $\ensuremath{CT}$ scan. Always follow by imaging if that option exists since it can be reviewed and verified   |
| A lesion which was solid at baseline has become necrotic in the<br>centre. How should this be measured?   | The longest diameter of the entire lesion should be followed. Eventually<br>necrotic lesions which are responding to treatment decrease in size. In<br>reporting the results of trials, you may wish to report on this<br>phenomenon if it is seen frequently since some agents (e.g. angiogenesis<br>inhibitors) may produce this effect  |
| If I am going to use MRI to follow disease, what is minimum size<br>for measurability?  | MRI may be substituted for contrast enhanced CT for some sites, but no<br>lung. The minimum size for measurability is the same as for CT (10 mm<br>as long as the scans are performed with slice thickness of 5 mm and no<br>gap. In the event the MRI is performed with thicker slices, the size of a<br>measurable lesion at baseline should be two times the slice thickness. Ir<br>the event there are inter-slice gaps, this also needs to be considered in<br>determining the size of measurable lesions at baseline   |
| Can PET-CT be used with RECIST?   | At present, the low dose or attenuation correction CT portion of a<br>combined PET-CT is not always of optimal diagnostic CT quality for use<br>with RECIST measurements. However, if your site has documented that<br>the CT performed as part of a PET-CT is of the same diagnostic quality as<br>a diagnostic CT (with IV and oral contrast) then the PET-CT can be used<br>for RECIST measurements. Note, however, that the PET portion of the CT<br>introduces additional data which may bias an investigator if it is not<br>routinely or serially performed |

### REFERENCES

- 1. Paesmans M, Sculier JP, Libert P, et al. Response to chemotherapy has predictive value for further survival of patients with advanced non-small cell lung cancer: 10 years experience of the European Lung Cancer Working Party. Eur J Cancer 1997;33:2326-32
- 2. Buyse M, Thirion P, Carlson RW, et al. Relation between tumor response to first-line chemotherapy and survival in advanced colorectal cancer: a meta-analysis. Meta-analysis group in Cancer. Lancet 2000;356:373-8.
- Goffin J, Baral S, Tu D, et al. Objective responses in patients with malignant melanoma or renal cell cancer in early clinical studies do not predict regulatory approval. Clin Cancer Res 2005;15:5928–34. 4. El-Maraghi RH, Eisenhauer EA. Review of phase II trial designs
- used in studies of molecular targeted agents: outcomes and predictors of success in phase III. J Clin Oncol 2008;10:1346-54.
- 5. Miller AB, Hoogstraten B, Staquet M, Winkler A. Reporting results of cancer treatment. Cancer 1981;47:207-14.

- 6. Tonkin K, Tritchler D, Tannock I. Criteria of tumor response used in clinical trials of chemotherapy. J Clin Oncol 1985;3:870-5.
- 7. Baar J, Tannock I. Analyzing the same data in two ways: a demonstration model to illustrate the reporting and misreporting of clinical trials. J Clin Oncol 1989;7:969–78.
- 8. Therasse P, Arbuck SG, Eisenhauer EA, et al. New guidelines to evaluate the response to treatment in solid tumors (RECIST Guidelines). J Natl Cancer Inst 2000;92:205–16.
- 9. Therasse P, Eisenhauer EA, Verweij J. RECIST revisited: a review of validation studies on tumour assessment. Eur J Cancer 2006;42:1031-9.
- 10. Bogaerts J, Ford R, Sargent D, et al. Individual patient data analysis to assess modifications to the RECIST criteria. Eur J Cancer 2009:45:248-60.
- 11. Moskowitz CS, Jia X, Schwartz LH, Gönen M. A simulation study to evaluate the impact of the number of lesions measured on response assessment. Eur J Cancer 2009:45:300-10.
- 12. Sargent D, Rubinstein L, Schwartz L, et al. Validation of novel imaging methodologies for use as cancer clinical trials end-points. Eur J Cancer 2009;45:290-9.

### EUROPEAN JOURNAL OF CANCER 45 (2009) 228-247

- Macdonald DR, Cascino TL, Schold Jr SC, Cairneross JG. Response criteria for phase II studies of supratentorial malignant glioma. J Clin Oncol 1990;8:1277–80.
- Cheson BD, Pfistner B, Juweid ME, et al. Revised response criteria for malignant lymphoma. J Clin Oncol 2007;10:579-86.
- Schwartz LH, Bogaerts J, Ford R, et al. Evaluation of lymph nodes with RECIST 1.1. Eur J Cancer 2009;45:261–7.
- Rustin GJ, Quinn M, Thigpen T, et al. Re: New guidelines to evaluate the response to treatment in solid tumors (ovarian cancer). J Natl Cancer Inst 2004;96:487–8.
- Bubley GJ, Carducci M, Dahut W, et al. Eligibility and response guidelines for phase II clinical trials in androgen-independent prostate cancer: recommendations from the Prostate-Specific Antigen Working Group. J Clin Oncol 1999;17:3461–7.
- Scher H, Halabi S, Tannock I, et al. Design and end points of clinical trials for patients with progressive prostate cancer and castrate levels of testosterone: recommendations of the Prostate Cancer Clinical Trials Working Group. J Clin Oncol 2008;26:1148–59.
- Vergote I, Rustin GJ, Eisenhauer EA, et al. Re: new guidelines to evaluate the response to treatment in solid tumors [ovarian cancer]. Gynecologic Gancer Intergroup. J Natl Cancer Inst 2000;92:1534–5.

- Van Glabbeke M, Verweij J, Judson I, Nielsen OS. EORTC Soft Tissue and Bone Sarcoma Group: Progression-free rate as the principal end-point for phase II trials in soft-tissue sarcomas. Eur J Cancer 2002;38:543–9.
- Dancey JE, Dodd LE, Ford R, et al. Recommendations for the assessment of progression in randomised cancer treatment trials. Eur J Cancer 2009;45:281–9.
- Ford R, Schwartz L, Dancey J, et al. Lessons learned from independent central review. Eur J Cancer 2009;45: 268–74.
- Catalano C, Francone M, Ascarelli A, Mangia M, Iacucci I, Passariello R. Optimizing radiation dose and image quality. Eur Radioi 2007;17(Suppl 6):F26-32.
- Low RN. Abdominal MRI advances in the detection of liver tumours and characterization. Lancet Oncol 2007;8(6):525–35.
- Barrett T, Choyke PL, Kobayashi H. Imaging of the lymphatic system: new horizons. Contrast Media Mol Imaging 2006;1(6):230–45.
- Shankar LK, Hoffman JM, Bacharach S, et al. National Cancer Institute. Consensus recommendations for the use of 18F-FDG PET as an indicator of therapeutic response in patients in National Cancer Institute Trials. J Nucl Med 2006;47(6):1059–66.

Study number D-FR-01087-001 - Protocol v 8.0 Date: 12 June 2020

# 19.5 Attachment 5 – Statistical Details and Simulation Results for Implementation of Adaptive Bayesian Dose Escalation Design

## 19.5.1 Statistical Model

After study screening procedures, eligible subjects will receive up to 2 i.v. administrations of <sup>177</sup>Lu-3BP-227 in two consecutive treatment cycles 4 weeks apart. A range of doses will be tested, including possibly 2.5, 3, 3.5, 4, 4.5, 5, 5.5, 6, 6.5, 7 and 7.5 GBq per cycle. Additional or alternative dose levels may be chosen. These possible dose levels are labelled as d=1, ..., N, where N is the maximum number of dose levels for the study.

The analysis is based on a method reported by Guede et al.[9] and Tibaldi et al. [37].

If, during Cycle 1, a subject receives dose  $d_1$  and in Cycle 2 dose  $d_2$ , the probability, p, of observing a subject with DLTs up to 4 weeks after the two cycles of dosing is modelled as a function of dose as:

$$\text{logit}(p) = \text{logit}(p_0) + \frac{\text{logit}(33\%) - \text{logit}(p_0)}{MTSA} d_1 + \beta * d_2$$

where  $p_0$ , MTSA and  $\beta$  are the model parameters having the following interpretation:

- $p_0$  is the probability of observing DLTs under placebo ( $d_1=d_2=0$ ),
- MTSA is the maximum tolerated single activity, being defined as the dose  $d_1$  at which the predicted probability of DLTs is 33% for dosing Cycle 1 (single dose), and
- $\beta$  is a slope coefficient indicating the impact of a second dose d<sub>2</sub> on DLT probability.

The prior distribution on  $p_0$ , MTSA and  $\beta$  determine the prior distribution on the probability of DLT as a function of up to two doses. For cycle 1 data analysis, the dose  $d_2$  is set to zero. So, only parameters  $p_0$  and MTSA are involved.  $\beta$  is involved when subjects undergo their second cycle. A positive  $\beta$  indicates an increase in DLT probability with second dose.

The MTCA is defined as the maximum tolerated cumulative activity  $(d=d_1+d_2)$ , where  $d_1=d_2$ ) that may be administered following fractionated i.v. administrations of the same dose  $d_1$  at least 4 weeks apart so that no more than 33% of the subjects experience a DLT during Cycles 1 and/or 2.

The MTCA can be presented as a derived parameter from the logistic regression model. It is expressed as:

MTCA = 2\* MTSA \* 
$$\frac{\text{logit } 33\% - \text{logit } p_0}{\text{logit } 33\% - \text{logit } p_0 + \beta * MTSA}.$$

The MTCA will be equal to twice the MTSA when  $\beta$  equals 0 (i.e. the DLT probability is not changed after a second dose). When the DLT probability is increased ( $\beta$ >0), the MTCA will be decreased (and vice versa).

During the study, the target organ exposure will be monitored after each dose and doses may be adapted if the exposure exceeds organ-specific maximum acceptability limits. As such, if the predicted MTCA exceeds the maximum acceptability limits for target organ exposure, the maximum radioactivity dose will be used instead. This feature was not considered in the simulation study, where it was assumed that a cumulative dose up to 15 GBq would induce an acceptable organ exposure.

Prior distributions were elicited based on historical knowledge about the placebo DLT rate and the MTSA target. The following prior distributions are used in the trial:

 $p_0 \sim beta(a = 1, b = 9),$ MTSA~gamma (shape = 4, rate = 0.5), and

## $\beta \sim normal \ (\mu = 0, sd = 0.25).$

Based on these distributions, 20 dose-DLT curves from the prior distribution were simulated after Dose 1 (black) and after Dose 1 and 2 (red) (see Figure 3). The prior probability distributions for the MTSA is summarised in Table 10. The prior probability distributions for the MTCA is summarised in Table 11



Figure 3 40 Random Observations (Curves) from the Prior Distribution

Table 10 Prior Probability Distribution of MTSA

| Dose range for Cycle 1 | Prior MTSA probability |
|------------------------|------------------------|
| 2.5 GBq                | 5.2%                   |
| 2.5 to 5 GBq           | 20.6%                  |
| 5 to 7.5 GBq           | 28.6%                  |
| >7.5 GBq               | 45.6%                  |

MTSA=Maximum tolerated single activity.

 Table 11
 Prior Probability Distribution of MTCA

| Dose range (cumulative) | Prior MTCA probability |
|-------------------------|------------------------|
| ≤5 GBq                  | 32.5%                  |
| 5 to 10 GBq             | 24.6%                  |
| 10 to 15 GBq            | 15.7%                  |
| >15 GBq                 | 27.2%                  |

MTCA= Maximum tolerated cumulative activity.

Given any set of data, the prior distribution can be updated, yielding a posterior distribution of the parameters. From this posterior distribution many different quantities of relevance can be calculated:

- $E(p|d_1, d_2)$ =Posterior mean probability of DLT for Cycle 1 dose  $d_1$  and Cycle 2 dose  $d_2$ .
- TS<sub>d</sub>=(tolerable) Probability that the DLT rate is below 0.33,  $Pr(p \le 0.33 | d,0)$ , for a Cycle 1 dose d (d<sub>2</sub>=0).
- QS<sub>d</sub>=Probability of Cycle 1 dose d being the maximum dose where  $E(p|d,0) \le 0.33$ .
- TC<sub>d</sub>=(tolerable) Probability that the DLT rate is below 0.33,  $Pr(p \le 0.33 | d/2, d/2)$ , for a cumulative dose d (d<sub>1</sub>=d<sub>2</sub>=d/2).
- QC<sub>d</sub>=Probability of cumulative dose d being the maximum dose where  $E(p|d/2,d/2) \leq 0.33$ .

Note that, in this model, the MTSA and MTCA are parameters. They have a probability distribution. A consequence of the Bayesian modelling is the probabilities,  $QS_d$ , which is the probability that a dose is the MTSA for Cycle 1 and  $QC_d$ , which is the probability that a cumulative fractional dose is the MTCA after Cycle 2. This is a powerful question that cannot be answered from a frequentist perspective.

## 19.5.2 Trial Simulation Details

This section present results of trial simulation study that was performed to determine the operational characteristics of the adaptive dose escalation procedure. The design of that simulation study is very similar to the actual study design, with some changes necessary to enable the automation of trial simulations. The impact of any discrepancy should be negligible.

The trial design involves two sets of rules, namely:

(1) dose-escalation rules for the next cohort when the current cohort completes Cycle 2, and

(2) rules for Cycle 2 dosing after completion of Cycle 1.

A cohort will be considered as ended once three subjects of the cohort complete Cycle 2 or early discontinue during Cycle 2. Subjects with DLTs in Cycle 1 will be discontinued prior to Cycle 2.

Once five subjects are enrolled in a cohort, the cohort will be considered as ended, after all five subjects complete Cycle 2 or discontinue early during Cycle 1 or 2.

For the purpose of simulations, the dose decisions will be made at the following milestones:

- Cycle 2 dosing will be determined only once the full cohort, including any replacement subjects, completes Cycle 1 (as opposed to on an individual basis in the study).
- The dose for the next cohort will be decided once the current cohort has ended.

The timing of these decisions will slightly vary during the study, as it is currently planned to decide the Cycle 2 dose on an individual subject basis and to decide for the next cohort once up to three subjects in the current cohort are 3 weeks into Cycle 2. The impact of these changes should be minimal.

The inter cohort dose escalation will be performed as follows:

- (1) The starting dose will be d=5 GBq fractionated into two administrations of  $d_1=d_2=2.5$  GBq per cycle and the first cohort will include three subjects.
- (2) If there is no DLT during Cycle 1, the radioactivity for the next cohort will be increased by 3 GBq (1.5 GBq/cycle) up to 11 GBq (5.5 GBq/cycle) and then by 2 GBq (1 GBq/cycle) up to the last planned cohort of 15 GBq (7.5 GBq/cycle). Cohort size will be three subjects.
- (3) If at least one subject in a cohort reports any DLT, wait for the corresponding cohort (including any replacement) to end the study, then fit the Bayesian model to all available Cycle 1 and Cycle 2 data, and select next dose as the dose d at which QS<sub>d</sub> is maximum, or the maximum allowed dose if any of the following constraints are met:
  - (a) if there are two DLTs out of three, three DLTs out of five, or three DTLs out of six subjects receiving a single or cumulative dose, d must be lower than or equal to that dose;
  - (b) if there are three DLTs out of three, or at least three DLTs out of four, or at least four DLTs out of five, or at least four DLTs out of six subjects receiving a single or cumulative dose, d must be lower than that dose;
  - (c) dose d must be at most 1 GBq larger than the maximum Cycle 1 dose tested previously.

Repeat steps 2 and 3 again, assigning the subjects/cohort to this dose. Terminate study enrolment when any of the following stopping rules happen:

- A the MTSA is precisely estimated: CV (MTSA) calculated as the inter-quartile range over the median is lower than 30%;
- B maximum possible dose (7.5 GBq) is safe:  $TS_{7.5}$ >80% and  $d_1$ =7.5 GBq administered;
- C minimum dose tested (2.5 GBq) is toxic:  $TS_{2.5} < 20\%$ ;
- D total sample size for next dose is already  $\geq 6$ .

The MTSA will be declared to be found when study stops for criteria A or D.

Individual dosing decision for the second treatment cycle will be done as follows:

- Subject will be eligible for Cycle 2 only if subject had no DLT during Cycle 1. Subjects with DLTs in Cycle 1 will be discontinued and replaced with a maximum of two replacements.
- The same fractional dose as in Cycle 1 will be repeated during Cycle 2 ( $d_2=d_1$ ), if the dose having the largest probability to be the MTCA is larger than 2 x  $d_1$ .
- The fractional dose for Cycle 2 will be decreased to  $d_2$ =MTCA- $d_1$  dose, if the predicted cumulative exposure after Cycle 2 is above the MTCA target. In this situation, the cumulative exposure after two cycles should not exceed the MTCA.

The study will be terminated when all cohorts have ended the study, as defined above.

## 19.5.3 Simulation Scenarios

Five dose-DLT profiles were considered to evaluate the operational characteristics of the adaptive dose escalation trial (see Figure 4 and Table 12 below). The five scenarios are:

Scenario A – Maximum dose safe: This scenario represents a negative scenario under which all doses are safe for the two dosing occasions. The true MTSA (15 GBq) and true MTCA (30 GBq) are larger than the maximum possible doses in the study.

**Scenario B - Late MTCA:** This is a scenario is which the MTCA=MTSA=14 GBq. All Cycle 1 doses are safe. After Cycle 2, cumulative doses up to and including 14 GBq are safe. The probability of a DLT is gradually increasing across all dose levels.

Scenario C – Early MTCA: In this scenario, the rate of DLT progression is faster over doses so that the MTSA (7 GBq) during Cycle 1 appears near the end of the dosage range. The MTCA after Cycle 2 is 9 GBq.

**Scenario D - Toxic**: Under this scenario, all doses are toxic, with a DLT rate of 46% at the lowest single dose of 2.5 GBq or at the repeated cumulative dose of 5 GBq. The MTSA (2 GBq) is below but close to the lowest single dose of 2.5 GBq. The MTCA is also equal to 2 GBq, so that any Cycle 2 dosing is toxic.

**Scenario E - Peak:** In this scenario, there is a step function in the probability of DLTs occurring in the middle of the dose escalation process during Cycle 1 and early in Cycle 2. The first doses up to and including MTSA=5.5 GBq are very safe (p=5%), and all subsequent single doses are toxic (p=80%). After Cycle 2, the MTCA is 5 GBq and all subsequent doses are toxic.



| Table 12 | <b>Simulated Dose-DLT Probabilities</b> |
|----------|---|
| 1        |   |

| Dose (GBq) | A: Safe | B: Late MTD | C: Early MTD | D: Toxic | E: Peak |
|------------|---------|-------------|--------------|----------|---------|
| Cycle 1    |         |             |              |          |         |
| 2.5        | 7%      | 7%          | 10%          | 46%      | 5%      |
| 3.0        | 8%      | 8%          | 12%          | 60%      | 5%      |
| 3.5        | 8%      | 8%          | 14%          | 72%      | 5%      |
| 4.0        | 9%      | 9%          | 16%          | 82%      | 5%      |
| 4.5        | 9%      | 10%         | 18%          | 89%      | 5%      |
| 5.0        | 10%     | 10%         | 21%          | 93%      | 5%      |
| 5.5        | 11%     | 11%         | 23%          | 96%      | 5%      |
| 6.0        | 11%     | 12%         | 26%          | 98%      | 80%     |
| 6.5        | 12%     | 13%         | 30%          | 99%      | 80%     |
| 7.0        | 13%     | 14%         | 33%          | 99%      | 80%     |
| 7.5        | 14%     | 15%         | 37%          | 100%     | 80%     |
| Cycle 2    |         |             |              |          |         |
| 5.0        | 7%      | 10%         | 15%          | 46%      | 5%      |
| 6.0        | 8%      | 12%         | 18%          | 60%      | 80%     |
| 7.0        | 8%      | 13%         | 21%          | 72%      | 80%     |
| 8.0        | 9%      | 15%         | 26%          | 82%      | 80%     |
| 9.0        | 9%      | 17%         | 30%          | 89%      | 80%     |
| 10.0       | 10%     | 20%         | 36%          | 93%      | 80%     |
| 11.0       | 11%     | 22%         | 41%          | 96%      | 80%     |
| 12.0       | 11%     | 25%         | 47%          | 98%      | 80%     |
| 13.0       | 12%     | 28%         | 53%          | 99%      | 80%     |
| 14.0       | 13%     | 32%         | 58%          | 99%      | 80%     |
| 15.0       | 14%     | 35%         | 64%          | 100%     | 80%     |

DLT=dose limiting toxicity, MTD=maximum tolerated dose.

## **19.5.4** Operational Characteristics

We used 1000 trial simulations to monitor the following properties of the adaptive design under the five scenarios:

- accuracy and precision of the MTSA and MTCA
- probability of archiving each of the four stopping rules (MTSA precision, last dose is safe, first dose is toxic, maximum sample size)
- total sample size after Cycle 1 and Cycle 2

- total number of cohorts
- number and proportion of subjects being overdosed during Cycle 1 and Cycle 2 (i.e. relative to the true MTSA and MTCA, respectively)
- subject distribution across dose levels.

Calculations were performed in R statistical software version 2.7.2 and using the BRugs package running on OpenBUGS version 2.2.0 for the Bayesian estimation. The posterior distribution was estimated using one MCMC chain based on 1000 iterations, after a burn-in of 2000 samples. The initial values were set to  $p_0=5\%$ , MTD=5 GBq and  $\beta=0$ . Convergence was monitored by visual inspection of the diagnosis plots for a sample dataset spread across all five scenarios.

## 19.5.5 Simulation Results

The operational characteristics for the five trial simulation scenarios are presented in Table 13. The mean (SD) sample size per dose is shown in Figure 5.

A box plot of the total sample size is in Figure 6 for Cohort 1 and in

Figure 8 for Cohort 2.

The number of cohorts is in

## Figure 8

Overall, simulations performed as expected, under all scenarios, taking into account the constraints about the maximum fold-increase, the repeated dosing, the overdose control and the maximum sample size. The dose escalation for Cycle 1 was pursued up to the top single dose of 7.5 GBq under a similar pattern for the Safe, Early and Late scenarios, where the MTSA was either above the top dose (Safe, Late) or close to it (Early). Under the Toxic scenario, only low doses were administered, with an MTSA estimated near the first single dose (2.5 GBq) in most trials. When the MTSA and MTCA were within the dosage range (Peak), the escalation was performed up to the target dose of 5 to 5.5 GBq, followed by an adaptation around it. The MTSA and MTCA were found with precision in most trials and their relative errors remain good. There was a high proportion of trials stopped with the MTSA found (either based on CV or sample size criteria). As expected, sample size and cohort size were larger when escalation had to proceed up to the top doses (Safe, Late, Early) than when the MTSA was low (Toxic and Peak). In 80% of trial simulations, the total number of subjects was not larger than 28 in a maximum of seven cohorts. The number of overdoses was low in Cycle 1. It did not increase during Cycle 2, except when the second dose brought more toxicity (Early, Peak).

These results validate the use of the Bayesian model and the adaptive dose escalation plan for this study.

| Criterion   | A: Safe     | B: Late     | C: Early    | D: Toxic    | E: Peak     |
|---|-------------|-------------|-------------|-------------|-------------|
| % stops for reason: 2.5 GBq is toxic  |             |             |             | 23.4        |             |
| % stops for reason: 7.5 GBq is safe   | 58.9        | 46.4        | 5.90        |             |             |
| % stops for reason: CV (MTSA)≤30%   | 34.0        | 42.4        | 69.0        | 34.1        | 92.1        |
| % stops for reason: N≥6   | 7.10        | 11.2        | 25.1        | 42.5        | 7.90        |
| % trials (MTSA>top dose)  | 94.9        | 92.5        | 48.8        |             | 0.8         |
| % trials (MTSA <first dose)<="" td=""><td></td><td></td><td></td><td>38.1</td><td></td></first> |             |             |             | 38.1        |             |
| Median MTSA (GBq)   | 10.5        | 9.9         | 7.4         | 2.7         | 4.8         |
| Median MTSA Rel. Error (%) [a]  | 30.2%       | 29.2%       | 15.3%       | 34.2%       | 13.8%       |
| Mean CV (MTSA) (%)  | 28.8%       | 28.9%       | 28.9%       | 43.5%       | 21.5%       |
| Median MTCA (GBq)   | 17.4        | 14.7        | 9.5         | 4.2         | 6.2         |
| Median MTCA Rel. Error (%) [a]  | 41.9%       | 19.7%       | 16.9%       | 112%        | 25.2%       |
| Mean CV (MTCA) (%)  | 17.3%       | 26.9%       | 39.5%       | 54.2%       | 25.7%       |
| Mean (SD) number of overdosed subjects (Cycle   | 0.0 (0.0)   | 0.0 (0.0)   | 1.8 (2.1)   | 9.1 (3.6)   | 5.5 (1.5)   |
| 1)  |             |             |             |             |             |
| [Min, Max]  | [0.0, 0.0]  | [0.0, 0.0]  | [0.0, 5.0]  | [5.0, 19.0] | [0.0, 14.0] |
| Mean (SD) number of overdosed subjects (Cycle   | 0.0 (0.0)   | 1.7 (1.5)   | 6.3 (2.9)   | 3.6 (1.9)   | 7.8 (1.9)   |
| 2)  |             |             |             |             |             |
| [Min, Max]  | [0.0, 0.0]  | [0.0, 3.0]  | [0.0, 12.0] | [0.0, 9.0]  | [0.0, 15.0] |
| Mean (SD) number of subjects (Cycle 1)  | 17.5 (3.0)  | 17.7 (3.3)  | 18.7 (4.3)  | 9.1 (3.6)   | 15.9 (2.5)  |
| [Min, Max]  | [5.0, 25.0] | [5.0, 28.0] | [5.0, 31.0] | [5.0, 19.0] | [5.0, 28.0] |
| Mean (SD) number of subjects (Cycle 2)  | 15.7 (2.5)  | 15.7 (2.7)  | 14.3 (3.3)  | 3.6 (1.9)   | 10.8 (2.0)  |
| [Min, Max]  | [2.0, 21.0] | [2.0, 21.0] | [2.0, 21.0] | [0.0, 9.0]  | [2.0, 18.0] |
| Mean (SD) number of cohorts   | 5.3 (0.8)   | 5.3 (0.9)   | 5.1 (1.1)   | 2.0 (0.9)   | 4.4 (0.6)   |
| [Min, Max]  | [1.0, 7.0]  | [1.0, 7.0]  | [1.0, 7.0]  | [1.0, 4.0]  | [1.0, 7.0]  |

| Table 13         Main Operational C | Characteristics for the Five | <b>Trial Simulation Scenarios</b> |
|-------------------------------------|------------------------------|-----------------------------------|
|-------------------------------------|------------------------------|-----------------------------------|

CV=coefficient of variation; Max=maximum; Min=minimum; MTSA=maximum tolerated single activity. a Rel. Error=|MTD<sub>true</sub>-MTD<sub>median</sub>|/MTD<sub>true</sub>









Figure 7 Boxplot of Total Number of Subjects During Cycle 2 for the Five Simulation Scenarios





Figure 8 Boxplot of Total Number of Cohorts for the Five Simulation Scenarios

## **19.6** Attachment 6 – RECIST to PERCIST

| TABLE 7. Comparison of                  | of EORTC and PERCIST 1.0 (36)   |   |
|---|---|---|
| Characteristic                          | EORTC   | PERCIST 1.0   |
| Measurability of<br>lesions at baseline | <ol> <li>Tumor regions defined on pretreatment scan should be drawn on<br/>region of high <sup>18</sup>F-FDG uptake representing viable tumor. Whole<br/>tumor uptake should also be recorded.</li> <li>Same ROI volumes should be sampled on subsequent scans and<br/>positioned as close to original tumor volume as possible. Coregistration<br/>method should be recorded.</li> <li>Uptake measurements should be made for mean and maximal tumor<br/>ROI counts per pixel per second calibrated as MBq/L.</li> <li>Alterations in extent of <sup>18</sup>F-FDG uptake should be documented, i.e.,<br/>increase in orthogonal tumor dimensions including longest tumor<br/>dimension.</li> <li>Partial volume may affect measurement of <sup>18</sup>F-FDG uptake. Tumor<br/>size from anatomic imaging in relation to PET scanner resolution should<br/>be documented where possible.</li> </ol> | <ol> <li>Measurable target lesion is hottest single tumor lesion SUL of<br/>"maximal 1.2-cm diameter volume ROI in tumor" (SUL peak).<br/>SUL peak is at least 1.5-fold greater than liver SUL mean + 2<br/>SDs (in 3-cm spherical ROI in normal right lobe of liver). If liver is<br/>abnormal, primary tumor should have uptake &gt; 2.0 × SUL<br/>mean of blood pool in 1-cm-diameter ROI in descending<br/>thoracic aorta extended over 2-cm z-axis.</li> <li>Tumor with maximal SUL peak is assessed after treatment.<br/>Although typically this is in same region of tumor as that<br/>with highest SUL peak at baseline, it need not be.</li> <li>Uptake measurements should be made for peak and<br/>maximal single-voxel tumor SUL. Other SUV metrics, including<br/>SUL mean at 50% or 70% of SUV peak, can be collected as<br/>exploratory data; TLG can be collected ideally on basis of voxels<br/>more intense than 2 SDs above liver mean SUL (see below).</li> <li>These parameters can be recorded as exploratory data on<br/>up to 5 measurable target lesions, typically the 5 hottest lesions,<br/>which are typically the largest, and no more than 2 per organ.<br/>Tumor size of these lesions can be determined per RECIST 1.1.</li> </ol> |
| Normalization of uptake                 | Scanners should provide reproducible data. Reporting would need to be<br>accompanied by adequate and disclosed reproducibility measurements<br>from each center. An empiric 25% was found to be a useful cutoff point,<br>but reproducibility analysis is needed to determine appropriate cutoffs for<br>statistical significance.  | Normal liver SUL must be within 20% (and <0.3 SUL mean units) for baseline and follow-up study to be assessable. If liver is abnormal, blood-pool SUL must be within 20% (and <0.3 SUL mean units) for baseline and follow-up study to be assessable. Uptake time of baseline study and follow-up study 2 must be within 15 min of each other to be assessable. Typically, these are at mean of 60 min after injection but no less than 50 min after injection. Same scanner model at same site, injected dose, acquisition protocol (2- vs. 3-dimensional), and software for reconstruction, should be used. Scanners should provide reproducible data and be properly calibrated.   |
| Objective response                      | CMR: complete resolution of <sup>18</sup> F-FDG uptake within tumor volume so that it<br>was indistinguishable from surrounding normal tissue.  | CMR: complete resolution of <sup>18</sup> F-FDG uptake within measurable<br>target lesion so that it is less than mean liver activity and<br>indistinguishable from surrounding background blood-pool<br>levels. Disappearance of all other lesions to background blood-<br>pool levels. Percentage decline in SUL should be recorded from<br>measurable region, as well as (ideally) time in weeks after treatment<br>was begun (i.e., CMR –90, 4). No new <sup>18</sup> F-FDG-avid lesions in<br>pattern typical of cancer. If progression by RECIST, must verify<br>with follow-up.  |

| TABLE 7. continu | ied  |  |
|------------------|--|--|
| Characteristic   | EORTC  | PERCIST 1.0  |
|                  | PMR: reduction of minimum of 15% ± 25% in tumor <sup>19</sup> F-FDG SUV after 1<br>cycle of chemotherapy, and > 25% after more than 1 treatment cycle;<br>reduction in extent of tumor <sup>18</sup> F-FDG uptake is not a requirement for PMR.                                | PMR: reduction of minimum of 30% in target measurable tumor<br><sup>16</sup> F-FOG SUL peak. Absolute drop in SUL must be at least 0.8<br>SUL units, as well. Measurement is commonly in same lesion<br>as baseline but can be another lesion if that lesion was<br>previously present and is the most active lesion after treatment.<br>ROI does not have to be in precisely same area as baseline scan,<br>though typically it is. No increase, >30% in SUL or size of target or<br>nontarget lesions (i.e., no PD by RECIST or IWC) (if PD anatomically,<br>must verify with follow-up). Reduction in extent of tumor <sup>16</sup> F-FDG<br>uptake is not requirement for PMR. Percentage decline in SUL<br>should be recorded, as well as (ideally) time in weeks after<br>treatment was becaus (i.e. PMR = 40.3). No new lesions.  |
|                  | SMD: increase in tumor <sup>18</sup> F-FDG SUV < 25% or decrease of <15% and no<br>visible increase in extent of <sup>18</sup> F-FDG tumor uptake (20% in longest<br>dimension).   | SMD: not CMR, PMR, or PMD. SUL peak in metabolic target lesion<br>should be recorded, as well as (ideally) time from start of most<br>recent therapy, in weeks (i.e., SMD – 15, 7).  |
|                  | PMD: Increase in <sup>18</sup> F-FDG tumor SUV of >25% within tumor region defined<br>on baseline scan; visible increase in extent of <sup>18</sup> F-FDG tumor uptake<br>(20% in longest dimension) or appearance of new <sup>18</sup> F-FDG uptake in<br>metastatic lesions. | PMD: >30% increase in <sup>16</sup> F-FDG SUL peak, with >0.8 SUL unit<br>increase in turnor SUV peak from baseline scan in pattern typical<br>of turnor and not of infection/treatment effect. OR: Visible increase<br>in extent of <sup>18</sup> F-FDG turnor uptake (75% in TLG volume with no<br>decline in SUL. OR: New <sup>18</sup> F-FDG-avid lesions that are typical of<br>cancer and not related to treatment effect or infection. PMD<br>other than new visceral lesions should be confirmed on follow-up<br>study within 1 mo unless PMD also is clearly associated with<br>progressive disease by RECIST 1.1. PMD should be reported to<br>include percentage change in SUV peak, (ideally, time after<br>treatment, in weeks) and whether new lesions are present/absent<br>and their number (i.e., PMD, +35, 4, new: 5). Because SUL is<br>continuous variable, dividing response criteria into limited<br>number of somewhat arbitrary response categories loses much<br>data. For this reason, PERCIST preserves percentage declines in<br>SUV peak in each reported category. Because rapidity with which<br>scan normalizes is important (faster appears better), PERCIST asks<br>for time from start of treatment as part of reporting. For example,<br>CMR 90, 1, is probably superior to CMR 90, 10, especially if latter<br>patient were SMD 20, 1. More than one measurement of PET<br>response may be needed at differing times, and it may be<br>treatment type-dependent. PERCIST 1.0 evaluates SUL peak of only<br>hottest turnor. This is possible limitation of approach, but lesions<br>and their responses are highly correlated in general. Additional<br>data are required to determine how many lesions should be<br>assessed over 1.4 suggested option is to include the 5 hottest<br>lesions, or the 5 observed on RECIST 1.1 that are most measurable.<br>Percentage change in SUL can be reported for single lesion with<br>largest increase in uptake or smallest decline in uptake. Additional<br>studies will be needed to define how many lesions are optimal for<br>assessment. |

Study number D-FR-01087-001 - Protocol v 8.0 Date: 12 June 2020

| TABLE 7. continued   |  |   |
|----------------------|--|---|
| Characteristic       | EORTC  | PERCIST 1.0   |
|                      | Nonmeasurable disease: CR, disappearance of all known disease,<br>confirmed at ≥4 wk; PR, estimated decrease of ≥50%, confirmed<br>at 4 wk; PD, estimated increase of ≥25% in existent lesions; NC,<br>neither PR nor PD criteria met. | Nontarget lesions: CMR, disappearance of all <sup>18</sup> F-FDG-avid lesions:<br>PMD, unequivocal progression of <sup>18</sup> F-FDG-avid nontarget lesions<br>or appearance of new <sup>18</sup> F-FDG-avid lesions typical of cancer;<br>non-PMD: persistence of one or more nontarget lesions or tumor<br>markers above normal limits.      |
| Overall response     |  | <ol> <li>Best response recorded in measurable disease from treatment<br/>start to disease progression or recurrence.</li> <li>Non-PMD in measurable or nonmeasurable nontarget lesions will<br/>reduce CR in target lesion to overall PMR.</li> <li>Non-PMD in nontarget lesions will not reduce PR in target lesions.</li> </ol>               |
| Duration of response |  | <ol> <li>Overall CMR: from date CMR criteria are first met; to date<br/>recurrent disease is first noted.</li> <li>Overall response: from date CMR or PMR criteria are first met<br/>(whichever status came first); to date recurrent disease is first noted.</li> <li>SMD: from date of treatment start to date PMD is first noted.</li> </ol> |

TLG = total lesion glycolysis; CMR = complete metabolic response; PMR = partial metabolic response; PD = progressive disease; SMD = stable metabolic disease; PMR = partial remission; NC = no change. For PERCIST: Single-voxel SUL is commonly used but has been reported to be less reproducible than SUL peak, especially with very small single-voxel values. It is suggested, but not required, that lesions assessed on PERCIST be larger than the 1.5-cm-diameter volume ROI used to minimize partial-volume effects. Percentage changes are proposed to deal with SUL peak, especially with very small single-voxel values. It is suggested to be asserved as exploratory approach, it is suggested that sum of SULs of baselines for study. After treatment, sum of same 5 lesions should be used. Percentage change in SUL is based on change in SUL in individual lesions and averaging them. This may produce different result. We believe summed SUL approach will be less prone to minor errors in measurements. measurements

Tor total lesion glycolysis: Exploratory analysis can include either all foci of turnor with maximal SUL > 2 SDs above normal liver, 5 lesions with highest SUL, or lesion with highest SUL, it is suggested that threshold approach, typically at 2 SDs above normal liver SUL, be used to generate lower bounds of ROI (3 SDs could be used for very active turnors). We believe this approach will be less variable than methods based on maximal SUL with percentage of maximal cutoff. Criteria for progression include 75% growth in TLG for SUL and are conservatively placed at 75% increase. Because 20% increase in EORTC linear size scales to 73% volume increase, the figures are comparable. Progression is judged from best response if being assessed after first scan was performed. For response by TLG, we propose 45% reduction as useful starting point, but more data are needed to make firm recommendations. If TLG is determined, explicit methodologic details should be provided. It should not be a primary metric, but a secondary endpoint at this time.

## 19.7 Attachment 7 – Protocol Amendment #7 Summary

## **PROTOCOL VERSION 8.0**

| STUDY NUMBER:                               | D-FR-01087-001   |
|---|--|
| PROTOCOL TITLE:                             | An international multicentre, open-label first in human phase I/II study to evaluate the safety, tolerability, biodistribution and antitumour activity of <sup>177</sup> Lu-3BP-227 for the treatment of subjects with solid tumours expressing neurotensin receptor 1 |
| AMENDED PROTOCOL VERSION<br>NUMBER AND DATE | Version 8.0: 12 June 2020  |

## THE FOLLOWING AMENDMENTS ARE PROPOSED:

|                          | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>  |
|--------------------------|---|--|--|
| Section                  | WAS   | IS   |  |
| Cover Page,<br>Agreement | Protocol Version 7.0, dated 20 June 2019  | Protocol Version 8.0, dated 12 June 2020   | Administrative update  |
| Cover Page               | <b>Serious Adverse Event Reporting</b><br>Fax: PPD  | Serious Adverse Event Reporting<br>Fax Number: PPD<br>USA and Latin America Fax Number: PPD  | Administrative update  |
| Cover Page               | Sponsor Authorised Protocol Approver         Name. Title:       PPD   | Sponsor Authorised Protocol Approver<br>Name. Title: PP<br>D   | Administrative update  |
| Cover page               | Sponsor Medical Monitor<br>Name. Title: PPD   | Sponsor Medical Monitor<br>Name. Title: PPD  | Administrative update  |
| Agreement                | Sponsor<br>Name: PPD  | Sponsor<br>Name: PPD   | Administrative update  |
| Synopsis, 2.1            | <b>STUDY HYPOTHESIS</b><br><b>Phase I</b><br><sup>177</sup> Lu-3BP-227 is sufficiently well-tolerated to permit<br>clinical investigation in the phase II part. | <b>STUDY HYPOTHESIS</b><br><b>Phase I</b><br><sup>177</sup> Lu-3BP-227 is sufficiently well-tolerated to permit<br>clinical investigation in phase II. | Clarification  |
| Synopsis, 2.2            | STUDY HYPOTHESIS<br>Phase II  | STUDY HYPOTHESIS<br>Phase II   | FDA request to clarify the eligible<br>subject population: inclusion |

|   | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>  |
|---|--|--|--|
| Section   | WAS  | IS   |  |
|   | <sup>177</sup> Lu-3BP-227 yields higher objective response rates in<br>subjects who have NTSR1 expressing metastatic or<br>locally advanced cancers, based on RECIST version 1.1<br>by central review and as compared with the historical<br>ORR obtained by current standard-of-care treatment for<br>each tumour type.   | <sup>177</sup> Lu-3BP-227 yields higher objective response rates in<br>subjects who have NTSR1 expressing <b>cancers that are</b><br><b>unresectable</b> , locally advanced <b>or metastatic</b> , based<br>on RECIST version 1.1 by central review and as<br>compared with the historical ORR obtained by current<br>standard-of-care treatment for each tumour type.   | criterion should clearly state<br>nonresectable locally advanced<br>disease.   |
| Synopsis, 1.5, 1.7,<br>3.1.1, 3.1.2, 4.1,<br>4.2.1, 4.3 | subjects with <del>metastatic or</del> locally advanced  | subjects with <b>unresectable</b> , locally advanced <b>or metastatic</b>  | FDA request to clarify the eligible<br>subject population: inclusion<br>criterion should clearly state<br>nonresectable locally advanced<br>disease        |
| Synopsis, 3.2.1   | Secondary objectives<br><u>Phase I</u><br><br>b) To determine the radiation dosimetry of <sup>177</sup> Lu-<br>3BP-227 (organ exposure to radiation) <del>after each</del><br><del>administration</del> .  | Secondary objectives<br><u>Phase I</u><br><br>b) To determine the radiation dosimetry of <sup>177</sup> Lu-<br>3BP-227 (organ exposure to radiation).  | Clarification  |
| Synopsis, 3.3.1   | <ul> <li>Exploratory objectives <u>Phase I/II</u> <ul> <li>a) To explore the correlation between the tumour uptake of <sup>177</sup>Lu-3BP-227 and the NTSR1 expression on tumours.</li> <li>b) To explore renal and haematological safety by measuring urinary specific biomarkers and deoxyribonucleic acid double strand breaks (DNA DSB) in peripheral lymphocytes (at selected centres).</li> <li>c) To evaluate the tumour microenvironment, transcriptomics, and other markers of interest for the disease through assessment of tumour biopsies.</li> <li>d) To explore genomic alterations in circulating cell-free DNA (cfDNA) as compared to germline DNA.</li> </ul></li></ul> | <ul> <li>Exploratory objectives</li> <li>Phase I/II</li> <li>a) To explore the correlation between the tumour uptake of <sup>177</sup>Lu-3BP-227 and the NTSR1 expression on tumours.</li> <li>b) To explore renal safety by measuring urinary specific biomarkers.</li> <li>c) To evaluate the tumour microenvironment, transcriptomics, and other markers of interest for the disease through assessment of tumour biopsies.</li> <li>d) To explore genomic alterations in circulating cell-free DNA (cfDNA) and in germline DNA.</li> </ul> | Assessment of haematological<br>safety markers (DNA-DSB in<br>peripheral lymphocytes) removed<br>and clarification of the genomic<br>alteration assessment |
| Synopsis, 4.4.3   | <ul> <li>Exploratory endpoints</li> <li>Tumour microenvironment and other markers of interest (such as NTSR1 expression, Ki67, gene</li> </ul>   | <ul> <li>Exploratory endpoints</li> <li>Tumour microenvironment and other markers of interest (such as NTSR1 expression, Ki67, gene</li> </ul>   | Clarification  |

|                              | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>   |
|------------------------------|--|--|---|
| Section                      | WAS  | IS   |   |
|                              | expression and DNA-DSB) in tumour biopsies taken<br>at baseline, at EOCT visit or at disease progression,<br>whichever occurs earlier;   | expression and DNA <b>damage</b> ) in tumour biopsies<br>taken at baseline, at EOCT visit or at disease<br>progression, whichever occurs earlier;  |   |
| Synopsis, 4.4.3              | Exploratory endpoints<br>(c) Number of DNA DSB per cell in peripheral<br>lymphocytes and the correlation with absorbed<br>dose in the blood at baseline and , 4 and 72 to 96<br>hours after each administration (in selected centres<br>only).   | Exploratory endpoint deleted   | Assessment of haematological<br>safety markers (DNA-DSB in<br>peripheral lymphocytes) removed   |
| Synopsis, 4.2.1.1,<br>19.5.1 | <ul> <li>The MTCA is defined as the maximum tolerated cumulative activity that may be administered following fractionated i.v. administrations of at least 4 weeks apart, so that:</li> <li>No more than 33% of the subjects experience a dose limiting toxicity (DLT) during Cycle 1 or 2 and</li> <li>The cumulative radiation in each target organ does not exceed the acceptability limits</li> </ul>  | <ul> <li>The MTCA is defined as the maximum tolerated cumulative activity that may be administered following fractionated i.v. administrations of at least 4 weeks apart, so that:</li> <li>No more than 33% of the subjects experience a dose limiting toxicity (DLT) during Cycles 1 and/or 2 and</li> <li>The cumulative radiation in each target organ does</li> </ul>   | Clarification   |
|                              |  | not exceed the acceptability limits.   |   |
| Synopsis, 4.7.1.1            | The DLTs are defined DLT assessment period (from<br>the first administration of <sup>177</sup> Lu-3BP-227 to<br>EOCT/ED):<br>  | The DLTs are defined DLT assessment period (from<br>the first administration of <sup>177</sup> Lu-3BP-227 to<br>EOCT/ED):  | FDA request to revise the DLT<br>criteria to adequately describe the<br>grading as stated in the CTCAE<br>v5.0 dictionary:  |
|                              | <ul> <li>Grade 4 unonhoodynopenia nor-seven of more consecutive days;</li> <li>bleeding requiring transfusion with Grade 3 thrombocytopenia;</li> <li>any Grade 3 or higher laboratory abnormalities on haemoglobin requiring transfusion for &gt;4 weeks;</li> <li>any Grade 3 or higher laboratory abnormalities on aspartate amino transferase/alanine amino transferase (AST/ALT) and/or bilirubin, which do not resolve in ≤4 weeks;</li> <li>any Grade 3 or higher chronic kidney disease (estimated glomerular filtration rate (eGFR) &lt;30 mL/min/1.73 m<sup>2</sup>), which does not resolve within 1</li> </ul> | <ul> <li>Grade 3 of 4 thiombocytopenia (platelet count decreased) with clinically meaningful bleeding (i.e. requiring urgent hospitalisation or transfusion to manage the bleeding);</li> <li>Grade 4 thrombocytopenia for seven or more consecutive days;</li> <li>Any Grade 3 anaemia (Hb&lt;8.0 g/dL; transfusion indicated) or Grade 4 anaemia (life-threatening consequences; urgent intervention indicated);</li> <li>Any Grade 3 or higher laboratory abnormalities in aspartate aminotransferase/alanine aminotransferase (AST/ALT) with accompanying Grade 2 or higher bilirubin (Hy's law);</li> </ul> | <ul> <li>a clinically significant bleeding is a DLT, irrespective of the duration of thrombocytopenia. In addition, revised the DLT criterion so a Grade 3 thrombocytopenia with a clinically meaningful bleeding (i.e. requiring urgent hospitalisation or transfusion) is a DLT.</li> <li>Clarified the haemoglobin DLT criterion</li> <li>Revised the ALT/AST DLT criteria so a Grade 3 ALT/AST</li> </ul> |

|                   | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>   |
|-------------------|--|--|---|
| Section           | WAS  | IS   |   |
|                   | <ul> <li>week on nephroprotective treatment (e.g. forced diuresis);</li> <li>any Grade 3 or higher GI AE with the following specifications: Grade 4 (life threatening) nausea, vomiting, diarrhoea, constipation, dry mouth or Grade 3 nausea, vomiting, diarrhoea, constipation not resolved to Grade ≤2 within 48 hours despite optimal adequate medical management;</li> <li>any toxicity related to <sup>177</sup>Lu-3BP-227 resulting in a treatment delay of more than four weeks due to either delayed recovery to baseline or resolution of any AE of Grade ≤2 (exception of alopecia and lymphopenia).</li> </ul> | <ul> <li>Any Grade 3 or higher renal injury/toxicity (estimated glomerular filtration rate (eGFR) &lt;30 mL/min/1.73 m<sup>2</sup>);</li> <li>Any Grade 3 or higher GI AE, not resolved to Grade ≤2 within 48 hours despite optimal adequate medical management, with the following specifications:         <ul> <li>Grade 3 nausea, vomiting (inadequate oral caloric or fluid intake; tube feeding, total parenteral nutrition or hospitalisation indicated)</li> <li>Grade 3 diarrhoea (increase of ≥7 stools per day over baseline; hospitalisation indicated; severe increase in ostomy output compared to baseline; limiting self-care ADL) or Grade 4 diarrhoea (life-threatening consequences; urgent intervention indicated)</li> <li>Grade 3 constipation (obstipation with manual evacuation indicated);</li> </ul> </li> <li>Any toxicity related to <sup>177</sup>Lu-3BP-227 resulting in a treatment delay of more than four weeks due to either delayed recovery to baseline or resolution of any AE to Grade ≤2 (with the exception of alopecia and lymphopenia).</li> </ul> | <ul> <li>increase simultaneous with a Grade 2 increase in bilirubin is considered a DLT (Hy's law).</li> <li>Revised the DLT criteria so Grade 3 renal injury/toxicity is a DLT, irrespective of the toxicity being "chronic" or acute.</li> <li>Revised the DLT criteria to delete Grade 4 nausea and dry mouth (there is no Grade 4 nausea toxicity described in the CTACE v5.0 dictionary).</li> </ul> |
| Synopsis, 4.2.1.4 | <b>Study design following phase I dose escalation</b><br><b>results</b><br>Upon termination of phase I dose escalation and/or<br>determination-upon reaching MTCA and stated jointly<br>by the safety review committee (SRC) and the sponsor,<br>and in consideration of the accumulated subject data,<br>cohorts of subjects will be studied to further<br>characterise safety and efficacy of <sup>177</sup> Lu-3BP-227.   | <b>Study design following phase I dose escalation</b><br><b>results</b><br>Upon <b>completion</b> of <b>the</b> phase I dose escalation or upon<br>reaching <b>the</b> MTCA and <b>confirmed</b> jointly by the<br>safety review committee (SRC) and the sponsor, and in<br>consideration of the accumulated subject data, cohorts<br>of subjects will be studied to further characterise <b>the</b><br>safety and efficacy of <sup>177</sup> Lu-3BP-227.  | Clarification   |

|                 | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>  |
|-----------------|--|--|--|
| Section         | WAS  | IS   |  |
| Synopsis, 5.1.1 | <ul> <li>Inclusion criteria:</li> <li>(3a) Histologically or cytologically confirmed<br/>metastatic or locally advanced disease and no<br/>compelling treatment option as per standard-of-<br/>care and documented decision by a<br/>multidisciplinary oncology board including a<br/>specialist of the concerned pathology.</li> </ul>  | <ul> <li>Inclusion criteria:</li> <li>(3) Histologically or cytologically confirmed unresectable, locally advanced or metastatic disease and has received prior lines of standard-of-care chemotherapy/treatment and has no further suitable treatment options and a documented decision by a multidisciplinary oncology board including a specialist of the concerned pathology.</li> </ul> | FDA request to clarify the<br>inclusion criterion to clearly state<br>that no further suitable treatment<br>options are available for subjects<br>eligible for the study                               |
| Synopsis, 5.1.1 | <ul> <li>Inclusion criteria:</li> <li>(5) Tumour tissue expressing NTSR1, as determined<br/>by uptake of <sup>177</sup>Lu-3BP-227 (screening<br/>formulation) in tumour lesions being judged by the<br/>investigator to more avid than in the non-tumoral<br/>surrounding tissue based on whole body scan<br/>(planar scintigraphy); single photon emission<br/>computed tomography (SPECT)/ computed<br/>tomography (CT) can be performed as per<br/>investigator's judgement.</li> </ul>   | Inclusion criteria:(5) Tumours showing:(a) uptake of 177Lu-3BP-227 (screening formulation) in known primary or metastatic sites as judged by the investigator to be greater than background; or(b) uptake of 111In-3BP-227 in known primary or metastatic sites (for subjects who participated in Study D-FR-01087-002) as judged by the investigator to be greater than background.         | To allow inclusion of subjects<br>screened for NTSR1+ in the<br>D-FR-01087-002 phase I study<br>( <sup>111</sup> In-IPN01087) in the<br>D-FR-01087-001 phase I<br>( <sup>177</sup> Lu-IPN01087) study. |
| Synopsis, 5.1.1 | <ul> <li>Inclusion criteria:         <ul> <li>(12a) Female subjects must not be pregnant or lactating at study entry and during the course of the study and must not become pregnant for at least 6 months following the last study treatment. Women of childbearing potential must agree to use a highly effective method of contraception (see note below). Because CT seanning is part of the long term follow up (every 12 ±2 weeks), female subjects should not become pregnant until the end of study assessments.</li> <li>(13a)</li> </ul> </li> </ul> | <ul> <li>Inclusion criteria:</li> <li>(12) Female subjects must not be pregnant or lactating at study entry and during the course of the study and must not become pregnant for at least 6 months following the last study treatment. Women of childbearing potential must agree to use a highly effective method of contraception (see note below).</li> <li>(13)</li> </ul>                | Clarification that inclusion criteria<br>are related to study treatment<br>procedures  |
| Synopsis, 5.2   | Exclusion criteria:  | Exclusion criteria:  | Clarification  |
|                 | (1) Prior treatment received   | (1) Prior treatment received   |  |

|                 | Version 7.0, 20 June 2019                                    | Version 8.0, 12 June 2020                                       | <b>RATIONALE FOR CHANGE</b>                                    |
|-----------------|--|---|--|
| Section         | WAS  | IS  |  |
|                 | (a) Any antitumour treatment since last                      | (a) Any antitumour treatment since last                         |  |
|                 | documentation of disease progression                         | documented disease progression                                  |  |
| Synopsis, 4.4.3 | Exploratory endpoints  | Exploratory endpoints   | Clarification  |
|                 |  |   |  |
|                 | c) Genomic profiling in circulating cfDNA as                 | c) Genomic profiling in circulating cfDNA and in                |  |
| 1.7             | compared to germline DNA.                                    | germline DNA.   |  |
| 1.5             | Rationale for the Study                                      | Rationale for the Study   | FDA request to clarify the                                     |
|                 | <br>1771 2DD 227   | <br>1771 2DD 22714 h :  | inclusion criterion to clearly state                           |
|                 | who expansion available effective abarratherapy and          | "Lu-3BP-22/ could be investigated in those subjects             | that no further suitable treatment                             |
|                 | their tumour tissue overexpresses NTSP1 knowing that         | chemotherapy/treatment and have no further                      | eligible for the study   |
|                 | the percentage of CRC tumours expressing NTSR1 is            | suitable treatment ontions and their tumour tissue              | engible for the study  |
|                 | very high, with high degree of expression (see Section       | overexpresses NTSR1, knowing that the percentage of             |  |
|                 | 1.7).  | CRC tumours expressing NTSR1 is very high, with                 |  |
|                 |  | high degree of expression (see Section 1.7).                    |  |
| 1.7             | In this FIH study, the subject population enrolled to        | In this FIH study, the subject population enrolled to           | To allow inclusion of subjects                                 |
|                 | receive the IMP will be restricted to PDAC, CRC, GC,         | receive the IMP will be restricted to PDAC, CRC, GC,            | screened for NTSR1+ in the                                     |
|                 | squamous-cell carcinoma of head and neck (SCCHN)             | GIST, squamous-cell carcinoma of head and neck                  | D-FR-01087-002 phase I study                                   |
|                 | and ES, which are locally advanced and expressing            | (SCCHN) and ES, which are unresectable, locally                 | ( <sup>111</sup> In-IPN01087) in the                           |
|                 | NTSR1.   | advanced <b>or metastatic</b> and expressing NTSR1.             | D-FR-01087-001 phase I<br>( <sup>177</sup> Lu-IPN01087) study. |
|                 | Tumours/metastases expressing NTSR1 will be                  | Tumours/metastases expressing NTSR1 will be                     |  |
|                 | identified and documented through the lesion uptake of       | identified and documented through the lesion uptake of          |  |
|                 | <sup>177</sup> Lu-3BP-227 (at low dose) during the screening | <sup>177</sup> Lu-3BP-227 (at low dose) during the screening    |  |
|                 | period.  | period. Subjects who participated in the imaging                |  |
|                 |  | study, Study D-FR-01087-002, with <sup>111</sup> In-3BP-227     |  |
|                 |  | (also called <sup>111</sup> In-IPN01087) and who have an uptake |  |
|                 |  | of "In-3BP-227 in tumour lesions that is more avid              |  |
|                 |  | than in the nontumoural surrounding tissue based                |  |
|                 |  | on whole body imaging (planar scintigraphy), as                 |  |
|                 |  | for enrolment in this study provided that they fulfil           |  |
|                 |  | all other inclusion criteria and do not meet any of the         |  |
|                 |  | exclusion criteria and were imaged with                         |  |
|                 |  | <sup>111</sup> In-3BP-227 before the first treatment            |  |
|                 |  | administration (therapeutic dose) of <sup>177</sup> Lu-3BP-227. |  |
|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>  |
|---------|--|---|--|
| Section | WAS  | IS  |  |
| 1.8     | The participating subjects will be closely monitored<br>during the study and during follow-up for 2 years after<br>the last administration.  | The participating subjects will be closely monitored<br>during the study and during <b>the long-term</b> follow-up<br><b>period until lost to follow-up, withdrawal of consent,</b><br><b>death or a maximum of 5</b> years, whichever occurs<br>first.   | To extend the long-term follow-up<br>period from 2 years to a maximum<br>of 5 years based on peptide<br>receptor radionuclide therapy<br>experience on the risks of<br>developing secondary<br>malignancies, and the median time<br>to occurrence of such<br>malignancies. This will ensure<br>continued benefit-risk assessment<br>of <sup>177</sup> Lu-3BP-227 for a longer<br>period of time after the end of<br>treatment.   |
| 4.2.1   | The study screening period includes a screening<br>administration of <sup>177</sup> Lu-3BP-227 to assess the tumour<br>uptake of each subject by whole body scan (planar<br>scintigraphy); single photon emission computed<br>tomography (SPECT)/CT <del>can be done</del> at the<br>investigator's discretion <del>to further evaluate the tumour<br/>uptake; o</del> nly subjects with a tumour uptake higher than<br>nontumoural surrounding tissue (based on<br>investigator's decision) will be deemed eligible for<br>treatment. | The study screening period includes a screening administration of <sup>177</sup> Lu-3BP-227 to assess the tumour uptake of each subject by whole body scan (planar scintigraphy, <b>1 or 2 timepoint(s) at the investigator's discretion) and optional</b> single photon emission computed tomography (SPECT)/CT <b>scans (up to 2</b> at the investigator's discretion). Only subjects with a tumour uptake higher than nontumoural surrounding tissue (based on investigator's decision) will be deemed eligible for treatment. | After reviewing subjects'<br>SPECT/CT images at different<br>timepoints following the<br>therapeutic dose of<br><sup>177</sup> Lu-IPN01087, it is concluded<br>that the quality of SPECT image<br>with respect to the background<br>noise has substantially improved<br>at the 48-hour and later<br>timepoints. Considering the<br>significant number of screen<br>failures due to lack of radioactive<br>uptake, which is contrary to the<br>information available in the<br>literature on the expression of<br>NTSR1 receptor, the image<br>acquisition timepoint has been<br>moved to 48 hours after the<br>screening dose. |
| 4.2.1   | <br>The screening period will last 3 weeks but can be<br>extended by up to 2 weeks if this is required for<br>logistical reasons.  | The screening period will last 3 weeks but can be<br>extended by up to 2 weeks if this is required for<br>logistical reasons. Subjects who participated in the<br>imaging study, Study D-FR-01087-002, with   | To allow inclusion of subjects<br>screened for NTSR1+ in the<br>D-FR-01087-002 phase I study<br>( <sup>111</sup> In-IPN01087) in the   |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>          |
|---------|---|--|--------------------------------------|
| Section | WAS   | IS   |                                      |
|         |   | <sup>111</sup> In-3BP-227 (also called <sup>111</sup> In-IPN01087) and who | D-FR-01087-001 phase I               |
|         |   | have an uptake of <sup>111</sup> In-3BP-227 in tumour lesions              | ( <sup>177</sup> Lu-IPN01087) study. |
|         |   | that is more avid than in the nontumoural                                  |                                      |
|         |   | surrounding tissue based on whole body imaging                             |                                      |
|         |   | (planar scintigraphy), as judged by the investigator,                      |                                      |
|         |   | can also be considered for enrolment in this study,                        |                                      |
|         |   | provided that they fulfil all other inclusion criteria                     |                                      |
|         |   | and do not meet any of the exclusion criteria and                          |                                      |
|         |   | were imaged with <sup>111</sup> In-3BP-227 before the first                |                                      |
|         |   | treatment administration (therapeutic dose) of                             |                                      |
|         |   | <sup>177</sup> Lu-3BP-227.   |                                      |
| 4.2.1   | A long-term follow-up lasting 24 months will start after                                | A long-term follow-up <b>period</b> will start after the EOCT,             | Duration of long-term follow-up      |
|         | the end of last cycle (either EOCT, early   | early discontinuation (ED) or end of additional cycles                     | updated                              |
|         | discontinuation (ED) visit or end of additional cycles                                  | (EOAC) visit and subjects will be followed up every                        |                                      |
|         | (EOAC)) and subjects will be followed up every  | 3 months $(\pm 2 \text{ weeks})$ until <b>lost to follow-up</b> ,          |                                      |
|         | 3 months ( $\pm 2$ weeks) until $\frac{24 \text{ months}}{24 \text{ months}}$ or death, | withdrawal of consent, death or a maximum of                               |                                      |
|         | whichever occurs first.   | 5 years, whichever occurs first.   |                                      |
| 4.2.1   | Figure 1  | Figure 1   | Figure updated to show new           |
|         | Long-term follow up   | Long-term follow up  | long-term follow-up period           |
|         | <del>24 months</del>  | 5 years  |                                      |
| 4.2.1   | Figure 1  | Figure 1   | FDA request to clarify the eligible  |
|         | New footnote added  | Footnote added:  | subject population, i.e.             |
|         |   | Note: Population screened includes subjects with                           | nonresectable locally advanced or    |
|         |   | unresectable disease.  | metastatic disease.                  |
| 4.2.1.2 | The size of the cohorts will be three to five treated                                   | The size of the cohorts will be three to five treated                      | Clarification of the maximum         |
|         | subjects, to secure a minimum of three evaluable  | subjects, to secure a minimum of three evaluable                           | number of subjects per cohort.       |
|         | subjects per cohort who completed 2 cycles of   | subjects per cohort who completed 2 cycles of                              |                                      |
|         | treatment. Once five subjects receive one   | treatment. A cohort will be considered as completed                        |                                      |
|         | administration of <sup>+++</sup> Lu 3BP 227 in a cohort, enrolment                      | once three subjects of the cohort complete Cycle 2 or                      |                                      |
|         | will be stopped in that cohort. A cohort will be  | early discontinue during Cycle 2 (except for cohort 1,                     |                                      |
|         | considered as completed once three subjects of the                                      | see Proceeding to the Next Cohort described in Section                     |                                      |
|         | cohort complete Cycle 2 or early discontinue during                                     | 4.2.1.4). If a subject replacement is needed to                            |                                      |
|         | Cycle 2 (except for cohort 1, see Proceeding to the Next                                | complete the cohort enrolment, see Section 4.7.1.3                         |                                      |
|         | Cohort described in Section 4.2.1.4).   | for replacement rules.   |                                      |
| 4.2.1.3 |   |  | Clarification                        |

|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>   |
|---------|--|---|---|
| Section | WAS  | IS  |   |
|         | In the first step, in order to minimise the number of subjects receiving subtherapeutic activity, the inter-<br>cohort radioactivity escalation will be performed by increment of 3 GBq (1.5 GBq per administration), up to the first study drug related AE (Grade 2 or higher, except hair loss) or until a DLT occurs.   | In the first step, in order to minimise the number of<br>subjects receiving subtherapeutic activity, the inter-<br>cohort radioactivity escalation will be performed by<br>increment of 3 GBq (1.5 GBq per administration), up to<br>the first study drug related <b>treatment-emergent</b> AE<br><b>(TEAE)</b> (Grade 2 or higher, except hair loss) or until a<br>DLT occurs.   |   |
| 4.5     | For the phase I, the maximum duration of subject<br>participation in the core trial is 21 weeks. However, if a<br>clinical benefit is observed, subjects may receive up to<br>four additional cycles after the EOCT, provided they<br>have an acceptable tolerability profile, the organ dose<br>limits are not exceeded and according to investigator's<br>judgement, subject's discretion and must be discussed<br>with and agreed by the sponsor. | For the phase I, the maximum duration of subject<br>participation in the core trial is 21 weeks. However, if,<br><b>according to the investigator</b> , a <b>subject has</b> clinical<br>benefit, <b>the</b> subject may receive up to four additional<br>cycles after the EOCT, provided they have an<br>acceptable tolerability profile <b>and</b> the organ dose limits<br>are not exceeded. <b>This is at the</b> subject's discretion and<br>must be discussed and agreed <b>with</b> the sponsor. | Clarification   |
| 4.5     | In all cases, a long-term follow-up lasting 24 months will start after the last cycle is administered (after EOCT, ED visit or EOAC) and subjects will be followed up every three months ( $\pm 2$ weeks).   | In all cases, a long-term follow-up <b>period</b> will start after<br>the EOCT, ED or EOAC <b>visit</b> and subjects will be<br>followed up every three months (±2 weeks) <b>until lost</b><br><b>to follow-up, withdrawal of consent, death or a</b><br><b>maximum of 5 years, whichever occurs first</b> .  | Duration of long-term follow-up<br>updated  |
| 4.7.1   | <ul> <li>Individual Discontinuation Rules If any of the following occur, no further treatment will be administered: <ul> <li>subject withdraws their consent to further treatment;</li> <li></li> </ul> </li> </ul>  | <ul> <li>Individual Discontinuation Rules</li> <li></li> <li>If any of the following occur, no further treatment will be administered:</li> <li>life-threatening toxicities outside of the DLT reporting period</li> <li>subject withdraws their consent to further treatment;</li> <li></li> </ul>   | FDA request to revise the discontinuation rules so if there are life-threatening toxicities outside of the DLT period, treatment is discontinued. |
| 5.2     | <ul> <li>(1) Prior treatment received:</li> <li></li> <li>(b) Any chemotherapy within 3 weeks or nitrosourea prior to first treatment IMP administration</li> </ul>  | <ul> <li>(1) Prior treatment received:</li> <li></li> <li>(b) Any chemotherapy within 3 weeks or nitrosourea within 6 weeks prior to first treatment IMP administration</li> </ul>  | Revision of inconsistent exclusion<br>criteria between synopsis and core<br>text.   |
| 6.2     | New text added   | If the COVID-19 pandemic prevents subjects from coming to the site, subjects can have their study visit   | Clarification following COVID-19 pandemic.  |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>  |
|---------|---|---|--|
| Section | WAS   | IS  |  |
|         |   | assessments performed remotely as judged<br>appropriate by the investigator. This must be<br>discussed with the sponsor before being<br>implemented. In such a case, the investigator will<br>perform a telemedicine visit and will make every<br>effort, where applicable, to contact the subject's<br>general practitioner or specialist physician to ensure<br>all important medical information and safety<br>event(s) occurring since the last visit are collected.<br>Guidance on how to collect protocol-planned<br>assessments will be provided to the investigator in a<br>separate document. This document will be filed in<br>the electronic trial master file. Independent ethics<br>committees (IECs)/institutional review boards<br>(IRBs) will be notified of the changes as applicable<br>locally. Of note, as the adapted visit deviates from<br>the regular protocol plan, the changes will be<br>recorded as protocol deviations related to<br>COVID-19. |  |
| 6.2.2.2 | If according to the investigator, a subject has clinical<br>benefit, good tolerability <del>of the compound and if</del> the<br>organ dose limits are not exceeded, he/she can receive<br>up to four additional cycles after EOCT. For these<br>subjects, an EOAC visit will be done 6 weeks after last<br>dose administration. | If, according to the investigator, a subject has clinical<br>benefit, the subject may receive up to four additional<br>cycles after the EOCT, provided they have an<br>acceptable tolerability profile and the organ dose<br>limits are not exceeded. This is at the subject's<br>discretion and must be discussed and agreed upon<br>with the sponsor. For these subjects, an EOAC visit<br>will be done 6 weeks after last dose administration.   | Clarification  |
| 6.2.3   | For both phase I and phase II, a long-term follow-up<br>period will start after ED, EOCT or after EOAC and<br>subjects will be followed up every 3 months (±2 weeks)<br>until <del>24 months or</del> death, whichever occurs first.  | For both phase I and phase II, a long-term follow-up period will start after the ED, EOCT or EOAC visit and subjects will be followed up every 3 months (±2 weeks) until lost to follow-up, withdrawal of consent, death or a maximum of 5 years, whichever occurs first.   | Duration of long-term follow-up<br>updated   |
| 6.2.3   | <br>New text added  | <br><u>Five Years After the EOCT, ED or EOAC Visit</u>  | Specific assessments to be<br>performed during the long-term<br>safety follow-up detailed. |

|         | Version 7.0, 20 June 2019 | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>  |
|---------|---------------------------|--|--|
| Section | WAS                       | IS   |  |
|         |                           | In the long-term follow-up period, subjects will be<br>followed up every 3 months (±2 weeks) until lost to<br>follow-up, withdrawal of consent, death or a<br>maximum of 5 years, whichever occurs first. The<br>following safety assessments will be performed by<br>the site: these assessments can be performed on-site<br>or by remote visit:<br>• Survival status   |  |
|         |                           | <ul> <li>Haematology and biochemistry tests<br/>(haemoglobin, platelet count, white blood cell<br/>count (absolute counts), AST, ALT, total<br/>bilirubin (with conjugated bilirubin if total<br/>bilirubin is abnormal i.e. outside the<br/>laboratory normal range), creatinine and<br/>urea) performed until the start of new<br/>antitumour treatment or up to 2 years after the<br/>EOCT, ED or EOAC visit, whichever comes<br/>first. Laboratory tests can be performed<br/>locally; results should be collected from the<br/>local laboratory.</li> </ul> |  |
|         |                           | <ul> <li>All AEs and serious adverse events (SAEs) are<br/>to be reported up to 6 months after the EOCT,<br/>ED or EOAC visit or until new antitumour<br/>treatment starts, whichever comes first.<br/>Thereafter, only AEs and SAEs related to the<br/>IMP or study procedure should be reported for<br/>the remainder of the safety follow up period.</li> <li>Subsequent antitumour treatment</li> </ul>  |  |
| 6.3.1   | <br>New text added        | An concerce data will be recorded in the eCKF.<br>Any further antitumour treatment for the disease<br>under study that may have been taken during the<br>long-term follow-up period should be reported on<br>the prior and concomitant medications page of the<br>eCRF.  | Details of requirements during the long-term safety follow-up added. |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>   |
|---------|---|---|---|
| Section | WAS   | IS  |   |
| 7.1.1   | One vial can contain up to 17.5 GBq of <sup>177</sup> Lu-3BP-227, this amount takes into account decay to ensure the administration of the highest planned activity of 7.5 GBq of <sup>177</sup> Lu-3BP-227 from the vial.  | One vial can contain up to 17.5 GBq of <sup>177</sup> Lu-3BP-227<br><b>at the end of manufacturing</b> , this amount takes into<br>account decay to ensure the administration of the<br>highest planned activity of 7.5 GBq of <sup>177</sup> Lu-3BP-227<br>from the vial.  | Clarification   |
| 7.1.1.2 | <br>The screening and treatment IMP are formulated in a<br>solution of water for injection containing<br>DiethyleneTriaminePentaAcetate (DTPA) to complex<br>free <sup>177</sup> Lu and ascorbic acid to protect the drug product<br>from radiolysis and to have a longer shelf life. The IMP<br>consists of a sterile clear solution, free from visible<br>particles, filled in 25 mL Type I borosilicate glass vials,<br>closed with a chlorobutyl rubber stopper and crimped<br>with aluminium seal. | <br>The screening and treatment IMP are formulated in a<br>solution of <b>sterile</b> water for injection containing<br>DiethyleneTriaminePentaAcetate (DTPA) to complex<br>free <sup>177</sup> Lu and <b>ascorbate buffer</b> to protect the drug<br>product from radiolysis and to <b>extend the</b> shelf life. The<br>IMP consists of a sterile clear <b>to slightly yellow</b><br>solution, free <b>of suspended</b> particles, filled in 25 mL<br>Type I borosilicate glass vials, closed with a chlorobutyl<br>rubber stopper and crimped with aluminium seal. | Clarification   |
| 7.1.3   | <br>In case of central manufacturing by the central<br>manufacturing organisation, the IMP should be stored<br>between +2°C and +8°C.<br>Since the IMP shelf life will vary depending on the<br>manufacturing setup (decentralised or centralised) and<br>potentially on any updates supported by available<br>updated supporting data during the conduct of the study,<br>it is referred to the IMP label for the exact shelf life.  | <br>In case of manufacturing by the central manufacturing<br>organisation, the IMP should be stored between +2°C<br>and +8°C.<br>Since the IMP shelf life will vary depending on the<br><b>mode of</b> manufacturing (decentralised or centralised)<br>and potentially on supporting data <b>generated</b> during the<br>conduct of the study, the exact shelf <b>life is indicated on</b><br><b>the IMP label</b> .  | Clarification   |
| 8.1.1   | All AEs irrespective of causality up to 6 month<br>follow up from end of treatment period should be<br>reported to the sponsor. AEs reported after this<br>timepoint up to 2-year follow-up should be reported if<br>evaluated as related to the IMP by the investigator.   | <br>All AEs, irrespective of causality, are to be reported<br>to the sponsor up to 6 months after the EOCT, ED or<br>EOAC visit or until new antitumour treatment<br>starts, whichever comes first. After this timepoint, up<br>to the end of the 5-year follow-up period, AEs should<br>only be reported if the event is evaluated as related to<br>the IMP or study procedure by the investigator.  | Clarification of reporting<br>requirements during the long-term<br>safety follow-up |
| 8.1.1   | <br>Events that are reported during the study and cannot be<br>differentiated with PD should be captured as AE/SAEs.<br>Death due to disease progression should be reported as  | Events that cannot be differentiated <b>from progressive</b><br><b>disease</b> (PD) should be captured as AE/SAEs. Death<br>due to disease progression should be reported as <b>an</b>  | Clarification   |

|         | Version 7.0, 20 June 20 <u>1</u> 9                        | Version 8.0, 12 June 2020                                   | RATIONALE FOR CHANGE              |
|---------|---|---|-----------------------------------|
| Section | WAS   | IS  |                                   |
|         | SAE irrespective of causality if reported during the      | SAE irrespective of causality if reported during the        |                                   |
|         | study period and its early follow up up to 6 months after | study period and up to 6 months after the end of the        |                                   |
|         | the last dose of study drug.                              | cycle of the last dose of study drug.                       |                                   |
| 8.1.2.3 | Any AE/SAE occurring during the study, from               | Any AEs/SAEs, irrespective of causality, are to be          | Clarification of reporting        |
|         | informed consent up to 6 months after long term           | reported to the sponsor up to 6 months after the            | requirements during the long-term |
|         | follow up period start (after EOCT or EOAC or ED),        | EOCT, ED or EOAC visit or until new antitumour              | safety follow-up                  |
|         | must be reported to the sponsor.                          | treatment starts, whichever comes first. After this         |                                   |
|         | Any AE considered related to IMP administration that      | timepoint, up to the end of the 5-year follow-up            |                                   |
|         | the investigator becomes aware of after completion of     | period, AEs/SAEs should only be reported if the             |                                   |
|         | the EOS/ED visit must be reported to the sponsor and      | event is evaluated as related to the IMP or study           |                                   |
|         | will be recorded in the safety database. EOS for AE       | procedure by the investigator.                              |                                   |
|         | reporting is defined as the end of 24 month follow-up     | Any AE considered related to IMP administration that        |                                   |
|         | or death.   | the investigator becomes aware of after completion of       |                                   |
|         |   | the EOS/ED visit must be reported to the sponsor and        |                                   |
|         |   | will be recorded in the eCRF. EOS for AE reporting is       |                                   |
|         |   | defined as the end of <b>the 5-year</b> follow-up or death. |                                   |
| 8.1.3   | New text added  | In addition to the above criteria, any additional AE        | Clarification following           |
|         |   | that the sponsor or an investigator considers serious       | COVID-19 pandemic.                |
|         |   | should be immediately reported to the sponsor. This         |                                   |
|         |   | includes any suspected or confirmed coronavirus             |                                   |
|         |   | COVID-19 (SARS CoV-2) infection (seriousness                |                                   |
|         |   | criteria should be "other medically significant" if no      |                                   |
|         |   | other seriousness criteria are present (e.g.                |                                   |
|         |   | hospitalisation)).  |                                   |
|         |   | In case of suspected or confirmed COVID-19                  |                                   |
|         |   | infection, the INP administration may be                    |                                   |
|         |   | temporarily discontinued depending on the                   |                                   |
|         |   | subject's clinical presentation. In some cases, the         |                                   |
|         |   | the IMD administration is request                           |                                   |
| 912     | Any appropriate many of notification may be used          | the livir authillistration is resumed.                      | Clarification                     |
| 0.1.3   | The appropriate means of nonneation may be used.          | Eman is the preferred method of SAE nouncation.             | Clarification                     |
|         |   | it chian is not available, factimile transmission is the    |                                   |
|         |   | and many memory in fact of cumstances and in the            |                                   |
|         |   | telephone is acceptable                                     |                                   |
|         |   | נכוכטווטווכ וא מכנכטומטוכ.                                  |                                   |

| ance to the<br>the reporting<br>pregnancy<br>le partner of<br>ential and |
|--|
| ance to the<br>the reporting<br>pregnancy<br>le partner of<br>ential and |
| the reporting<br>pregnancy<br>le partner of<br>cential and               |
| pregnancy<br>le partner of<br>cential and                                |
| le partner of<br>tential and   |
| tential and  |
|  |
| of AEs/SAEs  |
| nate/child and   |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |
|  |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020  | RATIONALE FOR CHANGE  |
|---------|---|--|---|
| Section | WAS   | IS   |   |
| 8.2.4   | Table 6 Timepoints and Replicates of 12-lead ECG<br>Recordings  | Table 6 Timepoints and Replicates of 12-lead ECG Recordings  | Clarification   |
|         | Timepoint column:   | Timepoint column:  |   |
|         | Before infusion   | Before infusion (minus 15 minutes)   |   |
|         | End of infusion   | End of infusion (±15 minutes)  |   |
|         | 4 hours after the end of infusion   | 4 hours (±30 minutes) after the end of infusion  |   |
|         | Before infusion   | Before infusion (minus 15 minutes)   |   |
|         | End of infusion   | End of infusion (±15 minutes)  |   |
|         | 4 hours after the end of infusion   | 4 hours (±30 minutes) after the end of infusion  |   |
|         | 24 hours after the end of infusion  | 24 hours (±4 hours) after the end of infusion  |   |
| 9.1.2   | <br>The samples for urine total activity concentration  | <br>The samples for urine total activity concentration   | Clarification   |
|         | analysis will be taken from urine collected during four different periods at Cycle 1 only: from the start of the <sup>177</sup> Lu-3BP-227 infusion to 6 hours, 6 to 12 hours, 12 to  | analysis will be taken from urine collected during four different periods at Cycle 1 only: from the start of the <sup>177</sup> Lu-3BP-227 infusion to 6 hours, 6 to 12 hours, 12 to   |   |
|         | 24 hours and 24 to 48 hours after the end of infusion (from the start of the <sup>177</sup> Lu-3BP-227 infusion to 6 hours  | 24 hours and 24 to 48 hours after the end of infusion<br>(from the start of the <sup>177</sup> Lu-3BP-227 infusion to 6 hours  |   |
|         | after the end of the infusion only for US sites) <del>, with an initial</del> -void <del>collection</del> shortly before the infusion and <del>a</del>  | after the end of the infusion only for US sites). <b>Subjects should</b> void shortly before the <sup>177</sup> Lu-3BP-227 infusion  |   |
|         | final collection 48 hours after the end of $^{+++}Lu$ 3BP 227   | (not to be collected) and shortly before the end of the  |   |
| 9.2     | During the treatment period, to improve the determination of the biokinetics of <sup>177</sup> Lu-3BP-227 and perform an absolute quantification of radioactivity in target organs, whole body scans (planar scintigraphy) and SPECT/CT will be performed at Cycles 1 and 2 at the following timepoint ranges just after the end of <sup>177</sup> Lu-3BP-227 infusion: | <b>Last collection period</b> .<br>During the treatment period, to improve the determination of the biokinetics of <sup>177</sup> Lu-3BP-227 and perform an absolute quantification of radioactivity in target organs, whole body scans (planar scintigraphy) and SPECT/CT will be performed at Cycles 1 and 2 at the following timepoints just after the end of <sup>177</sup> Lu-3BP-227 infusion: | To optimise the dosimetry<br>evaluation through adaptation of<br>the imaging schedule |
|         | • Day 1: 0 to 1 hour (before urination) and 2 to 4 hours  | • Day 1: 4 (±2) hours  |   |
|         | • Day 2: <del>16 to</del> 24 hours  | • Day 2: 24 (±6) hours   |   |
|         | • Day 3: 4 <del>0 to</del> 48 hours   | • Day 3: 48 (±6) hours   |   |
|         | • Day 4: 72 to 96 hours   | • Day 4: 72 to 96 hours  |   |
|         |   | • Day 7s to 8: 138 to 168 hours  |   |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>   |
|---------|---|---|---|
| Section | WAS   | IS  |   |
|         | At the timepoint between 16 to 24 hours, SPECT/<br>standard dose CT will be performed. For all other time<br>points, SPECT/low dose CT will be performed.   | Within each cycle, a single SPECT/standard dose CT<br>will be performed. For all other time points,<br>SPECT/low dose CT will be performed. Details of the<br>procedures will be provided in the Image<br>Acquisition Guidelines.   |   |
| 9.2     | <br>After the second and fourth additional administrations,<br>a single whole body scan and SPECT/standard dose CT<br>between 16 and 24 hours will be performed.  | After the second and fourth additional administrations, a single whole body scan and SPECT/standard dose CT <b>at 48 (±6)</b> hours will be performed.  | Change of timepoint to 48 hours<br>where the quality of SPECT<br>images (with respect to the<br>background noise) is expected to<br>be improved |
| 9.3.2   | <br>The samples for urine 3BP 227 concentration analysis<br>will be taken from urine collected during four different<br>periods at Cycle 1 only: from the start of the infusion to<br>6 hours, 6 to 12 hours, 12 to 24 hours and 24 to 48 hours<br>after the start of infusion (from the start of the infusion<br>to 6 hours after the end of the infusion only in US sites),<br>with an initial void collection shortly before the<br>infusion and a final collection 48 hours after the end of<br><sup>177</sup> Lu 3BP 227 infusion. | The samples for urine 3BP 227 concentration analysis will be taken from urine collected during four different periods at Cycle 1 only: from the start of the infusion to 6 hours, 6 to 12 hours, 12 to 24 hours and 24 to 48 hours after the start of infusion (from the start of the infusion to 6 hours after the end of the infusion only in US sites). <b>Subjects should</b> void shortly before the <sup>177</sup> Lu-3BP-227 infusion (not to be collected) and shortly before the end of the last collection period.  | Clarification   |
| 10.1.2  | <br>After EOCT or EOAC, subjects will be followed up<br>every 3 months (±2 weeks) to further evaluate the<br>efficacy (by CT or MRI) until <del>24 months</del> , disease<br>progression, administration of any other chemotherapy<br>or radiotherapy, <del>or</del> death, whichever occurs first (see<br>Section 6.2.3).  | After the EOCT, ED or EOAC visit, subjects will be<br>followed up every 3 months (±2 weeks) to further<br>evaluate the efficacy (by CT or MRI) until disease<br>progression, administration of any other chemotherapy<br>or radiotherapy, lost to follow-up, withdrawal of<br>consent, death or a maximum of 2 years, whichever<br>occurs first (see Section 6.2.3). After disease<br>progression is confirmed, no further CT/MRI scans<br>will be required for tumour/disease assessment. The<br>survival status and safety of the subjects will<br>continue to be monitored as indicated in<br>Section 8.1.2.3 until lost to follow-up, withdrawal of<br>consent, death or a maximum of 5 years, whichever<br>occurs first. | Clarification of requirements<br>during the long-term follow-up.  |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | RATIONALE FOR CHANGE                                       |
|---------|---|---|--|
| Section | WAS   | IS  |  |
| 10.1.3  | After EOCT, subjects will be followed up every<br>3 months ( $\pm 2$ weeks) to further evaluate the<br>efficacy (by CT or MRI) until 24 months, disease<br>progression, administration of any other chemotherapy<br>or radiotherapy, or death, whichever occurs first (see<br>Section 6.2.3). | After the EOCT, ED or EOAC visit, subjects will be<br>followed up every 3 months ( $\pm 2$ weeks) to<br>further evaluate the efficacy (by CT or MRI) until<br>disease progression, administration of any other<br>chemotherapy or radiotherapy, lost to follow-up, death<br>or a maximum of 2 years, whichever occurs first (see<br>Section 6.2.3). | Updated long-term follow-up<br>period.                     |
| 10.4    | 10.4 Double Strand Breaks in Peripheral<br>Lymphoevtes  | Section deleted   | Assessment of DNA-DSB in<br>peripheral lymphocytes removed |
| 10.5    | 10.5 Genomic Profiling in Circulating Cell-free<br>DNA<br>CCI   | 10.4 Genomic Profiling in Circulating Cell-free<br>DNA and Germline DNA<br>CCI  |  |

|         | Version 7.0, 20 June 2019 | Version 8.0, 12 June 2020 | RATIONALE FOR CHANGE   |
|---------|---------------------------|---------------------------|--|
| Section | WAS                       | IS                        |  |
|         |                           |                           | CCI  |
| 10.6    | 10.7 Tumour Bionsv        | 10.6 Tumour Bionsv        | Clarification on biopsy collection<br>and wording added for subjects<br>participating in the imaging study |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>   |
|---------|---|--|---|
| Section | WAS   | IS   |   |
| Section | WAS   |  |   |
| 12.3.1  | Subjects in each phase of the study will have 2-year safety follow-up after the end of the last <sup>177</sup> Lu 3BP 227   | Subjects in each phase of the study will have a 5-year safety follow-up after the EOCT, ED or EOAC visit.              | Updated long-term follow-up period.   |
| 19.1    | Table 8 Schedule of Assessments for Phase I – DoseEscalationTimepoint changed:Cycle 1 Day 8   | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Timepoint changed:<br>Cycle 1 Day 7                | To amend the visit schedule to<br>align with the change in imaging<br>schedule        |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose         Escalation         Timepoint changed:         Cycle 2 Day 8  | Table 8 Schedule of Assessments for Phase I – Dose         Escalation         Timepoint changed:         Cycle 2 Day 7 | To amend the visit schedule to<br>align with the change in imaging<br>schedule        |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose       Escalation         Timepoints updated       Image: Compare the second sec | Table 8 Schedule of Assessments for Phase I – Dose Escalation  | To optimise the dosimetry<br>evaluation through adaptation of<br>the imaging schedule |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>  |
|---------|---|---|--|
| Section | WAS   | IS  |  |
|         |   | Timepoint added at Cycle 1 Day 7 and Cycle 2 Day 7<br>for:<br>Planar scintigraphy<br>SPECT/CT scan  |  |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose Escalation         Timepoint deleted:         Haematology and biochemistry at Day 2 of Cycle 1   | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Timepoint deleted   | Day 2 samples were initially<br>proposed for the safety evaluation<br>of the short-term effects (within<br>24 hours) of the IMP on laboratory<br>parameters. Following evaluation<br>of these results after IMP<br>administration in the patients<br>treated so far, there was no change<br>in the laboratory parameters<br>within 24 hours of IMP<br>administration. Therefore, these<br>laboratory tests do not add any<br>value to the clinical safety or<br>efficacy assessment. The sites can<br>still perform the laboratory tests<br>on Day 2, as on any other day, if<br>needed. |
| 19.1    | Table 8 Schedule of Assessments for Phase I – DoseEscalationPeripheral lymphocytes for DNA DSB (selectedcentres) [s] assessment at Day 1 of Cycle 1, Day 4 ofCycle 1, Day 1 of Cycle 2 and Day 4 of Cycle 2 | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Assessment deleted  | Assessment removed   |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Timepoints deleted  | Table 8 Schedule of Assessments for Phase I – DoseEscalationTimepoints were removed from the long-termfollow-up for the following assessments:ECOG performance statusVital signsBody weightUrinalysis | Clarification of requirements<br>during the long-term follow-up.   |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose Escalation   | Table 8 Schedule of Assessments for Phase I – Dose Escalation   | Clarification of requirements during the long-term follow-up.  |

|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | RATIONALE FOR CHANGE  |
|---------|--|--|---|
| Section | WAS  | IS   |   |
|         | New assessment added   | New assessment added for the long-term follow-up:  |   |
|         |  | Survival status  |   |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose         Escalation         Footnote b:         follow-up visits will take place every 3 months and will         start after the last cycle of IMP administration.   | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Footnote b:<br>follow-up visits will take place every 3 months<br>(±2 weeks) and will start after the EOCT, ED or<br>EOAC visit. Efficacy will be assessed until disease<br>progression, administration of any other<br>chemotherapy or radiotherapy, lost to follow-up,<br>withdrawal of consent, death or a maximum of<br>2 years, whichever occurs first (see Section 6.2.3).<br>After disease progression is confirmed, no further<br>CT/MRI scans will be required for tumour/disease<br>assessment. The survival status and safety of the<br>subjects will continue to be monitored as indicated<br>in Section 8.1.2.3 until lost to follow-up, withdrawal<br>of consent, death or a maximum of 5 years, | Clarification of requirements<br>during the long-term follow-up.                      |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Footnote k:<br>ten whole body scans (planar scintigraphy) and<br>SPECT/CT acquisitions will be performed during the<br>treatment period. Whole body scans (planar<br>scintigraphy) and SPECT/CT scan will be performed<br>just after the end of the <sup>177</sup> Lu 3BP 227 infusion at 0 to<br>1 hour (before urination), 2 to 4 hours, 16 to 24 hours,<br>40 to 48 hours and at 72 to 96 hours. SPECT/standard<br>dose CT will be performed at the 16 to 24 hour<br>timepoint, SPECT/low dose CT will be performed at all<br>other timepoints. | whichever occurs first.Table 8 Schedule of Assessments for Phase I – DoseEscalationFootnote k:eleven whole body scans (planar scintigraphy) andSPECT/CT acquisitions will be performed during thetreatment period. Whole body scans (planarscintigraphy) and SPECT/CT scan will be performed atthe following timepoints just after the end of 177Lu-3BP-227 infusion: Day 1: 4 (±2) hours, Day 2: 24 (±6)hours, Day 3: 48 (±6) hours, Day 4: 72 to 96 hours,and Days 7 to 8: 138 to 168 hours. Within each cycle,a single SPECT/standard dose CT will be performed. ASPECT/low dose CT will be performed at all othertimepoints. Details of the procedures will be providedin the Image Acquisition Guidelines   | To optimise the dosimetry<br>evaluation through adaptation of<br>the imaging schedule |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose           Escalation  | Table 8 Schedule of Assessments for Phase I – Dose         Escalation  | After reviewing subjects'<br>SPECT/CT images at different                             |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>   |
|---------|---|--|---|
| Section | WAS   | IS   |   |
|         | Footnote k:<br><br>At screening, planar scintigraphy will be performed<br>after screening administration of <sup>177</sup> Lu-3BP-227 <del>;</del><br><del>SPECT/CT can be performed as per investigator's</del><br><del>judgement</del> .  | Footnote k:<br><br>At screening, planar scintigraphy (1 or 2 timepoint(s)<br>at the investigator's discretion) and optional<br>SPECT/CT scans (up to 2 at the investigator's<br>discretion) will be performed after screening<br>administration of <sup>177</sup> Lu-3BP-227.  | timepoints following the<br>therapeutic dose of<br><sup>177</sup> Lu-IPN01087, it is concluded<br>that the quality of SPECT image<br>with respect to the background<br>noise has substantially improved<br>at the 48-hour and later<br>timepoints. Considering the<br>significant number of screen<br>failures due to lack of radioactive<br>uptake, which is contrary to the<br>information available in the<br>literature on the expression of<br>NTSR1 receptor, the image<br>acquisition timepoint has been<br>moved to 48 hours after the<br>screening dose. |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose Escalation<br>Footnote n:<br>prior to and at the end of <sup>177</sup> Lu-3BP-227 infusion (0) as<br>well as $30\pm5$ minutes, $90\pm15$ minutes, 4 hours $\pm10$<br>minutes after the end of <sup>177</sup> Lu-3BP-227 infusion.  | Table 8 Schedule of Assessments for Phase I – Dose Escalation<br>Footnote n:<br>prior to and at the end of $^{177}$ Lu-3BP-227 infusion (0) as<br>well as 30±5 minutes, 90±15 minutes,<br>4 hours± <b>3</b> 0 minutes after the end of $^{177}$ Lu-3BP-227<br>infusion.  | To homogenise the time window<br>duration between vital signs and<br>PK samples   |
| 19.1    | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Footnote o:<br>during the screening period, a triplicate 12-lead ECG<br>will be recorded before the screening administration,<br>and a single 12-lead ECG will be recorded at the end of<br><sup>177</sup> Lu-3BP-227 infusion and 4 hours after the end of<br>infusion. At each treatment administration, a triplicate<br>12-lead ECG will be recorded on Day 1 before the<br>infusion (baseline) and a single 12-lead ECG recordings<br>at the end of <sup>177</sup> Lu-3BP-227 infusion, at 4 hours after<br>the end of <sup>177</sup> Lu-3BP-227 infusion and on Day 2 at | Table 8 Schedule of Assessments for Phase I – Dose<br>Escalation<br>Footnote o:<br>during the screening period, a triplicate 12-lead ECG<br>will be recorded before the screening administration<br>(minus 15 minutes), and a single 12-lead ECG will be<br>recorded at the end of <sup>177</sup> Lu-3BP-227 infusion<br>(±15 minutes) and 4 hours (±30 minutes) after the end<br>of infusion. At each treatment administration, a<br>triplicate 12-lead ECG will be recorded on Day 1 before<br>the infusion (baseline) (minus 15 minutes) and a single<br>12-lead ECG recordings at the end of <sup>177</sup> Lu-3BP-227 | Clarification   |

|         | Version 7.0, 20 June 2019                                       | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>        |  |
|---------|---|--|------------------------------------|--|
| Section | WAS   | IS   |                                    |  |
|         | 24 hours after the end of <sup>177</sup> Lu-3BP-227 infusion as | infusion (±15 minutes), at 4 hours after the end of              | 1 of                               |  |
|         | well as at EOCT. In any case, 12-lead computerised              | <sup>177</sup> Lu-3BP-227 infusion (±30 minutes) and on Day 2 at |                                    |  |
|         | standard ECG, will be recorded in supine position after         | 24 hours after the end of <sup>177</sup> Lu-3BP-227 infusion     |                                    |  |
|         | at least 5 minutes of rest.                                     | (±4 hours) as well as at EOCT. All 12-lead                       |                                    |  |
|         |   | computerised standard ECGs will be recorded in the               |                                    |  |
|         |   | supine position after at least 5 minutes of rest.                |                                    |  |
| 19.1    | Table 8 Schedule of Assessments for Phase I - Dose              | Table 8 Schedule of Assessments for Phase I – Dose               | Collection of germline DNA         |  |
|         | Escalation  | Escalation   | modified to allow more flexibility |  |
|         | Footnote s:   | Footnote s:  | to the sites for biobanking blood  |  |
|         | samples for DNA DSB will be taken, starting from                | one blood sample for germline DNA will be collected              | collection and to avoid blood      |  |
|         | Cohort 3, at each administration at the following               | at screening (after confirmation of <sup>177</sup> Lu-3BP-227    | sampling from screening failure    |  |
|         | timepoints: before infusion (baseline), 4 hours and 72 to       | tumour uptake) or at Day 1 of Cycle 1 before the                 | subjects.                          |  |
|         | 96 hours after the end of <sup>177</sup> Lu 3BP 227 infusion.   | infusion   |                                    |  |
|         |   |  |                                    |  |
| 10.1    | Table & Schedule of Assessments for Phase I Dose                | Table 8 Schedule of Assessments for Phase I Dose                 | Clarification of requirements      |  |
| 17.1    | Fscalation  | Fscalation   | during the long-term follow-up     |  |
|         | New footnote added  | Footnote <b>z</b> :  | during the fong term fonow up.     |  |
|         |   | Any AFs/SAFs irrespective of causality are to be                 |                                    |  |
|         |   | reported to the sponsor up to 6 months after the                 |                                    |  |
|         |   | EQCT ED or EOAC visit or until new antitumour                    |                                    |  |
|         |   | treatment starts, whichever comes first. After this              |                                    |  |
|         |   | timepoint, up to the end of the 5-year follow-up                 |                                    |  |
|         |   | period. AEs/SAEs should only be reported if the                  |                                    |  |
|         |   | event is evaluated as related to the IMP or study                |                                    |  |
|         |   | procedure by the investigator.                                   |                                    |  |

|         | Version 7.0, 20 June 2019   | Version 8.0, 12 June 2020   | <b>RATIONALE FOR CHANGE</b>   |
|---------|---|---|---|
| Section | WAS   | IS  |   |
| CCI     |   |   |   |
| 19.1    | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Timepoint changed:<br>Additional cycles Day <del>8</del>     | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Timepoint changed:<br>Additional cycles Day 7  | To amend the visit schedule to<br>align with the change in imaging<br>schedule  |
| 19.1    | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Timepoints updated   | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Timepoint added at Day 7 for:<br>Planar scintigraphy<br>SPECT/CT scan  | To optimise the dosimetry<br>evaluation through adaptation of<br>the imaging schedule   |
| 19.1    | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote a:<br>ceCT/MRI will be done at Cycles 4 and 6 only. | Table 9 Schedule of Assessment for Additional Cyclesin Case of Clinical Benefit and Good TolerabilityFootnote a:ceCT/MRI will be done every 2 cycles duringadditional administrations, to confirm the clinicalbenefit after 2 additional administrations (i.e. atCycle 4 and EOAC). In case of early discontinuation, | Clarification of when ceCT/MRI<br>scans should be performed during<br>additional cycles, to be consistent<br>with the core trial period<br>assessments. |

|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | RATIONALE FOR CHANGE  |
|---------|--|--|---|
| Section | WAS  | IS   |   |
|         |  | the CeCT/MRI should be performed at the ED visit,  |   |
|         |  | to confirm any disease progression.  |   |
| 19.1    | Table 9 Schedule of Assessment for Additional Cyclesin Case of Clinical Benefit and Good TolerabilityFootnote b:full dosimetry assessment will be performed after thefirst and third additional administrations as describedfor Cycles 1 and 2. After the second and fourthadditional administrations, only a singleSPECT/standard dose CT at 16 to 24 hours will beperformed  | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote b:<br>full dosimetry assessment will be performed after the<br>first and third additional administrations as described<br>for Cycles 1 and 2. After the second and fourth<br>additional administrations, only a single<br>SPECT/standard dose CT at <b>48</b> (±6) hours will be<br>performed  | Change of timepoint to 48 hours<br>where the quality of SPECT<br>images (with respect to the<br>background noise) is expected to<br>be improved |
| 19.1    | Table 9 Schedule of Assessment for Additional Cyclesin Case of Clinical Benefit and Good TolerabilityFootnote c:After the first and the third additional administration,blood samplings will be performed just before <sup>177</sup> Lu-3BP-227 infusion (baseline), at the end of infusion of <sup>177</sup> Lu-3BP-227 (0), 5 minutes ±1 minute, 30 minutes ±5minutes, 1 hour ±5 minutes and 4 hours ±30 minutes, 6hours ±30 minutes, 8 hours ±1 hour, 24 hours ±2 hoursand 48 hours ±2 hours, and 72 to 96 hours ±2 hours.After the second and fourth additional administrations asingle blood collection will be performed at 24 hoursonly (as close as possible to the SPECT/CT). | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote c:<br>After the first and the third additional administration,<br>blood samplings will be performed just before<br>$^{177}Lu-3BP-227$ infusion (baseline), at the end of<br>infusion of $^{177}Lu-3BP-227$ (0), 5 minutes ±1 minute,<br>30 minutes ±5 minutes, <b>90</b> minutes ±15 minutes and<br>4 hours ±30 minutes, 24 hours ±2 hours and 48 hours<br>±2 hours, and 72 to 96 hours ±2 hours. After the second<br>and fourth additional administrations a single blood<br>collection will be performed at 24 hours only (as close<br>as possible to the SPECT/CT). | Clarification of inconsistency<br>between core trial period<br>assessments and additional cycle<br>assessments                                  |
| 19.1    | Table 9 Schedule of Assessment for Additional Cyclesin Case of Clinical Benefit and Good TolerabilityFootnote d:prior to and at the end of <sup>177</sup> Lu-3BP-227 infusion (0) aswell as 30±5 minutes, 90±15 minutes, 4 hours±10minutes after the end of <sup>177</sup> Lu-3BP-227 infusion.  | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote d:<br>prior to and at the end of <sup>177</sup> Lu-3BP-227 infusion (0) as<br>well as 30±5 minutes, 90±15 minutes, 4 hours± <b>3</b> 0<br>minutes after the end of <sup>177</sup> Lu-3BP-227 infusion.   | Clarification of inconsistency<br>between core trial period<br>assessments and additional cycle<br>assessments                                  |
| 19.1    | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote e:<br>at each cycle, a triplicate 12-lead ECG will be<br>recorded on Day 1 before the infusion (baseline). A<br>single 12-lead computerised standard ECG, with   | Table 9 Schedule of Assessment for Additional Cycles<br>in Case of Clinical Benefit and Good Tolerability<br>Footnote e:<br>at each cycle, a triplicate 12-lead ECG will be recorded<br>on Day 1 before the infusion (baseline) (minus<br>15 minutes). A single 12-lead computerised standard  | Clarification   |

|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>    |
|---------|--|--|--------------------------------|
| Section | WAS  | IS   |                                |
|         | paper printout, will be recorded in supine position after          | ECG, with paper printout, will be recorded in supine                           |                                |
|         | at least 5 minutes of rest during each cycle on Day 1 at           | position after at least 5 minutes of rest during each cycle                    |                                |
|         | the end of <sup>177</sup> Lu-3BP-227 infusion and at 4 hours after | on Day 1 at the end of <sup>177</sup> Lu-3BP-227 infusion                      |                                |
|         | the end of <sup>177</sup> Lu-3BP-227 infusion as well as at EOAC.  | $(\pm 15 \text{ minutes})$ and at 4 hours $(\pm 30 \text{ minutes})$ after the |                                |
|         |  | end of <sup>1//</sup> Lu-3BP-227 infusion as well as at EOAC.                  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
| 19.2    | Assessment in blood sampling summary table:                        | Assessment in blood sampling summary table:                                    | Amend blood volumes to reflect |
|         | Core trial   | Core trial   | change in assessments          |
|         | Clinical laboratory tests [a]                                      | Clinical laboratory tests [a]  |                                |
|         | -Maximum blood volume per sample (mL): 25                          | -Maximum blood volume per sample (mL): 25                                      |                                |
|         | -Maximum number of blood samples: 11                               | -Maximum number of blood samples: 10   |                                |
|         | -Maximum total volume (mL): <del>275</del>                         | -Maximum total volume (mL): 250  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |
|         |  |  |                                |

|         | Version 7.0, 20 June 2019  | Version 8.0, 12 June 2020  | <b>RATIONALE FOR CHANGE</b>   |
|---------|--|--|---|
| Section | WAS  | IS   |   |
|         | -Maximum number of blood samples: <del>1</del><br>-Maximum total volume (mL): <del>10</del>                          | -Maximum number of blood samples: <b>2</b><br>-Maximum total volume (mL): <b>20</b>  |   |
| 19.2    | Assessment in blood sampling summary table:<br>EOCT/ED<br>CC   | Assessment in blood sampling summary table:<br>EOCT/ED   | Amend blood volumes to reflect<br>change in assessments   |
| 19.2    | Blood sampling summary table:<br>Total volume over study period: Approximately <del>900</del><br>mL                  | Blood sampling summary table:<br>Total volume over study period: Approximately<br>820 mL   |   |
| 19.3    | The following assessments in the haematology table<br>were removed from the LTFU:<br>RBC count<br>Haematocrit<br>MCV | LTFU timepoints deleted  | Clarification of requirements during the long-term follow-up.   |
| 19.3    | Footnote added to the clinical chemistry table   | Footnote added to the clinical chemistry table:<br>a only to be performed if the total bilirubin is<br>abnormal i.e. outside the laboratory normal range | The testing of conjugated bilirubin<br>has been made optional because<br>the differentiation between<br>conjugated vs unconjugated<br>bilirubin is usually needed only if<br>the total bilirubin is elevated, to<br>find the cause of<br>hyperbilirubinemia. Hence the<br>conjugated bilirubin test is not<br>required if the value of total<br>bilirubin is within the laboratory<br>normal range. |
| 19.3    | LTFU assessment added  | The following assessment in the clinical chemistry table<br>was added to the LTFU:<br>Total bilirubin  | Clarification of requirements during the long-term follow-up.   |
| 19.3    | The following assessments in the clinical chemistry table were removed from the LTFU:<br>Albumin                     | LTFU timepoints deleted  | Clarification of requirements during the long-term follow-up.   |

|         | Version 7.0, 20 June 2019                              | Version 8.0, 12 June 2020              | <b>RATIONALE FOR CHANGE</b>        |
|---------|--|--|------------------------------------|
| Section | WAS  | IS                                     |                                    |
|         | ALP  |  |                                    |
|         | Calcium  |  |                                    |
|         | Chloride   |  |                                    |
|         | CRP  |  |                                    |
|         | eGFR   |  |                                    |
|         | Glucose  |  |                                    |
|         | Potassium  |  |                                    |
|         | Sodium   |  |                                    |
|         | Total bilirubin  |  |                                    |
|         | Total cholesterol                                      |  |                                    |
|         | Total protein  |  |                                    |
|         | TG   |  |                                    |
|         | Uric acid  |  |                                    |
| 19.3    | Assessment in clinical chemistry table                 | Assessment in clinical chemistry table | Clarification                      |
|         | Glucose <del>(fasting)</del>                           | Glucose                                |                                    |
| 19.3    | Urinalysis table                                       | Urinalysis table                       | Clarification for consistency with |
|         | New assessment added                                   | Specific gravity added                 | the eCRF                           |
| 19.3    | The following assessments in the urinalysis table were | LTFU timepoints deleted                | Clarification of requirements      |
|         | removed from the LTFU:                                 |  | during the long-term follow-up.    |
|         | Bilirubin  |  |                                    |
|         | Blood  |  |                                    |
|         | Glucose  |  |                                    |
|         | Ketones  |  |                                    |
|         | Leucocytes   |  |                                    |
|         | Nitrite  |  |                                    |
|         | pH   |  |                                    |
|         | Protein  |  |                                    |
|         | Proteinuria  |  |                                    |
|         | Urobilinogen   |  |                                    |
|         | Proteinuria (urine collection)                         |  |                                    |

# **SUMMARY & OUTCOME OF CHANGES:**

| STUDY NUMBER                 | D-FR-01087-001  |                                     |  |
|------------------------------|---|-------------------------------------|--|
| AMENDED PROTOCOL             | Version 8.0, 12 June 2020   |                                     |  |
| VERSION NUMBER AND           |   |                                     |  |
| DATE                         |   |                                     |  |
| SUBSTANTIAL 🔀                | NON-SUBSTANTIAL   |                                     |  |
| <b>REASON(S) FOR CHANGES</b> | The protocol was amended to update  | the following:                      |  |
|                              | Clarification of the inclusio follows:  | n criteria for subject selection as |  |
|                              | - to clearly state nonresectable locally advanced disease   |                                     |  |
|                              | - to clearly state that no further suitable treatment options are available for subjects eligible for the study   |                                     |  |
|                              | <ul> <li>Allow subjects screened and found positive for NTSR1 in the <sup>111</sup>In-IPN01087 phase I diagnostic study to take part in this study without having the diagnostic dose of <sup>177</sup>Lu-IPN01087 during the screening phase</li> <li>Extend the long-term follow-up period from 2 years to a maximum of 5 years or until lost to follow-up, withdrawal of consent or death, whichever occurs first.</li> <li>Revision of the DLT criteria to adequately describe the grading as stated in the CTCAE v5.0 dictionary</li> <li>Revision of the subject discontinuation rules so if there are life threatening toxicities outside of the DLT period, treatment will be discontinued</li> <li>Optimise the dosimetry evaluation through adaptation of the imaging schedule</li> </ul> |                                     |  |
|                              | <ul> <li>Clarification about COVID-19 added following the recent pandemic</li> <li>CCI</li> </ul>   |                                     |  |
|                              | Make various clarifications or  | d minor corrections for consistency |  |
| Other Action Required?       | CRF UPDATE  | Yes X<br>No (tick one)              |  |
|                              | LOCAL CONSENT FORM UPDATE   | Yes<br>No<br>(tick one)             |  |
|                              | DATABASE UPDATE   | Yes X<br>No C<br>(tick one)         |  |
|                              | STATISTICAL ANALYSIS PLAN<br>(SAP) UPDATE   | Yes X<br>No (tick one)              |  |